Distribution System Planning in Smart Grids to Accommodate Distributed Energy Resources and Electric Vehicles

by

Abdullah Bin Humayd

A thesis presented to the University of Waterloo in fulfillment of the thesis requirement for the degree of Doctor of Philosophy in Electrical and Computer Engineering

Waterloo, Ontario, Canada, 2017

© Abdullah Bin Humayd 2017

Examining Committee Membership

The following served on the Examining Committee for this thesis. The decision of the Examining Committee is by majority vote.

External Examiner	Udaya Annakkage Professor
Supervisor	Kankar Bhattacharya Professor
Internal Member	Ramadan El-Shatshat Lecturer
Internal Member	Catherine Gebotys Professor
Internal-external Member	Tarek Hegazy Professor

I hereby declare that I am the sole author of this thesis. This is a true copy of the thesis, including any required final revisions, as accepted by my examiners.

I understand that my thesis may be made electronically available to the public.

Abstract

Major changes in planning paradigms have taken place in power systems in recent years because of deregulation of the power industry, environmental policy changes, advancements in technology, and the transformation of the grid to intelligent systems, referred to as the smart grid. These changes will continue to drive the distribution systems planning function to evolve in the coming years. It is therefore important to develop effective planning strategies to identify the qualities, capabilities, and attributes that are necessary for the future distribution grid.

Demand response (DR), distributed generation (DG), energy storage systems (ESS), and plug-in electric vehicles (PEV) are expected to be a part of the solution of these distribution system planning challenges. However, very little of the present research on distribution system planning have considered these options simultaneously. Moreover, traditional planning options such as substation expansion, new feeder connections and capacitor placements should also be simultaneously considered. Such a coordinated planning can help evaluate the alternatives to provide maximum benefits to the network owner and customers.

With the increase in gas prices driven by a foreseeable fossil fuel depletion in the future, development in the automotive sector, and environmental concerns, penetration of PEVs has been increasing in recent times. The charging load of PEVs will definitely impact the distribution grid. To mitigate these effects, the local distribution companies (LDCs) need to adopt the right actions and policies, and develop associated infrastructure.

In the current context of smart grids, the LDCs need to control PEV charging demand while also considering customer preferences, which can lead to benefits such as deferment of the decisions on reinforcement and other investments, and maximize the use of existing infrastructure. In addition, LDCs need to establish rate structures that incentivize the use of smart charging and increase the adoption and use of PEVs, which can benefit both the LDCs and the customer.

This research focuses on developing models to investigate and address the problem of distribution system planning in the presence of PEV charging loads. First, a comprehensive long-term distribution planning framework from the perspective of LDCs is proposed considering DG, substations, capacitors, and feeders. Apart from considering the usual demand profile, the proposed framework considers uncontrolled and controlled (smart) PEV charging demand, as well as DR options. Based on a back-propagation algorithm combined with cost-benefit analysis, a novel approach is proposed to determine the optimal upgrade plan, allocation, and sizing of the selected components in distribution systems, to minimize the total capital and operating cost. A new iterative method is proposed which involves post-processing the plan decisions to guarantee acceptable adequacy levels for each year of the planning horizon.

Second, a generic and novel framework is proposed to assess the Distribution System Loading Margin (DSLM) to accommodate uncontrolled and smart PEV charging loads without the need for any additional investments or upgrades in the distribution system. The model determines what percentage of the fleet can be served by uncontrolled charging and smart charging, respectively. Monte Carlo simulation has been carried out to simulate the uncertainty of demand, drivers' behaviour, market share of PEV class, and charging level. The maximum allowable penetration of uncontrolled and smart charging loads are determined based on the current available market data pertaining to PEV type and charging level, considering different charging scenarios.

Finally, a PEV smart charging approach is proposed where the charging loads are incentivized by the LDC for every unit of energy controlled. A novel framework is proposed to determine the optimal participation of PEVs in the smart charging program and optimal incentives paid by the LDC to PEV customers, such that both parties are economically benefited. The proposed framework models the relationship between customers' participation and incentives offered by the LDC. The relationship between the expected investment deferral and hence the economic benefits from smart charging participation are considered as well. Monte Carlo simulation is carried out to simulate the uncertainty of demand, electricity market price, drivers' behaviour, PEV market share, and charging level.

Acknowledgements

First of all, I shall praise and thank Allah almighty for helping me through this difficult but very inspiring work and for providing me with the ideas and patience necessary for the successful completion of this thesis.

Then, I would like to express my sincere gratitude and appreciation to Professor Kankar Bhattacharya, for the invaluable guidance, continual support and encouragement which he provided throughout my graduate studies at the University of Waterloo. Prof. Bhattacharya's commitment with high quality research is an inspiration for pursuing excellence and success not only in academic endeavors but in every other aspect of life.

I would like to thank Prof. Ramadan El-Shatshat, Prof. Catherine Gebotys, and Prof. Tarek Hegazy for being in my Advisory Committee. Their careful reading and comments have significantly improved the quality of this thesis. I am also very thankful to Prof. Udaya Annakkage, from University of Manitoba, for serving as the external thesis examiner and for his insightful comments and observations. I would like also to thank Prof. Martin Ross for chairing my thesis examination committee.

Special thanks to Umm Al-Qura University, Makkah, Saudi Arabia, for the scholarship to pursue my graduate studies. I also would like to thank the Ministry of Education, Saudi Arabia, and the Saudi Arabian Cultural Bureau in Canada for their support.

I am truly indebted to my parents, my grand parents, my brothers, and my sisters for their endless support, encouragement, and prayers. I also extend my deepest appreciation to my dear brother Omar for his sincerity, friendship and encouragements during my academic journey.

I cannot express enough gratitude to my wife, Nada, for her constant support. Without her encouragements, this thesis would not be completed. For sure, my beautiful kids, Munirah, Noora, and Saleh have made me pursue my research more deeply, with their unconditional love, which helped me overcome the tough times that one faces at some points during research work.

Finally, I would like to thank all my friends at the University of Waterloo, who helped me through the course of my studies, discussed new ideas about my research, and made my life at Waterloo enjoyable.

Thank you all!

Dedication

To my father, my stepmother, my mother (May Allah be merciful to her), and my grand parents (May Allah be merciful to them)

Table of Contents

Li	st of	Tables	3	xii
Li	st of	Figure	2S	xiv
Li	st of	Abbre	eviations	xvi
1	Intr	oducti	on	1
	1.1	Motiva	ntion	1
	1.2	Literat	cure Review	4
		1.2.1	Traditional Distribution System Planning	4
		1.2.2	Distribution System Planning in the Presence of DGs \ldots .	5
		1.2.3	Planning for PEVs in Distribution System	9
	1.3	Resear	ch Objectives	11
	1.4	Thesis	Outline	12
2	Bac	kgrour	ıd	13
	2.1	Distrib	oution Systems	13
	2.2	Distrib	oution System Planning	13
		2.2.1	Traditional Distribution System Planning	14
		2.2.2	Distribution System Planning in the Presence of DGs	16
		2.2.3	Supply Adequacy of Distribution Systems	19

		2.2.4	Demand Response	21
	2.3	Plug-in	n Electric Vehicles (PEVs)	22
		2.3.1	PEV Load Characteristics	23
	2.4	Chargi	ing level	24
	2.5	Conclu	lding Remarks	24
	2.6	Introd	uction	25
	2.7	Nomer	nclature	25
	2.8	Propos	sed Framework For Distribution System Planning	27
	2.9	Smart	Distribution System Planning Model	28
		2.9.1	Smart-DSPLAN	28
		2.9.2	Smart-DSPLAN2	33
	2.10	Propos	sed Approach To Solution of Planning Problem	34
		2.10.1	Level-1: ADEQ-OPTSELECT	34
		2.10.2	Level-2: ADEQ-OPTPERIOD	36
	2.11	Test S	ystem	37
	2.12	Case S	$tudies \ldots \ldots$	39
		2.12.1	Case-1: Base Case; 33 Bus System	39
		2.12.2	Case-2: PEV Penetration; 33-Bus Test System	41
		2.12.3	Summary of 33-Bus System Studies	45
		2.12.4	Case-3: Base Case; 69 Bus Test System	46
		2.12.5	Computational Aspects	47
	2.13	Conclu	nsions	47
3	A N Dist	Novel] ributio	Framework for Evaluating Maximum PEV Penetration into on Systems	50
	3.1	Introd	uction	50
	3.2	Nomer	nclature	51
	3.3	Propos	sed Framework For Assessment of Distribution System Loading Margin	53
		-	<i>v</i> 0 0	

3.3.1 St	age-I: Uncontrolled PEV Charging Load Modeling 5	4
3.3.2 St	age-II: Distribution System Loading Margin Assessment 5	7
3.3.3 St	age-III: Assessment of Allowable PEV Penetration	9
3.4 Case Stud	lies And Assumptions	1
3.4.1 D	escriptions of Case Studies	1
3.4.2 As	ssumptions	2
3.5 Results A	nd Discussions	5
3.5.1 U	ncontrolled Charging Profiles	5
3.5.2 D	stribution System Loading Margin 6	6
3.5.3 M	aximum Allowable Penetration Level 6	7
3.5.4 Ef	fect of Charging Level	'1
255 F4	fect of Reactive Compensation Devices	'1
5.5.5 EI		2
3.5.6 Va	alidation of Results	
3.5.5 Va 3.5.7 Ce	alidation of Results 7 omputational Aspects 7	3
3.5.5 En 3.5.6 Va 3.5.7 Co 3.6 Conclusio	alidation of Results 7 pomputational Aspects 7 pns 7	'3 '4
3.5.5 En 3.5.6 Va 3.5.7 Co 3.6 Conclusio	alidation of Results 7 omputational Aspects 7 ons 7 ons 7 otimal Incontinues for Smort Charging Considering Utility	'3 '4
3.5.5 En 3.5.6 Va 3.5.7 Co 3.6 Conclusion 4 Design of Op Customer In	alidation of Results 7 omputational Aspects 7 ons 7 otimal Incentives for Smart Charging Considering Utility- 7 teractions and Distribution Systems Impact 7	'3 '4 5
3.5.5 En 3.5.6 Va 3.5.7 Co 3.6 Conclusion 4 Design of Op Customer In 4.1 Introduct	alidation of Results 7 omputational Aspects 7 ons 7 otimal Incentives for Smart Charging Considering Utility- 7 teractions and Distribution Systems Impact 7 ion 7	3 4 5 5
 3.5.5 En 3.5.6 Va 3.5.7 Co 3.6 Conclusion 4 Design of Op Customer In 4.1 Introduct 4.2 Nomencla 	alidation of Results 7 pomputational Aspects 7 pms 7 potimal Incentives for Smart Charging Considering Utility- 7 pteractions and Distribution Systems Impact 7 ion 7 pture 7	 3 4 5 5 7
 3.5.5 En 3.5.6 Va 3.5.7 Co 3.6 Conclusion 4 Design of Op Customer In 4.1 Introduct 4.2 Nomencla 4.3 Proposed 	alidation of Results 7 pomputational Aspects 7 pns 7 potimal Incentives for Smart Charging Considering Utility- 7 pteractions and Distribution Systems Impact 7 ion 7 pture 7 pture 7 pramework 7	73 74 5 75 7 8
 3.5.5 En 3.5.6 Va 3.5.7 Co 3.6 Conclusion 4 Design of Op Customer In 4.1 Introduct 4.2 Nomencla 4.3 Proposed 4.3.1 Co 	alidation of Results 7 pomputational Aspects 7 pass 7 potimal Incentives for Smart Charging Considering Utility- 7 potimal Incentives for Systems Impact	3 4 5 5 7 8 9
3.5.5 En 3.5.6 Va 3.5.7 Co 3.6 Conclusion 4 Design of Op Customer In 4.1 Introduct 4.2 Nomencla 4.3 Proposed 4.3.1 Co 4.3.2 Co	alidation of Results 7 pomputational Aspects 7 pms 7 potimal Incentives for Smart Charging Considering Utility- 7 potimal Incentives for Construction Incentives for Construction Incentives for Construction Incenting Utility- 7 pot	3 4 5 7 7 8 9 4
3.5.5 En 3.5.6 Va 3.5.7 Co 3.6 Conclusion 4 Design of Op Customer In 4.1 Introduct 4.2 Nomencla 4.3 Proposed 4.3.1 Co 4.3.2 Co 4.3.3 O	alidation of Results 7 pomputational Aspects 7 pms 7 point in a line of the system of the	³ 4 5 57 789 45
3.5.5 En 3.5.6 Va 3.5.7 Co 3.6 Conclusion 4 Design of Op Customer In 4.1 Introduct 4.2 Nomencla 4.3 Proposed 4.3.1 Co 4.3.2 Co 4.3.3 Op 4.4 Test Syst	alidation of Results 7 pomputational Aspects 7 pons 7 potimal Incentives for Smart Charging Considering Utility- 7 potime 7 ion 7 ature 7 pramework 7 post of Capacity Saving Model (CCSM) 7 ustomer Participation Model (CPM) 8 ptimal Incentives for Smart Charging Model (OISCM) 8 em and Assumptions 8	3 4 5 5 7 8 9 4 5 7 8 9 4 5 7
3.5.5 En 3.5.6 Va 3.5.7 Co 3.6 Conclusion 4 Design of Op Customer In 4.1 Introduct 4.2 Nomencla 4.3 Proposed 4.3.1 Co 4.3.2 Co 4.3.3 Op 4.4 Test Syst 4.5 Results A	alidation of Results 7 pomputational Aspects 7 ons 7 potimal Incentives for Smart Charging Considering Utility- 7 pteractions and Distribution Systems Impact 7 ion 7 ion 7 ptimel Incentives for Smart Charging Considering Utility- 7 teractions and Distribution Systems Impact 7 ion 7 store 7 framework 7 of Capacity Saving Model (CCSM) 7 ustomer Participation Model (CPM) 8 ptimal Incentives for Smart Charging Model (OISCM) 8 em and Assumptions 8 nd Discussions 8	3 4 5 5 7 8 9 4 5 7 9
3.5.5 En 3.5.7 Co 3.6 Conclusion 4 Design of Op Customer In 4.1 Introduct 4.2 Nomencla 4.3 Proposed 4.3.1 Co 4.3.2 Co 4.3.3 Op 4.4 Test Syst 4.5 Results A 4.5.1 Co	alidation of Results 7 pomputational Aspects 7 pms 7 potimal Incentives for Smart Charging Considering Utility- 7 potimal Incentives for Smart Charging Considering Utility- 7 potimal Incentives for Smart Charging Considering Utility- 7 teractions and Distribution Systems Impact 7 ion 7 ature 7 Framework 7 post of Capacity Saving Model (CCSM) 7 ustomer Participation Model (CPM) 8 ptimal Incentives for Smart Charging Model (OISCM) 8 em and Assumptions 8 ost of Capacity Saving Function 8 ost of Capacity Saving Function 8	34 578945799
3.5.5 En 3.5.6 Va 3.5.7 Ca 3.6 Conclusion 4 Design of Op Customer In 4.1 4.1 Introduct 4.2 Nomencla 4.3 Proposed 4.3.1 Ca 4.3.2 Ca 4.3.3 Op 4.4 Test Syst 4.5 Results A 4.5.1 Ca 4.5.2 Ca	alidation of Results 7 omputational Aspects 7 ons 7 otimal Incentives for Smart Charging Considering Utility- 7 ion 7 ion 7 otture 7 Framework 7 ost of Capacity Saving Model (CCSM) 7 Istomer Participation Model (CPM) 8 optimal Incentives for Smart Charging Model (OISCM) 8 optimal Incentives for Smart Charging Model (OISCM)<	3 4 5 7 8 9 4 5 7 9 9 9 9 9
3.5.5 En 3.5.7 Co 3.6 Conclusion 4 Design of Op Customer In 4.1 Introduct 4.2 Nomencla 4.3 Proposed 4.3.1 Co 4.3.2 Co 4.3.3 Op 4.4 Test Syst 4.5 Results A 4.5.1 Co 4.5.2 Co 4.5.3 Op	alidation of Results 7 pomputational Aspects 7 poms 7 potimal Incentives for Smart Charging Considering Utility- 7 ion 7 ature 7 by the reactions and Distribution Systems Impact 7 ion 7 ature 7 by the reactions and Distribution Systems Impact 7 ion 7 ion 7 by the reactions and Distribution Systems Impact 7 otture 7 for the reactions and Distribution Systems Impact 7 otture 7 by the reaction Saving Model (CCSM) 7 ost of Capacity Saving Model (CPM) 8 optimal Incentives for Smart Charging Model (OISCM) 8 ost of Capacity Saving Function 8 ost of Capacity Saving Function 8 ost of Capacity Saving Function 8 <t< td=""><td> 3 4 5 5 7 8 9 4 5 7 9 9 9 2 </td></t<>	 3 4 5 5 7 8 9 4 5 7 9 9 9 2

5	Con	clusions	95
	5.1	Summary and Conclusions	95
	5.2	Contributions	97
	5.3	Future Work	98
Re	efere	nces	99
A	PPEI	NDICES	111
A	Dist	ribution System Data	112

List of Tables

2.1	Objectives, decision variables, and constraints for DG planning problems .	18
2.2	PEV Charging Level Characteristics	24
2.3	Capital Cost of Utility Resources	38
2.4	Electricity Market Price	38
2.5	Reserve Margin, Added Capacity and LOLE	39
2.6	Step-by step Outcome of ADEQ-OPTSELECT with $\gamma=0.21$	40
2.7	Case-1 Results of ADEQ-OPTPERIOD	41
2.8	ADEQ-OPTPERIOD step-by-step procedure for year 6	42
2.9	PEV Charging Level Characteristics	42
2.10	Case-2a Results of ADEQ-OPTPERIOD	43
2.11	Case-2b: Results of ADEQ-OPTPERIOD	45
2.12	LDC Plan Comparison for 33-Bus Case Studies	46
2.13	Case-3 Results of ADEQ-OPTPERIOD	47
2.14	Model Statistics	48
3.1	PEV CLASSES CONSIDERED FOR STUDIES	56
3.2	Charging Level Scenarios Considered For Studies	57
3.3	Description of Case Studies	62
3.4	MAXIMUM ALOWABLE PENETRATION LEVEL	69
3.5	Bottleneck hours for the proposed case studies	69
3.6	Effect of reactive compensation devices on Case-3a	72

3.7	MODEL STATISTICS	74
4.2	EXPECTED PRR, CAPACITY DEFERRAL, AND BENEFIT FOR THE CONSIDERED CASE STUDIES	90
4.3	EXPECTED ANNUAL ENERGY, ENERGY COST, AND AVERAGE COST FOR THE CONSIDERED PEV CLASSES	91
4.4	Optimal Smart Charging Share, Incentives, and Rebate for the Considered Case Studies	93
A.1	Data for 33-Bus system: Feeder Parameters	113
A.2	Data for 33-Bus system: Load	114
A.3	Data for 69-Bus system: Feeder Parameters	115
A.3	Data for 69-Bus system: Feeder Parameters	116
A.3	Data for 69-Bus system: Feeder Parameters	117
A.4	Data for 69-Bus system: Load	118

List of Figures

2.1	Basic power system structure	14
2.2	Typical configuration of a radial distribution system	15
2.3	Traditional distribution planning framework	17
2.4	Framework for adequacy assessment in distribution system	20
2.5	Proposed framework	28
2.6	Architecture of the proposed approach to solution of planning problem	34
2.7	System demand without and with uncontrolled PEV charging	44
2.8	LOLE for different reserve margins, , with and without PEVs \ldots	46
3.1	Proposed framework for assessment of DSLM	53
3.2	Test system configuration and number of vehicles at each bus, shown inside the circles	63
3.3	RTS peak day load profile at some buses	63
3.4	Monte Carlo generated load profile at some buses for a given scenario	64
3.5	Probability of vehicle being at home	65
3.6	Uncontrolled charging profiles for the proposed case studies	66
3.7	DSLM profiles obtained from Stage-II	67
3.8	Case-3a stopping criteria	68
3.9	Expected Allowable Uncontrolled and Smart Charging Profiles	70
3.10	Impact of charging level to the penetration of PEVs	72
3.11	Case-3a voltage profile for the system buses at hour-14	73

4.1	Framework for LDC-PEV customer interaction based smart charging \ldots	80
4.2	Probability distribution function for home last arrival time $\ldots \ldots \ldots$	87
4.3	Probability distribution function for daily mileage driven	88
4.4	Expected PRR $(E[\lambda])$ versus smart charging participation	90
4.5	Expected benefit versus smart charging participation	91
4.6	Customer participation with respect to incentive and rebate $\ldots \ldots \ldots$	92
4.7	Convergence of OISCM Model for Case-3	93
4.8	Expected typical, uncontrolled, smart, and aggregated load profiles for the considered case studies	94

List of Abbreviations

All Electric Range
Battery Electric Vehicle
Distributed Generation
Depth of Discharge
Demand Response
Demand-Side Management
Forced Outage Rate
Hybrid Electric Vehicles
Local Distribution Company
Loss of Load Expectation
Loss of Load Probability
Optimal Power Flow
Plug-in Electric Vehicle
Plug-in Hybrid Electric Vehicle
Reliability Test System
State of Charge
Time of Use

Chapter 1

Introduction

1.1 Motivation

In the last decade, a major change in paradigm has taken place in power system planning because of deregulation of the power industry, policy changes driven by concerns on environmental emissions, advancement in technology including renewable energy generation, two-way communication between customer and utility, and the consequent transformation of the grid to a self-healing and intelligent system, referred to as smart grids. These changes have a significant bearing on the electrical distribution systems with increasing penetration of demand response (DR), distributed generation (DG), energy storage systems (ESS), and plug-in electric vehicles (PEV), together, known to as distributed energy resources (DERs), and proposed as part of the solution to today's energy and environmental challenges. Since DERs are mainly targeted at the distribution system level, design and planning issues at this level are key features for the development of the future distribution system.

DGs are expected to play an important role in the distribution system because of their numerous benefits. They can help in the deferral of system upgrades, such as large power plants and new transmission lines; and can also increase market competition, in generation, leading to better services and lower energy prices. DGs also help improve the reliability of the distribution system, improve voltage profiles, and reduce line losses and network congestion [1]. Moreover, renewable based DGs can bring about a reduction in fuel consumption and greenhouse gases. However, improper integration of DGs in the distribution system may lead to negative impacts such as poor voltage profiles, increased network losses and overloading of lines. In recent years, penetration of DGs into distribution systems has been increasing around the world. For instance, in the United States, demand growth combined with plant retirements is projected to require as much as 1.7 million GWh of additional electrical energy by 2020, almost twice the growth of the last twenty years. Over the next decade, the DG market in USA, in terms of installed capacity, is estimated to be 5 to 6 GW per year. Worldwide forecasts show that energy consumption is expected to rise by 41 per cent from 2012 to 2035, largely due to demand growth in developing countries. The projected embedded and renewable DG capacity increase associated with the global market is conservatively estimated at 20 GW per year over the next decade [2].

In Canada, widespread integration of DG in the form of wind and solar energy has been initiated in recent years. As of 2014, Canada had over 5,130 wind turbines operating on 225 wind farms for a total installed capacity of 9,694 MW, compared with only 60 wind turbines, 8 wind farms and 23 MW in 1997. Since 2004, the average annual growth rate of installed capacity of solar thermal power in Canada has been 13.8%, and that of solar photovoltaic power during 2008-2014 period was marked by significant growth, reaching a capacity of 1,843 MW in 2014 [3]. Changes in provincial and federal policies, together with new technological developments suggest that wind and solar will very likely play an increasingly important role in the future.

Due to environmental concerns and fossil fuel resource depletion, the penetration of PEVs in the transportation sector is expected to increase in the future. Many countries have set PEV penetration goals to reduce emissions and achieve energy independence. As of 2015, over than one million electric cars has been sold around the world, compared with only a few hundreds in 2005 [4]. The Electric Vehicles Initiative (EVI) sets a global target of 20 million PEVs by 2020. The Paris Declaration on Electro-Mobility and Climate Change and Call to Action sets a global target of 100 million PEVs by 2030. In Ontario, Canada, the government has set a PEV goal of 5 % of all vehicle sale by 2020 and 12% by 2030 [5]. Electrifying the transportation sector will have a potential impact on distribution systems such as increased system peak load, increased losses, deterioration in voltage profile and change in load pattern. Given that distribution networks are not inherently designed to accummodate PEV charging loads, local distribution companies (LDCs) are required to accummodate and their impacts, to decide on the right actions and policies and develop associated infrastructure.

The transition toward the smart grid has been taking place in power systems, which has a significant impact on distribution grids. A smart distribution grid involves the use of information and communication technology, system automation technologies, demandside management (DSM), integration of renewable energy sources (RES) based DG and Distributed Generation, and accommodation of electric vehicle loads. Customers play an important role in smart grids by introducing flexibility in electricity usage, in particular PEV charging loads. Therefore, new modelling strategies, and optimization techniques need be adopted in order to evaluate and incorporate these technologies in the planning process.

So far, planning of distribution systems considering all these resources, namely, DGs, capacitors, feeders and transformers, and including PEV charging loads (both uncontrolled and smart) and DR mechanisms, has not been reported in the literature. Distribution system planning should therefore be carried out in such a way that the resultant plan provides a reliable and cost effective service to customers while satisfying constraints. Considering all these resources is an important requirement of the planning process that can impact the outcomes significantly. As a result, it is important to develop and implement effective planning strategies that properly addresses these issues.

There is also a need to assess and determine the maximum allowable PEV penetration that can be accommodated by a distribution system and what percentage of the fleet can be served by uncontrolled charging and smart charging, respectively, without the need for any additional investments or upgrades. To this effect, there is a need to develop an approach to estimate the uncontrolled charging load profile using vehicle mobility data and considering different charging scenarios and the uncertainties associated with drivers' behaviour (arrival time and mileage driven), PEV market share, and share of charging level.

In the context of smart grids, the LDCs can control PEV charging demand while also considering customer preferences; such smart charging has benefits by way of deferment of the decisions on reinforcement and other investments, and maximizing the use of existing distribution infrastructure. To encourage PEV owners adopt smart charging, the LDC can offer appropriate programs and/or incentives to PEV customers. However, an LDC managed smart charging program involves a business relationship between the two parties. Assuming rational behaviour of PEV customers, their participation in smart charging programs will depend on the incentive amount, higher the incentive, more PEV customers are likely to adopt smart charging, while from the LDC's viewpoint, high incentives result in increased financial burden to itself. Adoption of smart charging is beneficial to the LDC in the short-run, because of the flattening of the overall system load profile to reduce its demand charges; and in the long-run, when deferral costs of capacity additions are taken into account.

Therefore, there is a need to develop a generic framework to determine these inter-relationships between the LDC and PEV customers, while considering their own perspectives of system operations and economic returns from such a program. There is also a need to determine the optimal participation of PEV customers that would result in the optimal benefits to both parties; and what optimal incentive would drive such an optimal participation.

1.2 Literature Review

Distribution system planning problems have been extensively researched over the years. However, these problems have undergone a change in paradigm over the past decade because of the structural transition that utilities and LDCs have gone through, advancement in tools available to researchers, changes to distribution systems, advancement in technology, and changes in policy. Therefore, distribution system planning is becoming a very complex problem. In this section, a review of the problems addressed and mathematical modeling approaches to distribution system planning is classified into three major groups as follows:

- Traditional distribution system planning
- Distribution system planning in the presence of DG
- Planning for PEVs in distribution system

1.2.1 Traditional Distribution System Planning

A comprehensive review on distribution system planning is presented in [6] wherein the planning models have been classified based on their solution techniques into optimization models and heuristic algorithms. The optimization based planning models have been further classified, based on the duration of the plan, into single-stage models and multi-stage models.

Single-stage optimization models are considered static models where the load growth is forecasted for the terminal year of the plan horizon. Such models are classified into different categories based on the particular problem addressed:

• Individual Feeders Models: The objective of this class of models is the optimal design of individual feeders, which includes configuration, length, and capacity, [7,8].

- System Feeder Models: In this class, the objective is to determine the optimal feeders routing such that the load points are served at minimum cost [9–11].
- Comprehensive Models: Substation and feeder plan decisions are optimized simultaneously in this class of models. In [12] an iterative two-stage model is used wherein, in the first stage, substation installation decisions are determined which are then used as an input to the second stage to determine the optimal feeders configurations. On the other hand, in [13], plan decisions involving new substation and feeder installation are optimized simultaneously.

Multi-stage optimization models are those where the distribution system plan is determined for the entire duration of the plan horizon. Such problems are solved either by treating the planning problem as a series of single-stage models or extending a single-stage model by adding binary and discrete selection variables and inter-temporal constraints. In the first case, each stage represents a step ahead of time, treating its output plan decisions as input to the proceeding stage. The optimal solution of each stage does not guarantee that the overall plan is optimal, since the stage does not consider the proceeding stage's input parameters. In the second case, the plan decisions for all years in the planning horizon are determined simultaneously.

Due to the large size of real distribution systems, practical life distribution planning problems can be computationally unmanageable or infeasible as the number of variables and constraints can be considerably large and very complex to solve. To overcome this problem, heuristic methods and algorithms have been proposed to simplify the planning problem into manageable and feasible ones by relaxing some constraints [14], dividing the planning problem into solvable phases or sub-problems [14–17], or transforming the dynamic optimization problem into a static one. However, there is no guarantee that the optimal solution of the simplified problem is an optimal solution for the main problem.

Driven by economic considerations, changes in policies and regulations, advancement in technology, and environmental concerns, the traditional planning strategies need be readjusted to include non-traditional options for capacity investment to address these issues. DG is one of the attractive alternative capacity options for distribution system planning.

1.2.2 Distribution System Planning in the Presence of DGs

In recent years, the integration of DG units in distribution systems has become increasingly important and the optimal DG allocation problem has attracted the interest of many researchers. Several strategies and models have been proposed to address the optimal DG sizing and siting problem. This section presents a modest attempt to survey the research literature related to DG allocation problems.

Various analytical models have been proposed for the DG allocation problem. The 2/3 rule, often used in capacitor planning studies, is proposed in [18] to allocate DG units in radial distribution systems with a uniformly distributed load. Although the model is simple and easy to implement, it is only applicable to radial systems with uniformly distributed loads. In [19], an analytical model is proposed to determine the optimal DG location in radial and networked systems by minimizing the power loss of the system.

Linear programming models have also been proposed to solve the DG allocation problem. A linear programming model in [20] is proposed to determine the optimal allocation of DG units, seeking to maximize the DG capacity. In [21], a model for planning and operation of non-firm output of DG units is proposed. Non-firm generation is the output of a DG unit that is greater than the granted allowable generation under the connection agreement between LDCs and perspective DG owners depending on the network constraint. The model aims to minimize the generators cost of non-firm access through coordinated operation.

The well-known ac optimal power flow (OPF) model is considered a powerful analysis tool in power systems. A number of models have been proposed to address the DG allocation problem using the nonlinear ac power flow. In [22], a deterministic planning model for optimal allocation of wind DG units is proposed, seeking to minimize the annual energy loss. A probabilistic model for generation and load are combined and incorporated in the deterministic planning model. The optimization model is a mixed integer non-linear programming (MINLP) model. A multi-period ac OPF model is proposed in [23] to determine the optimal DG accommodation while minimizing the system energy losses under smart grid operation. Smart control schemes such as coordinated voltage control and adaptive power factor control are considered in the proposed model.

An MINLP optimization model for determining the optimal location and number of DG units in hybrid electricity markets is proposed in [24]. Minimization of total fuel cost of conventional and DG sources and the line losses in the network comprise the objective function. The sensitivity of the results to the variation in the demand has been examined.

In the literature, heuristic techniques have also been proposed to address this problem. A genetic algorithm (GA) and e-constrained method is proposed in [25] to solve the multiobjective DG allocation problem. The cost of network expansion, cost of power losses, cost of unserved energy and cost of energy required to serve customers, comprise the multiobjective function. In [26], a multi-objective mixed integer programming model is solved using GA to maximize the LDCs benefits and hence to optimally size and site DG units within the distribution network. The deferral of investment upgrades, reduction of the cost of energy losses, and reliability improvements comprise the objective function. A GA is used to solve the DG allocation problem in [27]. The proposed model considers a multi-objective function with different load models to examine the effect of these models on the location and size of DG.

All the previous mentioned publications have only focused on DG allocation problems without considering other distribution system components, such as expansion or addition of substations and feeders. The DG allocation and traditional planning are dependent on each other and should be considered in the planning process simultaneously which will provide maximum benefits to LDCs and customers.

In [28], a comprehensive optimization model combined with planners experience is proposed to optimally allocate DG units and also determine other traditional planning options such as expanding an existing substation, adding new feeders, or/and purchasing power from neighboring LDCs via an existing intertie. The objective function aims to minimize investment and operating costs of the planning alternatives, and cost of system losses. However, this model only considers the system peak load and dispatchable DG units in the planning problem. Additionally, by considering only a peak load scenario, technical issues, such as poor voltage profile, increased network losses, or overloading the lines, might appear in other load scenarios. Moreover, reliability of the system is not considered in the planning model.

Wong et al. [29], propose a comprehensive multi-year optimization framework for distribution planning including DG units in a deregulated environment. The objective function aims to minimize the economic cost (including investment and operation cost of expanding an existing substation, adding new feeders, adding an intertie, or/and building DG units, and CO2 emissions tax. However, this model does not consider the reactive power and its associated constraints in the planning problem. Another drawback is the inaccuracy of representing the reserve of the system by a fixed percent of the peak load.

A multi-objective optimization model is proposed in [30] for distribution system expansion planning which is solved using GA. The model considers topology changes such as the installation of new switches or reconfiguring the system with existing switches, new DG units installation, rewiring of specific lines, and addition of new load points as alternatives for expansion. The uncertainties related to DG power output and load response growth are considered through the use of multiple scenarios. The costs of reliability, losses, power imported from transmission, and network investments comprise the multi-objective function. However, the model does not consider substation expansion as an expansion alternative.

In [31], an MINLP problem is proposed for multi-year distribution system planning considering DG units (natural gas generators). The objective function aims to minimize the investment costs, the operation and maintenance costs, and the cost of power losses. A GA approach is used to determine the optimal DG locations and transformer installation decisions and their installation time as a first stage and then fed into an OPF model for finding their optimum capacities. However, the DG penetration limit is assumed to be a percentage of the total load. This assumption is inaccurate since it does not consider minimum load conditions which might violate some technical limits such as maximum reverse power flow limit and equipment rating limit. Moreover, feeder expansion capacity is not considered in the proposed model. Another drawback is the inaccuracy of representing the reserve of the system by a fixed percent of the peak load and ignoring system reliability.

An optimization model for allocating different renewable DG systems is proposed in [32]. The reactive power capability of renewable DGs and the uncertainty related to the load and intermittent generation are considered in the proposed model. The optimization model is an MINLP and solved using particle swarm and ordinal optimization, minimizing the total cost. However, the model does not consider planning over years. Moreover, substation and feeder capacity expansion are not considered as an alternative option in the proposed model.

A multi-objective optimization approach for long-term distribution system planning is proposed in [33]. The model considers DG allocation, network reconfiguration, and feeder upgrades as alternatives for expansion. The proposed model is formulated as an MINLP and solved using non-dominated sorting GA. Economic and environmental objectives are considered in the proposed model through the cost of line upgrades, energy losses, switching operations, DG capital, operation and maintenance costs, and emissions. However, substation upgrade is not considered in the planning model and representing the DG units output as in terms of a fixed capacity factor of the total installed capacity is not accurate. Moreover, reliability of the system is not considered in the planning model.

In [34, 35], a multistage optimization model for distribution system planning model is proposed. The model considers expansion of existing substations, installing new ones, allocating DG units, feeder addition, load transfer between feeders; and replacement of conductors. The costs of substation and feeder installations, costs of maintenance and operation of the network and DG units comprise the objective function. However, load variation impact and reliability enhancement are not considered in this work.

When system upgrades are essentially driven by the continuously increasing demand,

capacitors can play a vital role in deferring the need for upgrades. They can improve voltage profiles, reduce feeder losses and network congestion by reducing the reactive power flows on the feeders. Coordination of capacitor and DG placement can therefore maximize the savings from loss reduction in distribution networks. Although, in the literature, there are several studies that consider the coordination of capacitors and DG placements [36–38], very few researchers have considered the comprehensive range of options such as capacitors, DGs, feeders and substations [39].

It is therefore important to investigate optimal DG placement with simultaneous placement of capacitors. Moreover, traditional planning options, such as substation and feeder addition or expansion should also be simultaneously considered which can provide maximum benefits to LDCs and customers.

1.2.3 Planning for PEVs in Distribution System

It is evident that the impact of PEV charging on the distribution grid will be significant as PEV penetration increases in the coming years. There is a large and growing body of literature that have been devoted to integrate PEVs into power system problems. The research topics mainly include impacts on distribution system operation [40–44], smart charging and discharging strategies [45–49], the interaction between PEV charging load and renewable resources [50–53]. In [54], a survey of PEV industry trends and impacts on distribution systems are presented.

A. Distribution System Planning in the Presence of PEVs

It is noted that most of the works have focussed on the impact of PEV charging loads on distribution system operations while only a few on planning [55–58]. In [55], a multi-objective planning model is presented to allocate PEV fast charging stations, substation, and feeders. While in [56], a multi-objective optimization model is proposed which considers the optimal PEV penetration, DG units and feeder upgrades in a distribution system.

In [57], a model for estimating the energy consumption of PEVs in the distribution network is presented. This model is utilized to optimally allocate DG units to mitigate the impact of high PEV penetration using a GA. While in [58], a model is presented to optimally allocate renewable DGs in the presense of PEV charging loads while minimizing the cost of capacity adequacy, network loss, and the cost of DGs' investment, operating, and maintenance. However, these models only focus on the DG allocation problem without considering other distribution system components, i.e., expansion or addition of substations and feeders.

B. Assessment of Distribution System to Accommodate PEVs

While most of the reported approaches have focussed on the impact of PEV charging loads on distribution system operations, a few studies have focused on assessing and determining whether the existing generation capacity would be sufficient for supplying the PEV charging loads. In [59,60], an optimization model is proposed for the transmission level to determine the maximum allowable penetration of PEVs in Ontario by 2025, without requiring the development of new infrastructure, considering overnight charging only. However, it is more important to assess and determine the maximum allowable PEV penetration that can be accommodated by a distribution system without the need for any additional investments or upgrades rather than assessing the system overall generation capacity.

In the context of distribution systems, an optimization model is proposed in [61] to determine the maximum penetration level of PEVs. The impact of three charging scenarios namely, uncontrolled charging, loss optimal, and price optimal, are considered to assess PEV accommodation in distribution systems. However, the uncertainty of different parameters such as drivers' behaviour, mileage driven, and demand have not been considered. In addition, PEV charging loads are expected to be a mix of controlled (smart) and uncontrolled charging, which is not considered [61].

So far, the assessment of Distribution System Loading Margin (DSLM) to accommodate PEV charging loads under a number of possible charging scenarios while considering both uncontrolled and smart charging, has not been reported in the literature.

C. Planning for PEV Smart Charging in Distribution Systems

The transition towards the smart grid has been taking place in power systems, which has a significant impact on distribution grids. A smart distribution grid involves the use of information and communication technology, system automation technologies, DSM, integration of RES and DG, and accommodation of PEV charging loads.

In the current context of smart grids, the LDCs can control PEV charging demand while also considering customer preferences. By planning for PEVs, LDCs can defer the decisions on reinforcement and other investments, and maximize the use of existing infrastructure. In addition, LDCs can establish rate structures that incentivize the use of smart charging and increase the adoption and use of PEVs. Therefore, electrifying the transport sector can benefit both the LDCs and the customer.

There is a growing body of literature on smart charging strategies of PEVs and their role in DR programs [42, 50–53, 62–69]. Studies have focused on PEV smart charging for frequency regulation [62, 63], load shaping [64], alleviating transformer overloads [65, 66], feeder congestion management [42, 67], distribution system reliability [70], and interaction between RES and PEV charging loads [50–53, 63, 68]. While other studies have examined the potential economic benefits of smart charging strategies [63, 69].

A number of studies have focused on incentive design to encourage PEV customers to participate in DR programs [71–75]. In [71], a smart charging algorithm is proposed where PEVs can inject power to the grid for frequency regulation or absorb power by utilizing renewable energy. In addition, an economic analysis is performed to evaluate the benefits of the proposed model. An incentive scheme is proposed in [72] for PEV battery exchange stations to participate in DR programs to reduce the peak to average ratio. In [73], an incentive mechanism is proposed to allow an LDC to schedule PEV charging on a day-to-day basis. An energy pricing scheme is proposed in [74] to control the PEV charging loads, where the longer a PEV customer is willing to defer its charging, the larger is its reduction in energy price. Real-time control of PEV charging loads is proposed in [75] by generating an incentive signal to PEV users in order to minimize the cost of electricity supply.

Utility managed smart charging programs involve a business relationship between the LDC or aggregator and the PEV customers. None of the reported works have considered this relationship in smart charging programs vis-a-vis the promotional incentives offered by the LDC. The optimal penetration of smart charging, arising from such inter-relationships and incentive mechanisms, from a long-term perspective, have not been investigated.

In addition to the need to determine optimal penetration of smart charging into distribution systems, there is also a need to examine the participation rate of customers in LDC smart charging programs.

1.3 Research Objectives

The main objectives of this research can be summarized as follows:

• Develop a comprehensive, multi-year, distribution planning framework that simultaneously determines the optimal sizing, placement and investment timelines of various resources for LDCs such as DGs, substations, capacitors and feeders, as well as ensuring an adequate and reliable distribution system in the long-term. In addition, the work considers PEV uncontrolled and smart charging loads as well as DR as a plan option to mitigate the growing impact of PEV charging loads.

- Develop a generic framework to assess the DSLM to accommodate PEV charging loads considering uncontrolled and smart charging without the need of any additional investments. In addition, using vehicle mobility data, develop an approach to estimate the uncontrolled PEV charging load profile while considering different charging scenarios, as well as the uncertainties associated with drivers' behaviour (arrival time and mileage driven), PEV market share, and share of charging level.
- Develop a generic framework to simultaneously determine the optimal participation of PEV customers in the smart charging program and the optimal incentives to be offered by the LDC, taking into consideration LDC-customer interactions and the distribution system operational aspects. The framework builds upon two aspects, the relationship between the incentives offered by the LDC and the participation of PEV customers; and the relationship between the participation of PEV customers and the long-term economic benefit of capacity deferral accrued from smart charging.

1.4 Thesis Outline

The remainder of this thesis is organized as follows:

Chapter 2 reviews the background of distribution systems, distribution system planning models, impact of DG on distribution systems, DG planning models, and PEV characteristics and their impacts on distribution system.

Chapter 3 presents the proposed distribution system planning model, the heuristic back-propagation approach, PEV charging load modelling, and results.

Chapter 4 presents a novel framework to assess the DSLM to accommodate PEV charging loads considering uncontrolled and smart charging under different charging scenarios, vehicle types, and charging levels.

Chapter 5 presents a generic and novel framework to assess the optimal participation of PEV smart charging and the optimal incentive paid to customers under different vehicle types, and charging levels.

In Chapter 6, summary and conclusions, main contributions, and directions for future research work.

Chapter 2

Background

2.1 Distribution Systems

The power system can mainly be divided into generation, transmission, sub-transmission and distribution. Traditionally, the generation systems task is to produce electricity, the transmission system is responsible for the delivery of power from generating stations to the sub-transmission system, at voltage levels of 230 kV or higher. Then, the sub-transmission system transmits the power at voltage levels between 69 kV - 138 kV to the distribution systems. Finally, the distribution systems deliver electricity to the customers at voltages typically under 34.5 kV [76]. Figure 2.1 illustrates a typical electricity supply system. The primary distribution system comprises of distribution substations and feeders. In the distribution substations, the voltage is stepped down from the sub-transmission system to between 34.5 kV and 4.16 kV. The main feeders in the primary distribution system branch out from the substation and then as lateral feeders to In secondary distribution system, the voltage is reduced to the serve local areas. customers level via distribution transformers, generally at 120/240 V and 480 V (Figure 2.2).

2.2 Distribution System Planning

Distribution system design and planning seeks the best expansion plan to provide reliable and economic services to meet the customers load demand in the long-term horizon considering the predicted load growth. In the recent years, distribution systems are



Fig. 2.1. Basic power system structure

undergoing a change in paradigm because of the transformation of the grid to a smart grid which is expected to drive distribution system planning in the coming years. In classical planning, the load growth typically is met by adding a new substation or upgrading the existing substation capacity along with their feeders. Today, with rapid penetration of DR, and DG, particularly intermittent sources such as wind and solar, ESS, and PEVs, it is important to develop effective planning strategies for the evolution of a smart distribution grid.

2.2.1 Traditional Distribution System Planning

Distribution system planning is essential to assure that the distribution network is capable to serve the growing electricity demand economically and reliably. In traditional planning, the load growth is typically met by adding a new substation or upgrading the existing substation capacity along with their feeders. Distribution system planners endeavor to determine the best expansion strategies to provide reliable and economic services to the customer. In the earlier years, the research focused on traditional planning problems such as the placement of substations and routing of feeders to minimize costs and losses to the LDC. Generally, traditional distribution system planning decisions can be classified according to the distribution system component, as follows:

• Substation: their selection can be based on multiple factors such as load growth, load density, and land availability; the planning problem determines the optimal new



Fig. 2.2. Typical configuration of a radial distribution system

substation location and capacity, optimal expansion capacity for existing substation, or optimal mix of transformers.

- Feeders: the planning problem determines their optimal routing, optimal upgrade capacity, or optimal individual feeder design. These decisions are required in combination with a new substation installation, the expansion of existing one, or independently if there is sufficient capacity in the existing substation. The design of the feeder must comply with the system operational constraints.
- Optimal load allocation: this includes load shedding or/and load transfer between substations.

These plan decisions are affected and constrained by different factors, such as:

- Capacity constraints: includes substation transformers capacities, and feeders thermal limits
- Operating constraints: such as demand supply balance, power flow, and voltage limits.

• Budget constraints: this constraint imposes a limit on how much capacity the LDC can invest in, over the plan period by imposing a limit on capital expenditure.

Distribution system design and planning involves four major activities which starts with load forecasting, and are discussed below with reference to Figure 2.3.

- Load Forecast: A detailed spatial future demand is forecasted. The inputs to this task includes population growth, load density, historical data, city plans, alternative energy sources, etc.
- Performance Check: Using the outcomes from the first step, performance analysis is carried out considering the LDCs policies and obligations to customers and involves tests for service continuity, voltage drop, maximum peak load capacity, reliability, power losses, etc. These analyses requires the use of tools such as load flow, voltage drop calculation, short circuit and fault calculation, etc. Based on the performance results, the planner determines whether the existing system is capable of handling the forecasted demand.
- Substation Planning: It starts with the identification of all possible alternatives ranging from improving an existing substation to build a new substation. Size of the substation, number of transformers, and their siting are determined for each alternative. After that, an evaluation of all the alternatives is carried out in order to find the best alternative that meets all the technical constraints at minimum cost.
- Feeder Planning: Once the substation decision is obtained, feeder planning involves feeder route selection, optimal number of feeders, and conductor sizing followed by calculating the total cost of the candidate plan and determining whether the plan total cost is within the budget. If it is accepted, then the final plan is obtained. If not, the planner has to select the second best alternative and repeat the same procedure. The total cost must contain all components such as equipment and site costs, maintenance and operation costs, cost of loss, and taxes.

2.2.2 Distribution System Planning in the Presence of DGs

Till now, there is no universal agreement regarding the definition of a DG. The general definition for DG commonly used in the literature is:



Fig. 2.3. Traditional distribution planning framework

Electric power generation within distribution networks or on the customer side of the network [1].

DGs can be classified based on their size [1]. The DG unit capacity ranges from 1 kW solar PV cells to 1 MW engine generators to 1000 MW offshore wind farms [77]. Typically, the classification is as follows:

- Micro DG: 1 W 5 kW
- Small DG: 5 kW 5 MW
- Medium DG: 5 MW 50 MW
- Large DG: 50 MW 300 MW

DGs have multifarious benefits on the distribution network when installed properly. DG units based on renewables can reduce environmental emissions. DG units can also have a beneficial impact on power quality and reliability such as improved voltage profile, reduced line losses and network congestion [78]. These can also help in the deferral of new transmission lines and large power plants, reduce system operating costs with increased penetration of renewable energy and increased overall efficiency. DGs also have the potential to increase competition in generation which can lead to better service and lower energy price.

Distribution systems were not planned originally to accommodate DG units, therefore these sources have to be carefully sited with appropriately sized. Improper integration of DG sources may have negative impacts on distribution systems such as line overloading, increased power loss, overvoltage, and increased short circuit current levels. In addition, the nature of renewable DGs and the uncertainty related to their output adds more complexity and challenges to the planning problem.

Over the last decade, the integration of DG units in distribution systems has become increasingly important. Therefore, the optimal DG allocation problem has attracted the interest of many research efforts. Several strategies and models have been proposed to address the DG allocation problem. Generally DG planning can be considered a single or multi-objective planning problem; the more commonly used objectives, decision variables, and related constraints are listed in Table 2.1 [78]:

Objective function	Decision variables	Constraints
System loss minimization	DG Location	Discrete size of DG units
Voltage limit loadability maximization	DG Size	Short-circuit level limit
Cost minimization	Type of DG technology	Limited buses for DG installation
Voltage deviations minimization	Number of DG units	Power generation limits
DG penetration maximization		Budget limit
Profit maximization		DG with constant power factor
Benefit/cost ratio maximization		DG penetration limit
System average interruption duration		Total harmonic voltage distortion limit
index (SAIDI) minimization		Maximum number of DG units
		Reliability constraints, e.g., max
		SAIDI

TABLE 2.1 Objectives, decision variables, and constraints for DG planning problems

2.2.3 Supply Adequacy of Distribution Systems

Power system reliability can be defined as the ability of a generation system to adequately and securely supply electrical energy to its end-customers. Power system reliability analysis involves system adequacy and security. Adequacy generally refers to ensuring sufficient supply capacity in the system to meet the demand considering long-term load growth, generator outage rates and maintenance schedules. Security is the ability of the system to respond to short-term disturbances such as the unexpected loss of a major generating plant. Generally, adequacy assessment is carried out considering system steady-state conditions, whereas security evaluation considers the system dynamics during disturbances [79]. In this thesis, reliability assessment of the distribution system is limited to adequacy assessment; hence, the terms reliability and adequacy are used interchangeably.

At the planning stage, it is necessary to determine the amount of supply capacity that need to be installed to satisfy the predicted electricity demand. The main criteria for supply adequacy assessment for power systems is presented below [79,80].

A. Deterministic Techniques

Deterministic techniques for adequacy assessment were first developed and used by many power utilities to determine the required generating capacity. These techniques seek the optimal installed supply capacity considering a fixed percentage reserve such as a fixed reserve equal to the largest generating unit, a fixed percentage of the total installed capacity or the peak load, or a mix of these. The main drawback of these techniques is that they are unable to respond to the actual stochastic nature of power systems that result from customer demand fluctuations or component failures [79].

B. Probabilistic Techniques

The probabilistic criteria considers the stochastic nature of power systems that result from customer demand fluctuations or component failures to evaluate the system risk state [81]. Two main probabilistic approaches exist for evaluating the reliability of power systems; analytical and simulation based, often known as Monte Carlo simulation (MCS). The analytical approach uses a mathematical model to represent system elements and their outage rates to evaluate system reliability indices. The MCS technique, on the other hand, estimates reliability indices by simulating the actual process and random behavior of the system. Each category of methods have their advantages and disadvantages, so the appropriate method is chosen based primarily on the type of evaluation desired and the nature of the problem.

Evaluating the adequacy of a distribution system is achieved by examining its main components; the supply side and the load model. The two models are then combined to evaluate and obtain the reliability indices (Figure 2.4).



Fig. 2.4. Framework for adequacy assessment in distribution system

1. Supply Side (DGs and Substation) Model in Probabilistic Methods

A DG or substation transformer is represented by a two-state model (fully rated state or failed state) [79]. The forced outage rate (FOR) is the probability of finding the DG or the substation transformer in an outage state and is measured by their unavailability (U). Usually, FOR is calculated based on the historical outage data of these components, if the failure rate and the repair rate are known, FOR can be calculated as follows.

$$FOR = U = \frac{\sum T_{down}}{\sum T_{up} + \sum T_{down}} = \frac{\lambda}{\lambda + \mu}$$
(2.1)

The basic outage characteristic used in the probabilistic techniques is a Capacity Outage Probability Table (COPT). The COPT is represented by an array of capacity levels with their associated probabilities of not serving the load.

2. Load Model in Probabilistic Methods

The load model is a representation of the systems energy demand over a specific period of time [79]. A variety of load models have been utilized for evaluating the adequacy of the supply side capacity. The simplest load model considers the peak load of the system as, fixed, for the entire study period. The hourly peak load variation curve, or the load duration curve is frequently used to represent the actual load model. It is created by arranging the individual hourly peak loads in descending order.
3. Reliability Evaluation Indices

Power system reliability is usually assessed by indices that measure the reliability and adequacy. These indices include Loss of Load Probability (LOLP), and Loss of Load Expectation (LOLE) [80].

LOLP: is the probability that the load will exceed the available generation.

LOLE: is the expected average number of hours during which the load is expected to exceed the existing generating capacity, as given below:

$$LOLE = \sum_{k}^{n} t_k p_k \qquad (L > C) \tag{2.2}$$

where

n : number of capacity outage states in the COPT

 p_k : probability of capacity outage

 t_k : time duration for which the loss of load will occur due to capacity outage

- L : load level
- C : capacity outage states in COPT

2.2.4 Demand Response

Demand response is one of the various options within the broad scope of demand side management. It is the determination of programs to manage the customer load demands and achieve least-cost system operation [82]. DR programs are classified into two broad types, incentive-based programs and time-based programs.

Incentive-based programs

In these types, customers provide load reductions when needed and receive incentives or direct payments. Generally, there are five subtypes of incentive-based programs, as follow:

• *Direct load control:* Operating and managing end-use devices, such as air-conditioning, pool pumps, or PEV charging. This type is considered one of the most common used programs. Typically, residential and small commercial customers are targeted in this type of program.

- *Demand buyback:* Customers choose to curtail upon request for event. Commercial and industrial customers are targeted in this type of programs.
- *Demand bidding:* Customers bid load reduction into utility or market on advance basis. This type is considered a variation of demand buyback program.
- *Interruptible rate:* discounted rate or credits for large industrial or commercial users that are willing to curtail operations. This type of programs is attractive to customers that have the ability to temporarily reduce, shut down, or shift their loads.
- Ancillary-services market: Customers, in this program, receive payment for agreeing to fast response to reduce load when requests.

Time-Based Programs

In these types, electricity prices are set for a specific time period on a forward basis. To apply time-based programs, there is a need for an advanced meters to records reads in time steps. Generally, there are three subtypes of time-based programs, as follow:

- *Time-of-use pricing:* Prices set for a specific time period on a forward basis. Based on the expected average real-time prices, TOU prices are set for each part of the day.
- *Critical peak pricing:* Established time-of-use prices in effect except for critical peak load days or hours, on which a critical peak price is in effect.
- *Real-time pricing:* In this type, electricity prices vary based on the market. The prices are changing every hour based on the real-time system conditions.

2.3 Plug-in Electric Vehicles (PEVs)

Fossil fuel depletion and environmental concerns, are factors leading to electrifying the transportation sector through PEVs. The PEV, as defined by IEEE, has a battery storage system of 4 kWh or more, a means of recharging the battery from an external source, and the ability to drive at least 10 miles in all electric mode [83]. These vehicles will have a significant impact on the existing power systems.

Electric vehicles can be classified into three main classes [84]:

• Battery Electric Vehicles (BEV)

BEVs use batteries to store the energy that will be transformed into mechanical power by electric motor(s) only, without the use of internal combustion engine (ICE). The battery is the only source of energy which as recharged from the grid.

• Hybrid Electric Vehicles (HEV)

In HEVs, propulsion is the result of two energy sources where it combined actions of electric motor, battery and ICE. The battery is recharged by utilizing vehicle's kinetic energy lost while braking.

• Plug-in Hybrid Electric Vehicles (PHEV)

PHEVs are essentially a combination of BEV and HEV, having the all-electric capability of a BEV in urban areas and a smaller onboard ICE for extended range capability as an HEV. PHEVs utilizes the battery energy during a trip until the battery gets depleted and then it switches to the ICE as the main energy source. The battery can be recharged from the grid.

It is to be noted that PEV is a general term for vehicles that is recharged from the electricity grid which includes BEV and PHEV. In this thesis, only BEV and PHEV are considered since their batteries are recharged from the grid which will have a potential impact to the electricity grid.

2.3.1 PEV Load Characteristics

The impact of PEVs, represented by a significant new load in the distribution network, should be considered in the planning process. The main parameters that defines PEV load is discussed below [54]:

• Driving patterns:

Driving patterns or the behavior of PEV owners is an important factor that determines where, and how many PEVs will be charged.

• Charging characteristics:

Charging characteristics determines the amount of load added to the base load and the total duration of the charge. The battery capacity, the state-of-charge (SOC), charging level, miles driven and charging efficiency are the main parameters that define PEV charging characteristics. • Time of charge:

Another important parameter to consider is when vehicles will be recharged. The timing of charge should consider the season of the year, type of day, i.e., weekdays, weekend, or holiday, and the time of the day.

• Penetration level of PEVs:

The penetration level is an important factor that may have a significant impact on the distribution system demand. Based on realistic statistics, customers acceptance, government and vehicle manufacturer trends, policies and regulations, the market share and the penetration level of PEVs need be determined for the studied area.

2.4 Charging level

The SAE J1772 standards defines two residential charging levels (Level-1 and Level-2) with a third ac level and a dc level for PEVs [54], as presented in Table 2.2.

Type	Power Level
Level 1: 120 VAC	1.2 - 2.0 kW
Level 2 (low): 208-240 VAC	2.8 - 3.8 kW
Level 2: (high): 208-240 VAC	6 - 15 kW
Level 3: 208-240 VAC	>15 kW-96kW
Level 3: DC Charging: 600VDC	>15 kW-240 kW

TABLE 2.2 PEV Charging Level Characteristics

2.5 Concluding Remarks

In this chapter, three topics have been briefly discussed: distribution systems, distributed system planning, and PEVs. In the first section, a brief background of distribution system is presented. In the second section, traditional distribution system planning, and distribution system planning in the presence of DERs and DGs are discussed. This section also includes a discussion on supply adequacy of distribution systems and demand response. Thereafter, a discussion on PEV types, mode of operations and PEV load characteristics is presented.

Chapter 3

Distribution System Planning to Accommodate Distributed Energy Resources and PEVs¹

3.1 Introduction

With deregulation of the power industry, environmental policy changes, advancements in technology, and the transformation to smart grid, the distribution planning paradigm has gone through significant changes in recent years. Concurrently, with increase in gas prices, driven by a foreseeable fossil fuel depletion in the future, developments in the automotive sector, and environmental concerns, penetration of PEVs has been increasing. These changes will continue to drive the distribution planning problem to evolve in the coming years.

This chapter presents a comprehensive long-term distribution planning framework from the perspective of LDCs considering DG, substations, capacitors, and feeders.

¹This chapter has been published in: A. Bin Humayd, and K. Bhattacharya. "Distribution system planning to accommodate distributed energy resources and PEVs." Electric Power Systems Research, 145 (2017),1-11.

Earlier versions of the work has been published in: A. Bin Humayd, and K. Bhattacharya. "Comprehensive multi-year distribution system planning using back-propagation approach." IET Generation, Transmission and Distribution, 7(12), 1415-1425.

A. Bin Humayd, and K. Bhattacharya. "Impact of PEV penetration on distribution system planning considering time-of-use electricity prices." IEEE PES General Meeting, 2014, National Harbor, MD (Washington, DC Metro Area).

Apart from considering the usual demand profile, the proposed framework considers uncontrolled and controlled (smart) PEV charging demand, as well as DR options. Based on a back-propagation algorithm combined with cost-benefit analysis, a novel approach is proposed to determine the optimal upgrade plan, allocation, and sizing of the selected components in distribution systems, to minimize the total capital and operating cost. A new iterative method is proposed which involves post-processing the plan decisions to guarantee acceptable adequacy levels for each year of the planning horizon. The performance of the proposed framework is examined considering several case studies on the 33-bus and 69-bus test systems.

The structure of the chapter is as follows. The nomenclature used in this chapter is presented in Section-2.7. In Section-2.9, the mathematical model of the proposed multi-year planning problem is presented and this is followed by a description of the proposed solution approach in Section-2.10. In Section-2.11, the two test systems considered for analysis are described and the results are presented in Section-2.12. Conclusions are drawn in Section-4.6.

3.2 Nomenclature

Indices

b	Index for load block, $b = 1, 2,B$
i, j	Index for buses, $i = 1, 2,N$
N	Total number of system buses
SS	Subset of substation buses $(SS \in i)$
t	Index for year, $t = 1, 2,T$

Parameters

BL	Budget limit, \$
C^C	Capital cost of capacitor, \$/p.u.
$C^{DG.F}$	Capital cost of DG unit, \$/p.u.
$C^{DG.O}$	Operating cost of DG units, \$/ p.u.
C^{DR}	Payment toward responsive demand, $/ p.u.$

$C^{Fdr.F}, C^{Fdr.V}$	Fixed and variable component of capital cost of feeder, and $/ p.u$.
$C^{SS.F}, C^{SS.V}$	Fixed and variable component of capital cost of substation, and $/ p.u.$
C^{UN}	Cost of unserved energy, \$/ p.u.
CBL	Capacitor budget limit,\$
DG^{CapMax}	Maximum allowable DG capacity, p.u.
DG^{NMax}	Maximum number of installed DG units
E	Energy needed to charge a PEV, p.u.
$Ge_{i,j}$	Geographic cost factor of feeder i - j , \$
Hr_b	Hours per day in load block b
$Le_{i,j}$	Length of feeder between i and j , km
M	Big number used in MIP modelling
NH_i	Total number of houses at bus i
$P^{PEV_{Cap}}$	Maximum allowable power drawn by PEV during charging, p.u.
$P_{i,i,b}^{Fdr}, Q_{i,i,b}^{Fdr}$	Active and reactive power flow from i to j , p.u.
$P_{i,b}^{PEV_{UNC}}$	Power drawn by PEV in uncontrolled mode, p.u.
$Pd_{i,b}, Qd_{i,b}$	Active and reactive power demand, p.u.
$PEV^{\%_{pen}}$	The percentage of PEV penetration
$Q^{C_{Max}}$	Maximum capacitor size, p.u.
$S^{Fdr_{Cap}}_{i,j}$	Existing feeder capacity, p.u.
$S^{SS_{Cap}}, P^{SS_{Cap}}$	Existing substation capacity, p.u.
V^{Min}, V^{Max}	Minimum and maximum allowble voltage, p.u.
$Y_{i,i}$	Magnitude of admittance matrix element, p.u.
γ	Capacity reserve margin with respect to Base Case peak load
$\dot{\theta}_{i,i}$	Angle of bus admittance matrix element, rad
ρ_b	Electricity market price, \$/ p.u.
α_0	Share of demand that is available for DR, p.u.
-	/ ±

Variables

ion, p.u.
ion,

$Q_i^{C_{Cap}}$	Capacity of shunt capacitor, p.u.
$Q_{i,b}^C$	Reactive power injected by capacitor i , p.u.
$S_{i,i}^{NFdr}$	Capacity added to feeder $i - j$, p.u.
$S_{ss}^{NSS}, P_{ss}^{NSS}$	Capacity added to substation, p.u.
$S_{ss,b}$	Apparent power imported via substation, p.u.
$S_{i,b}^{UN}$	Unserved apparent power, p.u.
$V_{i,b}$	Voltage magnitude at bus i and load block b , p.u.
z_i^{DG}	Binary decision on DG investment, $(0/1)$
$z_{i,j}^{Fdr}$	Binary decision on feeder upgrade, $(0/1)$
$z_{ss}^{\ddot{S}S}$	Binary decision on substation upgrade, $(0/1)$
$\theta_{i,i,b}^{Fdr}$	Power angle of the flow from i to j , rad
$\theta_{ss,b}^{SS}$	Power angle of the power imported by substation, rad
$\delta_{i,b}$	Voltage phase angle at bus i and load block b , rad

3.3 Proposed Framework For Distribution System Planning

Fig. 2.5 presents the overall schematic of the proposed framework for multi-year distribution system planning which comprises two main levels. The proposed approach is based on a back-propagation algorithm starting from the terminal year and arriving at the first year. The proposed approach uses a bi-level procedure as described next. In Level-1 (ADEQ-OPTSELECT), the determination of the optimal size and location of distribution system upgrades that are required to be in place, at the plan terminal year, as well as, an appropriate reserve margin (γ). While in Level-2 (ADEQ-OPTPERIOD), determination of the optimal period of commissioning, for the upgrades selected in Level-1.

A back-propagation of model solution, starting from the plan terminal and ending at the first year, is proposed. Within this framework, Two mathematical models are proposed, namely Smart-DSPLAN and Smart-DSPLAN2, in level-1 and 2 respectively. In addition, Benefit-to-Cost Ratio (BCR) and a novel adequacy check are introduced that involves post-processing the plan decisions at both the levels to ensure that plan is beneficial and the target plan adequacy level is satisfied for each year.



Fig. 3.1. Proposed framework

3.4 Smart Distribution System Planning Model

3.4.1 Smart-DSPLAN

The first proposed generic mathematical model for long-term distribution planning, considering the comprehensive set of decisions and issues discussed earlier, is referred to as, Smart-DSPLAN, and is presented below.

Objective Function

The objective function (J) aims to minimize the annualized cost of capital and operation of the LDC.

$$J = \sum_{i \in N} \left(C^{DG.F} P_i^{DG_{Cap}} + \sum_{b \in B} (C^{DG.O} P_{i,b}^{DG} Hr_b) \right) + \sum_{i \in SS} \left(C^{SS.F} z_i^{SS} + C^{SS.V} S_i^{NSS} + \sum_{b \in B} (\rho_b P_{i,b} Hr_b) \right) + \sum_{i,j \in N: \exists (i,j)} \left(C^{Fdr.F} Ge_{i,j} Le_{i,j} z_{i,j}^{Fdr} + C^{Fdr.V} S_{i,j}^{NFdr} \right) + \sum_{i \in N} \left(C^C Q_i^{C_{Cap}} \right) + \sum_{i \in N, b \in B} \left(C^{DR} P_{i,b}^{DR} Hr_b \right)$$
(3.1)

The first line of (2.3) includes the capital and operating cost of the candidate DG units. The second line includes the engineering, procurement, and construction (EPC) cost and the variable component of the capital cost to upgrade the substation, and payment toward purchased power by the LDC. The third line represents the EPC cost, and the variable component of the capital cost to upgrade the feeders, the last line denotes the capital cost of candidate capacitors, and the payment made by the LDC to DR customers. The associated operational and planning constraints are discussed next.

Power Flow Equations

The injected power at a bus is the power from the substation and DG units, net of the load, uncontrolled and smart PEV charging loads, and DR; and is governed by the traditional ac power flow equations:

$$P_{i,b} + P_{i,b}^{DG} - P_{i,b}^{PEV_S} - Pd_{i,b} + P_{i,b}^{DR} = \sum_{j \in N} V_{i,b} V_{j,b} Y_{i,j} \cos(\theta_{i,j} + \delta_{j,b} - \delta_{i,b}) \qquad \forall i, b$$
(3.2)

$$Q_{i,b} + Q_{i,b}^C - Qd_{i,b} = -\sum_{j \in N} V_{i,b} V_{j,b} Y_{i,j} \sin(\theta_{i,j} + \delta_{j,b} - \delta_{i,b}) \qquad \forall i,b$$
(3.3)

Feeder Capacity Limits

Power flow through any distribution feeder must comply with the thermal capacity of the feeder. This limit also takes into consideration the new investments in feeder upgrades.

$$-V_{i,b}^{2}Y_{i,j}\cos\theta_{i,j} + V_{i,b}V_{j,b}Y_{i,j}\cos(\theta_{i,j} + \delta_{j,b} - \delta_{i,b}) \leq (S_{i,j}^{Fdr_{Cap}} + S_{i,j}^{NFdr})\cos\theta_{i,j,b}^{Fdr} \quad \forall (i,j) \in N : \exists (i,j), \forall b$$

$$(3.4)$$

$$V_{i,b}^{2}Y_{i,j}sin\theta_{i,j} - V_{i,b}V_{j,b}Y_{i,j}sin(\theta_{i,j} + \delta_{j,b} - \delta_{i,b}) \leq (S_{i,j}^{Fdr_{Cap}} + S_{i,j}^{NFdr})\sin\theta_{i,j,b}^{Fdr} \quad \forall (i,j) \in N : \exists (i,j), \forall b$$

$$(3.5)$$

$$S_{i,j,b}^{NFdr} \le M z_{i,j}^{Fdr} \qquad \qquad \forall (i,j) \in N : \exists (i,j), \forall b \qquad (3.6)$$

In (2.8), M is a sufficiently large number often called the "big M", which renders the constraint (2.8) nonbinding. When $z_{i,j}^{Fdr} = 0$, $S_{i,j}^{NFdr}$ is zero while when $z_{i,j}^{Fdr} \neq 0$, the large value of M allows sufficient room for selection of new feeder capacity, and the value of M ensures that (2.8) is satisfied. The binary variables $z_{i,j}^{Fdr}$ act as a "switch" on the continuous constraints, appearing with a large number greater than or equal to the maximum allowable feeder upgrade capacity.

Substation Capacity Limits

These constraints ensure that the total power delivered by the substation transformer is within the substation capacity limit. These limits take into consideration new investments in substation upgrades.

$$Q_{ss,b} \le (S^{SS_{Cap}} + S^{NSS}_{ss}) sin(\theta^{SS}_{ss,b}) \qquad \forall b \qquad (3.7)$$

$$P_{ss,b} \le (S^{SS_{Cap}} + S^{NSS}_{ss})cos(\theta^{SS}_{ss,b}) \qquad \forall b \qquad (3.8)$$

$$S_{ss}^{NSS} \le M z_{ss}^{SS} \tag{3.9}$$

DG Capacity Limits

The power generated by a DG unit is limited by the DG capacity (2.12). The installed capacity of DG is limited by the maximum allowable DG size (2.13). Constraint (2.14) limits the number of allowable DG units. The maximum DG penetration is limited by the minimum load (including PEV charging load) plus 60% of maximum substation rating in order to limit the maximum reverse power flow over the transformer, as given in (2.15) [85].

$$P_{i,b}^{DG} \le P_i^{DG_{Cap}} \qquad \forall i,b \tag{3.10}$$

$$P_i^{DG_{Cap}} \le DG^{CapMax} z_i^{DG} \qquad \forall i \tag{3.11}$$

$$\sum_{i \in N} z_i^{DG} \le DG^{NMax} \tag{3.12}$$

$$\sum_{i \in N} P_{i,b}^{DG} \le \sum_{i \in N} (Pd_{i,b} + P_{i,b}^{PEV}) + 0.6P_{i,b} \qquad \forall b$$
(3.13)

Capacitor Limits

These constraints ensure that the installed capacitor size is limited by a specified maximum allowable capacitor size (2.16). The reactive power injected by a capacitor must be less than the installed capacity (2.17).

$$Q_i^{C_{Cap}} \le Q^{C_{Max}} \qquad \forall i \tag{3.14}$$

$$Q_{i,b}^C \le Q_i^{C_{Cap}} \qquad \forall i,b \tag{3.15}$$

PEV Smart Charging Constraints

These constraints determine the optimal PEV charging schedule by ensuring that the total energy required by PEVs is equal to their daily energy needed to charge the battery (E); and the power drawn by PEVs is within the charging level. It is to be noted that E is calculated based on the average mileage driven by a vehicle as determined from the National Household Travel Survey data (NHTS) [86].

$$\sum_{b \in B} P_{i,b}^{PEV} = NH_i \cdot E \cdot PEV^{\mathcal{H}_{Pen}} \qquad \forall i \qquad (3.16)$$

$$P_{i,b}^{PEV} \le NH_i \cdot P^{PEV_{Cap}} \cdot PEV^{\mathscr{H}_{Pen}} \qquad \forall i,h \qquad (3.17)$$

Capacity Adequacy Limit

This constraint ensures the installation of enough capacity in the system so that supply can be maintained during peak hours in case of resource failure.

$$(S^{SS_{Cap}} + S^{NSS}_{ss})cos(\theta^{SS}_{ss,peak}) + \sum_{i \in \mathbb{N}} P^{DG_{Cap}}_i \ge (1+\gamma)\sum_{i \in \mathbb{N}} (Pd_{i,peak})$$
(3.18)

DR Constraints

These constraints impose a limit on the DR at each time block and bus in the system.

$$P_{i,b}^{DR} \le \alpha_0 P d_{i,b} \qquad \forall i,b \qquad (3.19)$$

Voltage Limits

These constraints ensure that the voltage magnitude at a bus is within the allowable limits.

$$V^{Min} \le V_{i,b} \le V^{Max} \qquad \forall i,b \qquad (3.20)$$

Budget Limit-General

This constraint imposes a limit on the capital spending and consequently how much capacity the LDC can invest. The first term of (2.23) is the capital cost of DG units, the second term is the EPC cost and the variable component of the capital cost to upgrade the substation. The third term is the EPC cost and the variable component of the capital cost to upgrade the feeders. The capital cost of capacitors is presented in the last term. All these costs together, must be within the budget limit, BL.

$$\sum_{i \in N} \left(C^{DG.F} P_i^{DG_{Cap}} \right) + \sum_{i \in SS} \left(C^{SS.F} z_i^{SS} + C^{SS.V} S_i^{NSS} \right)$$

+
$$\sum_{i,j \in N: \exists (i,j)} \left(C^{Fdr.F} Ge_{i,j} Le_{i,j} z_{i,j}^{Fdr} + C^{Fdr.V} S_{i,j}^{NFdr} \right) + \sum_{i \in N} \left(C^C Q_i^{C_{Cap}} \right) \leq BL$$
(3.21)

Budget Limit on Capacitor Banks

This constraint imposes a limit on spending that the LDC can make on new capacitor banks.

$$\sum_{i \in N} C^C Q_i^{C_{Cap}} \le CBL \tag{3.22}$$

In it to be noted that the capital cost of capacitor banks are relatively smaller compared to that of DG units, and if a separate budget limit for capacitors is not imposed, the planning model will tend to over-select the number of capacitors, at the cost of DG units, which is undesirable. To alleviate this problem, a separate budget is therefore allocated to capacitor banks, without any loss of generality.

3.4.2 Smart-DSPLAN2

The second proposed mathematical model, referred to as, Smart-DSPLAN2, is presented below.

The objective function (J2) aims to minimize the annualized cost of capital and operation of the LDC.

$$J2 = \sum_{i \in N} \left(C^{DG.F} P_i^{DG_{Cap}} + \sum_{b \in B} (C^{DG.O} P_{i,b}^{DG} Hr_b) \right) + \sum_{i \in SS} \left(C^{SS.F} + C^{SS.V} S_i^{NSS} + \sum_{b \in B} (\rho_b P_{i,b} Hr_b) \right) + \sum_{i,j \in N: \exists (i,j)} \left(C^{Fdr.F} Ge_{i,j} Le_{i,j} + C^{Fdr.V} S_{i,j}^{NFdr} \right) + \sum_{i \in N} \left(C^C Q_i^{C_{Cap}} \right) + \sum_{i \in N, b \in B} \left(C^{DR} P_{i,b}^{DR} Hr_b \right) + \sum_{i \in N, b \in B} \left(C^{UN} P_{i,b}^{UN} Hr_b \right)$$
(3.23)

The associated operational and planning constraints are same as Smart-DSPLAN constraints with some differences as follows:

- Constraints (2.8), (2.11), (2.13), (2.14), (2.16), (2.20), (2.23), and (2.24) are excluded.
- Constraints 2.4, is modified to include unserved power variable as follows:

$$P_{i,b} + P_{i,b}^{DG} - P_{i,b}^{PEV_S} - P_{i,b}^{PEV_{UNC}} - Pd_{i,b} + P_{i,b}^{DR} = f(V_{i,b}, \delta_{i,b}) \qquad \forall i, b \qquad (3.24)$$

• The upgrades capacities are now considered as a fixed decisions.



Fig. 3.2. Architecture of the proposed approach to solution of planning problem

3.5 Proposed Approach To Solution of Planning Problem

The proposed approach to determine the optimal upgrade plan combines a bi-level procedure as follows:

Level-1 (ADEQ-OPTSELECT): determines an appropriate reserve margin (γ) and selects the optimal size and location of distribution system upgrades that are required to be in place, at the plan terminal year.

Level-2 (ADEQ-OPTPERIOD): determines the optimal period of commissioning, for the upgrades selected in Level-1.

A back-propagation of model solution, starting from the plan terminal year T and ending at the first year, is proposed. Within this framework, a novel adequacy check is introduced that involves post-processing the plan decisions at both the levels to ensure that the target plan adequacy level is satisfied for each year (Fig.2.6). The details of the proposed framework are discussed next.

3.5.1 Level-1: ADEQ-OPTSELECT

In this stage the optimal plan decisions for the plan terminal year are determined. The step-by-step procedure is discussed as follows.

- 1. Start with an initial value of reserve margin γ and consider all LDC components as candidates for upgrade set {L}.
- 2. Execute Smart-DSPLAN model, considering set {L}, to determine the optimal planning decisions, set {H}, which provides continuous values of capacities.
- 3. Standardize the upgrade capacities of {H} and consider them as fixed decisions in the Smart-DSPLAN. Also, now include an unserved power variable in the demand-supply balance (2.4) with an associated high cost of unserved power in the objective function (2.3). The modified version is now referred to as Smart-DSPLAN2.
- 4. Execute Smart-DSPLAN2 to calculate the Benefit-Cost-Ratio (BCR) of each upgrade in set {H} by uninstalling an upgrade, one at a time, from {H}, and calculate the marginal benefit of each upgrade. The marginal benefit for an upgrade is obtained from the difference in the objective function (1) before and after removing the element. The BCR for an upgrade is calculated by dividing the marginal benefit by the total cost of the upgrade as follow:

$$BCR_{i,Type} = \frac{MB_{i,Type}}{TC_{i,Type}}$$
(3.25)

Where $MB_{i,Type}$ is the marginal benefit obtained from the difference in the objective function (J) before and after removing a specific upgrade; $TC_{i,Type}$ is the total cost of installing an upgrade; and Type is the type of upgrade (substation, DG, capacitor or feeder)

- 5. Select all upgrades with BCR>1, and form a set {H1}, also construct a set {R} with rejected upgrades having BCR<1 and solve Smart-DSPLAN2 with set H1 to check if the above selected upgrades satisfy the system constraints and there is no unserved power.
- 6. If YES, go to next step. If NO, modify set {L} with all buses except those in set {R}. If the selected upgrades from {H1} do not satisfy system constraints, and all upgrades from set {L} have been tested for BCR>1, then choose upgrades with the highest BCR from the rejected set {R}. If otherwise, go to Step 2.
- 7. Develop a capacity outage probability distribution table. Calculate the Loss of Load Expectation (LOLE) [79] for the obtained plan {H1} and check if LOLE is less than the specified value.

8. If YES, the final solution is obtained. If NO, revise γ as: $\gamma = \gamma + \Delta \gamma$, which means the reserve margin, and hence supply capacity (DG and substation), is appropriately increased to meet the Capacity Adequacy Limit in (17), and go to Step 2.

It is to be noted that the choice of the starting value of γ has no significant impact on the distribution plan. If the initial choice of this margin is too far off, the number of iterations required for convergence will be more. But since this is a long-term planning problem, solved 10 years in advance, computational time is not a critical issue, and a few additional iterations will not affect the plan. Moreover, there is no unique starting value of γ that is universally appropriate for all distribution planning problems. This is because the LOLE depends on a number of factors such as the magnitude and the duration of each load block and the number of load blocks considered, the FOR, and the capacity of the generation units. Any knowledgeable guess by the distribution system planner is sufficient; in this work, a value of $\gamma = 0.15$, which is the reserve margin in year-0 is used; and γ is increased in steps of 0.01 p.u. per iteration.

The final plan for year T is now passed on to Level-2 wherein through the back propagation process, the optimal year of commissioning of the plan upgrades are obtained.

3.5.2 Level-2: ADEQ-OPTPERIOD

The ADEQ-OPTPERIOD starts from year (T-1), and propagates back to the first year of the plan horizon to find the optimal set of upgrades for each year of the plan.

Two essential verification tests are used at this level to determine the optimal set of upgrades for each year. The first test, BCR, is used to reject the non-beneficial upgrades from that year and earlier. While the second, is the adequacy check which ensures that the selected upgrades meet a specified adequacy level.

In each year of the plan, BCR and LOLE are calculated starting from year (T-1). Based on the BCR check, any upgrade with a BCR of less than unity will be a possible candidate for rejection. If the selected upgrade does not satisfy the system constraints and the adequacy level, the rejected upgrades will be re-selected one by one based on their BCR, until the system constraints are satisfied, and the adequacy level is met. The step-by-step procedure of Level-2 is discussed as follows:

1. Set initial value t = T - 1.

- 2. Solve Smart-DSPLAN2 for year t and calculate BCR for all upgrades.
- 3. Reject upgrades with BCR<1 from $\{H1\}$ and form rejected set $\{R2\}$. It is to be noted that the upgrades that are rejected, are only for year t and earlier, implying that these upgrades are made in year t + 1.
- 4. Check if selected upgrades in {H1} satisfy the system constraints. If YES, go to Step-5. If NO, re-select the upgrade with the highest BCR from {R2} and perform step-4 again.
- 5. Develop a capacity outage probability distribution table and calculate the Loss of Load Expectation (LOLE) for the obtained plan H1 and check if LOLE is less than the specified value. If NO, re-select the upgrade with the highest BCR from {R2} and perform step-5 again.
- 6. Modify set {H1} with all upgrades that are selected for year t.
- 7. Update t = t 1. If $t \neq 0$, go to Step-2. ELSE, the final plan is obtained.

3.6 Test System

The first test system under study comprises 33 buses in radial configuration [87]. The main substation at bus-1 has two transformers of 15 MVA each, and one of 16 MVA. The total system peak demand is 37 MW in year-0.

The second test system is the IEEE 69-bus radial distribution feeder comprising one main branch and seven laterals [88]. The main substation at bus-1 has two transformers of 1.5 MVA each, and one of 2.2 MVA. The total system peak demand is about 3.8 MW in year-0. The network parameters and the load data are given in the Appendix. The system demand has been scaled to suit the problem case requirements and supplemented with the additional technical details of feeder, and substation limits, and DG options. For both systems, the demand is assumed to grow at 3% annually, all the substation transformers are assumed to have a force outage rate (FOR) of 0.02 [89], and the feeder segments are assumed to be 1 km long, Ge = 0.4 [29]. Table 2.3 provides the capital costs of the resources available for planning. Details of various other parameters used for the studies are taken from [29] and [90]. Assuming a fuel consumption of 300 m³/hr and a gas price of 0.14 \$/L, the operating cost of the DG units are obtained to be 42 \$/MWh [91], and have a FOR of 0.05 [89]. A budgetary limit of \$10 million is imposed. The targeted LOLE is assumed to be 2.8 hrs/year and the unserved energy cost is 1000 \$/MWh [92]. The load profile is represented by load duration curves which are constructed by re-arranging the chronological load curve in descending order of magnitude. It determines the number of hours per day when the load is greater than any particular amount [82]. Electricity price or the price at which the LDC imports power through the substation are specified in terms of seven load blocks, as shown in Table 2.4, using load scaling factors (LSF). The average Hourly Ontario Energy Price (HOEP) in 2012 is considered in this study to be the expected electricity price for the plan period [93]. Note that the planning problem addressed in this chapter is from the perspective of an LDC, and hence the issues of electricity market prices, bidding, and risk management, are not considered. The price at which the LDC purchases power from the external grid is assumed to be known *a priori*, and the LDC does not participate in the wholesale electricity market to purchase this power, but has a standing power purchase contract.

TABLE 3.1 Capital Cost of Utility Resources

Element	Fixed Cost	Variable Cost (\$/MVA)
Feeder	150000 /km	1000
Substation	\$200,000	50000
Gas Turbine DG	-	825000
Capacitor	-	50000

TABLE 3.2 Electricity Market Price

Load Block (b)	1	2	3	4	5	6	7
Number of Hours	5	4	4	4	3	3	1
LSF	0.4	0.5	0.6	0.7	0.8	0.9	1
ρ (\$/MWh)	10	15	22	32	48	70	103

In the present work, it is assumed that the DG units are dispatchable, and are naturalgas based; therefore siting or sizing of DG units is not constrained by the availability of resources and can be optimally determined from the model as per system requirements. In this work, the DG units are assumed to operate at unity power factor, as per [94] and [56], however, the proposed algorithm is generic and computationally fast to handle DG reactive power support capability. The Smart-DSPLAN is a mixed-integer non-linear programming model, solved using the COINBONMIN solver; while Smart-DSPLAN2 is a non-linear programming model, solved using the SNOPT solver, and are coded in GAMS.

3.7 Case Studies

3.7.1 Case-1: Base Case; 33 Bus System

In this case, the PEV charging load and DR are not considered, accordingly (17), (18) and (20), and the associated variables are excluded from the Smart-DSPLAN model. The results are divided into two parts; the first part determines the optimal upgrade plan for the terminal year. Using this output, the year of commissioning of selected upgrades is obtained in the second part.

ADEQ-OPTSELECT

The proposed schematic discussed in Section-2.10.1 is used to determine the optimal γ simultaneously with the optimal size and location of LDC component upgrades in the plan terminal year (T). Table 2.5 presents the outcome of the iterative process to arrive at the optimal value of γ .

γ	Added Capacity (MVA)	LOLE (hr/yr)	Target LOLE (hrs/yr)	Adequacy Level Met?
0.15	21.4	3.47	2.8	No
0.2	24.1	3.17	2.8	No
0.21	24.7	2.29	2.8	Yes

TABLE 3.3 Reserve Margin, Added Capacity and LOLE

After three iterations, it is found that with $\gamma = 0.21$, LOLE is obtained as 2.29 hrs/year which meets the targeted adequacy level. The step-by-step outcome of the ADEQ-OPTSELECT process for $\gamma = 0.21$ is thereafter presented in Table 2.6. In the first iteration, set {H} of selected upgrades comprises the upgrade of 1 substation, 5 DG units, 6 capacitors and 5 feeders. The BCRs for each selected upgrade is calculated and found to be greater than unity except for DG at bus 32, which is hence rejected. In the second iteration, a new set {H} is obtained. After calculating BCRs for each selected upgrade, all of them are found to have a BCR > 1, and hence, the optimal plan for the terminal year is obtained, denoted by set {H1}.

ADEQ-OPTPERIOD

At this level, the final selected upgrades from ADEQ-OPTSELECT, *i.e.*, set $\{H1\}$, are considered again. The LOLE for the upgraded system and BCR corresponding to each

Itr.	Set H of upgrades	Size, MW	BCR	Set H1	Rejected set R
1	Substation	15.4	5.56	Substation	
	DG #13	2.2	1.1	DG #13	
	DG #24	0.7	1.4	DG #24	
	DG #29	3.5	1.2	DG #29	
	DG #31	2.3	1.1	DG #31	
	DG #32	0.7	0.8	-	DG #32
	Capacitor $#18$	0.1	10.22	Capacitor $\#18$	
	Capacitor $\#19$	0.3	15.09	Capacitor #19	
	Capacitor $#20$	0.3	15.12	Capacitor $#20$	
	Capacitor $#21$	0.3	15	Capacitor $#21$	
	Capacitor $#31$	0.3	110.81	Capacitor $#31$	
	Capacitor $#32$	0.2	106.02	Capacitor $#32$	
	Feeder 1-2	6.2	303.73	Feeder 1-2	
	Feeder 2-22	1.5	76.04	Feeder 2-22	
	Feeder 3-4	1.1	44.15	Feeder 3-4	
	Feeder 26-27	0.7	43.33	Feeder 26-27	
	Feeder 28-29	0.6	55.66	Feeder 28-29	
2	Substation	15.9	6.3	Substation	
	DG #17	0.7	1.3	DG #17	
	DG #21	0.4	1.2	DG #21	
	DG $\#24$	0.7	1.4	DG #24	
	DG #29	3.8	1.3	DG $#29$	
	DG #31	3.2	1.2	DG #31	
	Capacitor $\#13$	0.1	2.68	Capacitor $#13$	
	Capacitor $\#14$	0.1	2.67	Capacitor $#14$	
	Capacitor $\#15$	0.1	2.71	Capacitor $\#15$	None
	Capacitor $\#16$	0.1	2.74	Capacitor $\#16$	None
	Capacitor $\#17$	0.3	26.65	Capacitor $\#17$	
	Capacitor $#30$	0.3	126.86	Capacitor $#30$	
	Capacitor $#31$	0.3	126.99	Capacitor $#31$	
	Capacitor $#32$	0.2	125.23	Capacitor $#32$	
	Feeder 1-2	6.4	350.08	Feeder 1-2	
	Feeder 2-22	1.5	76.04	Feeder 2-22	
	Feeder 3-4	1.1	48.55	Feeder 3-4	
	Feeder 26-27	0.2	8.6	Feeder 26-27	
	Feeder 28-29	0.2	19.08	Feeder 28-29	

TABLE 3.4 Step-by step Outcome of ADEQ-OPTSELECT with $\gamma{=}0.21$

component of $\{H1\}$ are determined starting from year (T-1), and the optimal upgrade plan for each year is obtained. In the present case study, first, the BCR is calculated for year 9, considering the selected upgrades $\{H1\}$ from Table 2.6. It is found that three feeders have BCR <1 and hence these are possible candidates for rejection and the updated plan for year 9 is checked using Smart-DSPLAN2. It is found that the system is feasible and there is no unserved power resulting from these rejected upgrades. The LOLE considering the updated plan for year 9 is 1.57 hrs/yr and meets the targeted adequacy level. Therefore, the three feeders are removed from year 9 and earlier years, and installed at year 10. Table 2.7 presents the optimal period of commissioning of each upgrade as obtained from ADEQ-OPTPERIOD.

In Table 2.8, the step-by-step procedure of ADEQ-OPTPERIOD, considering a sample year, year 6 is presented. The revised set $\{H1\}$ at the beginning of year 6 is as follows: $\{Substation, DG \#29, \#31, all Capacitors, Feeder 1-2, 2-22\}$. After calculating BCR for all these upgrades, it is found that DG #29 and #31 have BCR < 1. When Smart-DSPLAN2 is executed without these upgrades, the system encounters unserved energy. As explained in the step-by-step procedure of ADEQ-OPTPERIOD, DG #29 having the highest BCR amongst the rejected upgrades, is re-introduced at the end of iteration-1. Following the same procedure, upgrade of DG #31 is rejected in iteration-2, but system LOLE is found to be higher than the targeted adequacy level and hence DG #31 is re-introduced. In iteration-3, with no rejected upgrades LOLE is calculated and it is found that the plan meets the targeted adequacy level for year-6.

Year	Upgrades size (MW) and site (Bus)	LOLE (hr/yr)
10	(1.1) Fdr 3-4, (0.2) Fdr 26-27, and (0.2) Fdr 28-29	2.29
9	(0.7) DG #17, and (0.7) DG #24	1.57
8	(0.4) DG #21	1.55
7	-	1.52
6	(3.2) DG #31	1.46
5	(1.5) Fdr 2-22	1.9
4	(6.4) Fdr 1-2	1.39
3	-	1.39
2	-	1.18
1	(15.9) Substation, (3.8) DG $\#$ 29, Capacitors: (0.1) $\#$ 13, (0.1) $\#$ 14, (0.1) $\#$ 15, (0.1) $\#$ 16, (0.3) $\#$ 17, (0.3) $\#$ 30, (0.3) $\#$ 31, and (0.2) $\#$ 32	0.48

TABLE 3.5 Case-1 Results of ADEQ-OPTPERIOD

3.7.2 Case-2: PEV Penetration; 33-Bus Test System

The impact of PEV charging load on distribution system planning is examined in this section. Two case studies are performed, smart charging and uncontrolled charging, to

Iteration of ADEQ-OPTPERIOD	Rejected upgrades	Unserved power	Re-Introduce Upgrade?	LOLE
1	DG #29 DG #31	Yes	DG #29	-
2	DG #31	No	DG #31	3.96
3		No		1.09

TABLE 3.6 ADEQ-OPTPERIOD step-by-step procedure for year 6

evaluate their impact on the plan. It is assumed that the entire load is residential and by the plan terminal year there would be one PEV every two houses; *i.e.* 50% PEV penetration. The number of PEVs connected at a bus is calculated based on a typical average hourly load of a house, 2.08 kW [47]. The SAE J1772 standards defines two residential charging levels (Level-1 and Level-2) with a third ac level and a dc level for PEVs [54], as presented in Table 2.9. In our work we have considered Level-2 (low) charging with a charging efficiency of 85%. PHEV60 is considered in this case study, with a battery capacity of 15.9 kWh [95].

TABLE 3.7 PEV Charging Level Characteristics

Туре	Power Level
Level 1: 120 VAC	1.2 - 2.0 kW
Level 2 (low): 208-240 VAC	2.8 - 3.8 kW
Level 2: (high): 208-240 VAC	6 - 15 kW
Level 3: 208-240 VAC	>15 kW-96kW
Level 3: DC Charging: 600VDC	>15 kW-240 kW

For this case there is a need for a real and detailed travel dataset, for example the NHTS [86], which is used herein to estimate the PEV driving pattern parameters such as daily mileage driven and the probability of latest home arrival time. The PEV charging load profile is developed considering the start of charging time, the energy required at each load bus, and the charging duration.

Case-2a: Smart Charging

In this case, the LDC is assumed to have a control on PEV charging schedules. Accordingly, the DR related variable and constraint (20) are excluded from the Smart-DSPLAN model. The approach presented in Section-III is executed and results from ADEQ-OPTSELECT are obtained. It is seen that the targeted adequacy level is

Year	Upgrades size (MW) and site (Bus)	PEV Loading in MW
10	(0.4) Fdr 22-23	22.21
9	(0.4) DG #24 and (1.9) #29	15.02
8	(0.4) DG #21	10.16
7	(2.5) DG #14	6.87
6	-	4.64
5	(1.8) Fdr 2-22	3.14
4	(5.5) Fdr 1-2	2.12
3	-	1.44
2	-	0.97
	(20.9) Substation, (4.6) DG #30, Capacitors:	
1	$(0.3) \ \#28, \ (0.3) \ \#29, \ (0.3) \ \#30, \ (0.3) \ \#31,$	0.66
	and $(0.3) \# 32$	

TABLE 3.8 Case-2a Results of ADEQ-OPTPERIOD

Note: PEV loading occurs during the base load (load block = 1) at each year of the plan

now met with a higher value of $\gamma = 0.31$, as against $\gamma = 0.21$ without PEVs; this is attributed to the increased system demand from PEV charging. The optimal plan outcome of this level comprises one substation upgrade, five DG units, five capacitors and three feeder upgrades; and one iteration is needed to arrive at the final plan.

At level-2 (ADEQ-OPTPERIOD), PEV penetration over the plan period is assumed to increase exponentially; starting at approximately 1.5% in year 1 and touching 50% in the plan terminal year. The plan outcome of this level (Table 2.10) shows that the optimal charging of PEVs occurs during the base load (load block = 1) because of the prevailing low market price shown in Table 2.4.

Case-2b: Uncontrolled Charging

In this case, the impact of uncontrolled PEV charging on the required distribution system upgrades is evaluated. The PEV smart charging related constraints (17), (18) and associated variables are now excluded from the Smart-DSPLAN model, while the DR variables and constraint (2.21) is now included. A high DR cost of 1000 \$/MWh is assumed, and a budgetary limit of \$15 million is imposed for this case study. The DR option is included in order to negotiate a distribution plan when budgetary constraints are imposed on the planner. In the second level (ADEQ-OPTPERIOD), the maximum forecasted DR penetration is assumed to be 1% in year 1, which increases exponentially; rising to 20% in the plan terminal year.

The PEV charging load is simulated in two ways to mimic the behaviour of PEV customers. The first group is assumed to start charging immediately after arriving home while the second group's behavior is modelled considering a Poisson distribution to mimic those drivers' behavior who arrive at on-peak electricity price hours and wait until the onset of off-peak price to plug-in their vehicles. The charging delay (λ) is assumed considering Ontario's Time-of-Use (TOU) electricity price; for vehicles arriving during off-peak hours (7 PM - 7 AM), $\lambda = 0.1$ hours, while for on-peak arrivals, λ is the wait time between their arrival and onset of off-peak TOU price. The energy required by a PEV is calculated based on its mileage driven. The charging duration is calculated from the power drawn at a given charging level, the charging efficiency, and the energy required. The resulting uncontrolled PEV charging profile is then combined with the base demand to form the total demand profile as shown in Fig. 2.7.



Fig. 3.3. System demand without and with uncontrolled PEV charging

The proposed approach presented in Section-2.10 is executed and results from ADEQ-OPTSELECT are obtained, the targeted adequacy level is met with $\gamma = 0.54$. It is noted that the optimal γ is higher compared to the Base Case and smart charging scenarios because of the significant increase in demand during peak periods (Fig. 2.7). The plan outcome of ADEQ-OPTSELECT comprises one substation upgrade, five DG units, six capacitors, 11 feeder upgrades and the optimal DR; and one iteration is needed to arrive at the final plan. It is noted that DR is allocated in load blocks 8 and 9 which have the highest LSF and only at the remote buses. The outcome of ADEQ-OPTPERIOD presented in Table 2.11 shows that there is a need for DR capacity starting from year 7 until the plan terminal year because of the increasing penetration of PEVs.

Year	Upgrades size (MW) and site (Bus)	DR Capacity (MW)
10	(1.2) Fdr 2-3, (2.8) Fdr 3-4, (1.6) Fdr 4-5, (1.7) Fdr 26-27, and (1.1) Fdr 27-28	1.21
9	(0.1) Capacitor #21, and (2.9) DG #29	0.266
8	(0.9) DG #32	0.191
7	(11.5) Fdr 1-2, (0.6) Fdr 1-18, (3.2) Fdr 22-23, (1.6) Fdr 23-24, (2) Fdr 28-29 and (1.7) DG #28	0.137
6	-	0
5	(5) Fdr 2-22	0
4	(5) DG #12	0
3	-	0
2	-	0
1	Capacitors: (0.2) #28, (0.3) #29, (0.3) #30, (0.3) #31, and (0.3) #32, (5) DG #30, and (23.3) Substation	0

TABLE 3.9 Case-2b: Results of ADEQ-OPTPERIOD

3.7.3 Summary of 33-Bus System Studies

A comparison of the distribution system plan outcomes obtained from the Base Case, smart PEV charging and uncontrolled PEV charging cases are presented in Table 2.12. It is noted that the present worth of the plan cost in the PEV smart charging case is slightly higher than the Base Case, while much higher for uncontrolled charging. The results also reveal that the impact of PEV penetration are much damped in terms of added cost and added capacity when smart charging is considered as compared to uncontrolled charging. The proposed model can be utilized by LDCs to determine the allowable PEV penetration limits for a given distribution system infrastructure, expected impacts on system operation, the capital spending, and the required upgrades in the distribution network in case of PEV smart charging (best case scenario) and uncontrolled charging (worst case scenario).

In Fig. 2.8, the variation of LOLE with γ and the optimal value of γ , for the three considered case studies are presented. It is noted that the optimal value of γ is reached

	Case-1	Case-2a	Case-2b
Present Worth of Total Cost (M\$)	8.64	9.59	14.92
Added Capacity (MVA)	24.7	30.7	38.8
Reserve Margin, γ	0.21	0.31	0.54

TABLE 3.10 LDC Plan Comparison for 33-Bus Case Studies

when γ is below the dashed line, the Target Adequacy Level. It is noted that γ increases from 0.21 in the Base Case to 0.54 with uncontrolled PEV charging, indicating a high reserve requirement because of the significant increase in charging demand during peak periods due to high PEV arrival rates. However, when PEV smart charging is considered, optimal γ is 0.31 and the burden on the system planner is much lower with reduced peak demand compared to Case-2b (Uncontrolled Charging) because the charging demand is mostly scheduled to off-peak periods.



Fig. 3.4. LOLE for different reserve margins, , with and without PEVs

3.7.4 Case-3: Base Case; 69 Bus Test System

It is to be noted that only Base Case studies are carried out for this test system. The results reveals that the targeted adequacy level is met with $\gamma = 0.17$; the selected upgrades from ADEQ-OPTSELECT comprises one substation, two DG units, two capacitors and two feeders. Two iterations are needed to arrive to the optimal plan for the terminal year. The final plan, after executing ADEQ-OPTPERIOD, are presented in Table 2.13.

TABLE 3.11 Case-3 Results of ADEQ-OPTPERIOD

Year	Upgrades size (MW) and site (Bus)
10	-
9	-
8	(0.5) Fdr 1-2 and (0.5) Fdr 2-3
7	-
6	(0.4) DG #64
5	-
4	-
3	-
2	(0.6) DG #61
1	Capacitors: $(0.1) \# 61$ and $(0.2) \# 62$, and (1.3) Substation

3.7.5 Computational Aspects

The Smart-DSPLAN is an MINLP model which is solved using the COINBONMIN solver which uses the branch and bound, branch and cut, and outer approximation algorithms. The Smart-DSPLAN2 is an NLP model solved using the SNOPT solver which uses a sequential quadratic programming algorithm that obtains search directions from a sequence of quadratic programming sub-problems. The considered test system is programmed and executed on a Dell PowerEdge R810 server, in GAMS environment, Windows 64-bit operating system, with 4 Intel-Xeon 1.87 GHz processors and 64 GB of RAM.

The model and solver statistics are given in Table-2.14. It can be seen that Smart-DSPLAN requires more CPU time per iteration as compared to Smart-DSPLAN2 in all case studies, because of the presence of binary variables in Smart-DSPLAN. In addition, the 69-bus test system requires more time per iteration compared to the 33-bus test system due to the increased number of buses and hence the number of variables and constraints. Moreover, the total CPU time varies across the case studies as it depends on the number of iterations to arrive at the optimal solution, and the required CPU time per iteration.

3.8 Conclusions

In this chapter, a comprehensive framework for multi-year distribution system planning was presented. The proposed framework aimed to determine the optimal upgrade plan,

Case Study	Model	Single Variables	Discrete Variables	CPU time, per Iteration, s	Total CPU time, s
Case-1	Smart-DSPLAN	1,840	98	770	10,783
	Smart-DSPLAN2	2,535	0	0.22	5
Case-2a	Smart-DSPLAN	2,071	98	721	$11,\!534$
	Smart-DSPLAN2	2,535	0	0.25	4
Case-2b	Smart-DSPLAN	2,597	98	496	6,950
	Smart-DSPLAN2	$3,\!259$	0	1.75	35
Case-3	Smart-DSPLAN	3,800	206	1,061	6,363
	Smart-DSPLAN2	$5,\!258$	0	11	284

TABLE 3.12 Model Statistics

siting and sizing of DGs, substations, capacitors and feeders in distribution systems, considering the penetration of PEVs in an uncontrolled as well as smart charging environment, and DR options. Two criteria, BCR and adequacy analysis, were incorporated within the proposed framework, to determine the optimal distribution system plan. The uncontrolled PEV charging load model considered driver behaviour, arrival time, mileage, and other information to develop a charging load profile for inclusion in the planning problem. Similarly, a novel set of PEV operational constraints were included in the planning model, to capture PEV smart charging effects. The proposed planning framework was applied to two test systems. Four case studies were considered to investigate the impact of PEV smart and uncontrolled charging loads as well as DR options on the distribution plan.

The studies revealed that it is important for planners to take into consideration the effects of PEV penetration while deriving their plan outcomes, particularly, when the share of uncontrolled PEV charging is high. It was noted that the present worth of the plan cost in the PEV uncontrolled charging case was much higher than that in the Base Case (No PEV), while the impact of PEV penetration was much damped in terms of added cost and added capacity when smart charging was considered. Compared to the Base Case, with uncontrolled PEV charging loads, there was a 77% and 63% increase in the present worth of the plan cost and in the added capacity (in MW), respectively. The reason behind this significant increase was that the system peak demand coincided with the uncontrolled charging demand due to the coincident home arrival rates during peak hours. On the other hand, it was noted that there was a 11% and 24% increase in the present worth of plan cost and in the added capacity (in MW) when smart charging was used. There was a

decrease in peak demand compared to uncontrolled charging because of smart charging which scheduled the charging demand to off-peak periods.

The proposed framework can be used by LDCs to quantify the impacts, and determine the required upgrades in the distribution networks; and can be readily applied to any radial distribution system. It is also noted that although the present work considered distribution systems connected to the grid, the proposed framework is easily extendable to isolated/islanded distribution systems or microgrids, without any loss of generality.

Chapter 4

A Novel Framework for Evaluating Maximum PEV Penetration into Distribution Systems¹

4.1 Introduction

In Chapter-3, a comprehensive distribution planning model was developed to address the current change in distribution systems. The proposed model simultaneously determines optimal sizing, placement and investment timelines of various resource alternatives for LDCs such as DG sources, substations, capacitors and feeders. In addition, the impact of PEV uncontrolled and smart charging loads on the planning problem was evaluated.

This chapter presents a novel framework to determine the appropriate level of PEV uncontrolled and smart charging penetration that distribution systems can accommodate without requiring any capacity reinforcement. Monte Carlo Simulation has been used to simulate the uncertainty of typical demand, drivers behaviour, PEV market share, and charging level share. Moreover, the maximum allowable penetration of uncontrolled and

Earlier versions of the work have been published in:

¹This chapter has been accepted for publication in:

A. Bin Humayd, and K. Bhattacharya. "A Novel Framework for Evaluating Maximum PEV Penetration into Distribution Systems." IEEE Transactions on Smart Grid. (in print).

A. Bin Humayd, and K. Bhattacharya. "Assessment of distribution system margins to accommodate the penetration of plug-in electric vehicles." IEEE Transportation Electrification Conference and Expo (ITEC), 2015, Dearborn, MI, USA.

smart charging loads are determined based on the current available market data pertaining to PEV type and charging level, considering different charging scenarios. The proposed framework is examined and compared across a number of scenarios.

The structure of the chapter is as follows. The nomenclature used in this chapter is presented in Section-3.2. In Section-3.3, a description of the proposed framework is presented. This is followed by a description of case studies and assumptions in Section-3.4. In Section-3.5, the proposed framework is applied and the results are presented to demonstrate the effectiveness of the proposed model. Conclusions are drawn in Section-3.6.

4.2 Nomenclature

Indices

c	Index for PEV class
g	Index for PEV groups classified by their State of Charge (SOC)
h	Index for hours
i, j	Index for buses, $i = 1, 2, N$
N	Total number of buses
s	Scenarios of uncertainty
SS	Subset of substation buses $(SS \in i)$

Parameters

AER_c	Driving range of a PEV of class c in electric mode, km
B_c^{Cap}	Battery capacity of a PEV of class c , p.u.
D_c^{km}	Mileage driven by a PEV of class c , km
DOD	Depth of discharge of a PEV, p.u.
$E_{c,g}$	Energy needed to charge a PEV battery of class c and SOC classification group g , kWh
N_i^V	Number of vehicles at bus i
P^{ChL}	Power drawn by a PEV at a given charging level, p.u.
$P_{i,h}^D, Q_{i,h}^D$	Real and reactive power demand at bus i and hour h , p.u.
$D^{PEV_{Unc}}$	Maximum allowable uncontrolled charging demand at bus i and hour h ,
<i>i</i> , <i>h</i>	p.u.

$S_{i,j}^{Fdr_{Cap}}$	Existing capacity of feeder i - j , p.u.
$S^{\ddot{S}S_{Cap}}$	Substation capacity, p.u.
SOC_c	State of charge of a PEV of class c , p.u.
SOC^{min}	Minimum allowable SOC for a PEV, p.u.
T_c^{Ch}	Charging duration of a PEV of class c , hours
V^{Min}, V^{Max}	Lower and upper limits of voltage magnitude, p.u.
\mathbf{V}^E	Share of total fleet of class c and group g that require $E_{c,g}$ amount of
$\Lambda_{c,g}$	energy, p.u.
$X_h^{PEV_{home}}$	Share of total fleet that are at home at hour h , p.u.
$X_h^{\tilde{U}nP}$	Uncontrolled charging demand at hour h , share of total fleet, p.u.
Xc_c	Share of PEVs of class c in the total fleet, p.u.
$Y_{i,j}$	Magnitude of bus admittance matrix element i - j , p.u.
	Share of total fleet that can be accommodated without violating the
$\alpha_{h,s}^{Unc}$	Distribution System Loading Margin (DSLM) at hour h in scenario s ,
,	p.u.
α_s^{min}	Minimum value of $\alpha_{h,s}^{Unc}$ over all operating hours
η	Charging efficiency
$ heta_{i,j}$	Angle of bus admittance matrix element i - j , rad

Variables

$P_{i,j,h}^{Fdr}, Q_{i,j,h}^{Fdr}$	Real and reactive power flow from i to j at hour h , p.u.
$P_{ss,h}, Q_{ss,h}$	Real and reactive power imported by LDC via substation at hour h , p.u.
$P_{i,h}^{PEV}$	Power drawn by PEV fleet at bus i and hour h , p.u.
$P_{c,q,i,h}^{PEVc}$	Power drawn by PEV of class c at bus i and hour h , p.u.
$S_{i,j,h}^{Fdr}$	Complex power flow from i to j at hour h , p.u.
$S_{ss,h}$	Complex power imported via substation at hour h , p.u.
$V_{i,h}$	Voltage magnitude at bus i and hour h , p.u.
X_h	DSLM, at hour h, share of total fleet, p.u.
$X_{i,h}^{PEV}$	Bus loading margin at hour h and bus i , share of total fleet, p.u.
$\alpha^{Smt_{Sys}}$	Share of total fleet that can be charged in smart charging mode without violating system constraints, p.u.
α_i^{Smt}	Share of fleet at bus <i>i</i> that can be charged in smart charging mode without
	violating system constraints, p.u.
$\delta_{i,h}$	Voltage phase angle at bus i and hour h , rad

4.3 Proposed Framework For Assessment of Distribution System Loading Margin

Fig. 3.1 presents the overall schematic of the proposed framework to assess the DSLM which comprises three main stages.



Fig. 4.1. Proposed framework for assessment of DSLM

In Stage-I, a large number of uncontrolled PEV charging profiles considering different charging scenarios, charging levels, and PEV types are constructed. At this stage, there is a need for a real and detailed travel dataset such as mileage driven, home arrival and departure times, and vehicle types, which can be used to simulate the driver behaviour and hence develop the charging profiles. Stage-II starts with generating a large number of random residential load profiles which are then fed to an optimization model to determine the hourly DSLM profiles. The outcomes of Stage-I and Stage-II are then fed to Stage-III to determine the expected allowable uncontrolled and smart charging PEV penetration.

Monte Carlo simulation is carried out considering a range of vehicles daily mileage driven and hence a multitude of scenarios of state of charges (SOC) of PEVs are generated. Monte Carlo simulation is also carried out to simulate the start time of charging for three different charging strategies. It is to be noted that for PEV charging patterns, to date, no real data is available. The simulations carried out in this work are extensive and takes into account realistic probability distributions from National Household Travel Survey (NHTS) data, and the PEV charging patterns developed in this chapter are based on the use of the battery technologies available in todays market.

4.3.1 Stage-I: Uncontrolled PEV Charging Load Modeling

This stage focusses on developing the PEV charging profile using vehicle type, charging level information and the charging scenarios. A number of daily PEV charging scenarios at each bus is estimated assuming that the whole fleet comprises PEVs. After that, the hourly PEV load is transformed to per unit as a fraction of the total fleet, X_h^{UnP} , which is used later in stage-III. The uncontrolled charging profile is constructed based on three main steps as follows:

Simulation of Driving Pattern Parameters

Using the NHTS data, the probability density function (pdf) of the daily mileage and home arrival times are determined. The cumulative distribution function of the two parameters are then obtained, which are used for Monte Carlo simulations to generate random trip mileage and home arrival times, to be used in Step-2 and 3.

Determine Charging Duration

The charging duration, T_c^{Ch} , for each vehicle in the fleet is estimated in this step, from the power drawn at a given charging level (P^{ChL}) , the charging efficiency (η) , the SOC of the vehicle, and the battery capacity, B_c^{Cap} , as follows:

$$T_c^{Ch} = \frac{(1 - SOC_c)B_c^{Cap}}{P^{ChL}\eta}$$

$$\tag{4.1}$$

the SOC is calculated from the mileage driven (D_c^{km}) , generated using the pdf of the daily mileage, the driving range in the electrical mode (AER_c) , the SOC^{min} and DOD of the vehicle, as given below:

$$SOC_{c} = \begin{cases} SOC^{min} + (1 - \frac{D_{c}^{km}}{AER_{c}})DOD & D_{c}^{km} \le AER_{c} \\ SOC^{min} & D_{c}^{km} > AER_{c} \end{cases}$$
(4.2)

There are several organizations that assess the emission level of car engines, fuel economy and driving range based on a series of tests, for example, the Environmental Protection Agency (EPA) in the US and the New European Driving Cycle (NEDC) in Europe. The driving range in the electrical mode (AER_c) , used in this chapter, is based on the EPA since their fuel economy tests have been correlated with US national average values for many real-world driving conditions, including stop-and-go traffic, cold weather, air conditioning use, and high speed and aggressive driving [96, 97]. In [98], a study is reported that analyzes fuel economy of PEVs and the factors influencing it. It is reported in the study that EPA ratings are used as a real-world electric driving range, or 0.75 of the NEDC value, if no EPA range is available.

The PEVs considered in this chapter are from eleven commonly found makes, based on their sales, and covers 95% of the total number of PEVs sold in the US between 2010 to September 2015 [99]. These PEVs are grouped into four major classes, plug-in hybrid electric vehicle-I (PHEV), PHEV-II, battery electric vehicle-I (BEV) and BEV-II, with their parametric details provided in Table-3.1.

The charging level is another important factor that affects the charging duration and the power drawn from the grid. The SAE J1772 standard [100] defines two residential charging levels (Level-1 and Level-2) for PEVs and the same have been used in several studies [44,54]. In this study too, these charging levels are used; the PEV owners survey [101] reports that 65% customers have Level-1 access while 35% have the potential to install Level-2 charging. Furthermore, to investigate the effect of different charging levels, three different scenarios of mix of charging levels are formulated, as listed in Table-3.2.
	Market Share	Battery Capacity Range (kWh)	AER (km)	Weighted Average Battery Size (kWh)	Weighted Average AER (km)	Vehicls' Type
PHEV-I	26.05%	4.4-7.6	18-32	5.76	24.03	Ford C-MAX Energi, Ford Fusion Energi, Toyota Prius Plug-in
PHEV-II	23.83%	18.4	85	18.40	85	Chevrolet Volt
BEV-I	34.22%	14-24	100-135	21.89	124.03	BMW i3, Chevrolet Spark, Fiat 500E, Ford Focus Electric, Nissan LEAF, Smart fortwo
BEV-II	15.90%	85	426	85.00	426	Tesla Model S

 TABLE 4.1

 PEV CLASSES CONSIDERED FOR STUDIES

Scenarios of Starting Time of Charging

The behaviour of PEV drivers has a direct impact on the starting time of charging and hence the charging profile, which plays a significant role in determining the maximum allowable PEV penetration in distribution grids. Previous researchers have considered various scenarios to simulate the starting time of charging. In [40–42, 44, 61], the authors assumed that PEV drivers will start charging immediately after arriving home after their last trip while in [59,60], overnight charging [8 PM - 8 AM] have been considered. To date, no real data are available with respect to starting time of charging and PEV charging profiles and therefore there is a need for rational assumptions.

In this work, the starting time of charging is modelled considering three realistic scenarios to mimic the behaviour of PEV customers and hence assess the impact of the charging load, arising there from, on the distribution.

- Arrive and Plug: In this scenario, it is assumed that PEV owners start charging their vehicles immediately after arriving home after their last trip, regardless of the electricity price or network limitations; the charging takes place at a constant rate until the battery is full.
- *TOU-I*: In this scenario, all PEV drivers are assumed to respond to TOU electricity price. Therefore, PEVs arriving at on-peak electricity price hours wait until the onset of the off-peak price to plug-in their vehicles. This scenario is considered the worst case scenario since all PEVs arriving home at peak hours, simultaneously start charging their vehicles at the onset of the off-peak period.

 TABLE 4.2

 Charging Level Scenarios Considered For Studies

Changing Samania	Percentage (%)			
Charging Scenario	Level-1	Level-2		
Scenario-1	100	0		
Scenario-2	0	100		
Scenario-3	65	35		

• TOU-II: The starting time of charging, in this scenario, is modelled considering a Poisson distribution with λ equals to the charging delay which is assumed considering Ontarios TOU electricity price. For vehicles arriving during off-peak hours (7 PM - 7 AM), $\lambda = 0.1$ hours, while for on-peak arrivals, λ is the wait time between their arrival and onset of off-peak TOU price.

After determining the charging duration and the starting time of charging of each vehicle in the fleet, the uncontrolled charging profile is constructed. For example, if the charging duration of a vehicle is 5 hours considering Level-1 charging, and the starting time of charging is 3 pm, the charging demand will take place from 3 pm to 8 pm with an amplitude of 1.44 kW (Level-1 charging power).

4.3.2 Stage-II: Distribution System Loading Margin Assessment

A Monte Carlo simulation of the peak day's demand is carried out to arrive at a large number of hourly load profiles $P_{i,h}^D$ and $Q_{i,h}^D$, which are then fed into an optimization model to determine the hourly DSLM. The randomness of the hourly loads are simulated considering Gaussian distributions with specified hourly averages and standard deviations.

Objective Function

The objective function (J_1) seeks to maximize the aggregate hourly DSLM which essentially maximizes the PEV charging load at every bus and hour, $P_{i,h}^{PEV}$, and is given as:

$$J_1 = \sum_h X_h \tag{4.3}$$

The associated operational constraints are as follows.

Power Flow Equations

The injected power at a bus i and hour h is the power from the substation, net of the load, and potential PEV charging loads; and is governed by the traditional power flow equations:

$$P_{i,h} - P_{i,h}^{PEV} - P_{i,h}^{D} = \sum_{j \in N} V_{i,h} V_{j,h} Y_{i,j} \cos(\theta_{i,j} + \delta_{j,h} - \delta_{i,h}) \quad \forall i \in N, \forall h \quad (4.4)$$

$$Q_{i,h} - Q_{i,h}^D = -\sum_{j \in N} V_{i,h} V_{j,h} Y_{i,j} sin(\theta_{i,j} + \delta_{j,h} - \delta_{i,h}) \qquad \forall i \in N, \forall h \quad (4.5)$$

PEV Accommodation Constraints

These constraints ensure that all buses have a loadability margin at least equal to DSLM, X_h , which is governed by the charging load that can be accommodated at a bus *i*, the bus loading margin, $X_{i,h}^{PEV}$, and is proportional to the number of vehicles at that bus, N_i^V .

$$P_{i,h}^{PEV} = X_{i,h}^{PEV} N_i^V P^{ChL} \quad \forall i \in N, \forall h$$

$$(4.6)$$

$$X_h \le X_{i,h}^{PEV} \quad \forall i \in N, \forall h \tag{4.7}$$

In addition, constraint (3.6) ensures that the power drawn by PEVs during an hour and at bus *i* is as per the charging level, P^{ChL} . Since the bus loading margins $(X_{i,h}^{PEV})$ should be greater than or at least equal to X_h , as per (3.7), the model effectively ensures that all bus loading margins are maximized as well, and the lowest of the bus loadings margins is at least equal to the DSLM (X_h) .

Feeder Capacity Limits

Power flow through any distribution feeder must comply with the thermal capacity limit of the feeder.

$$P_{i,j,h}^{Fdr} = -V_{i,h}^2 Y_{i,j} \cos\theta_{i,j} + V_{i,h} V_{j,h} Y_{i,j} \cos(\theta_{i,j} + \delta_{j,h} - \delta_{i,h}) \quad \forall (i,j) \in N : \exists (i,j), \forall h \quad (4.8)$$

$$Q_{i,j,h}^{Fdr} = V_{i,h}^2 Y_{i,j} sin\theta_{i,j} - V_{i,h} V_{j,h} Y_{i,j} sin(\theta_{i,j} + \delta_{j,h} - \delta_{i,h}) \quad \forall (i,j) \in N : \exists (i,j), \forall h \quad (4.9)$$

$$S_{i,j,h}^{Fdr} \leq S_{i,j}^{Fdr_{Cap}} \qquad \forall (i,j) \in N : \exists (i,j), \forall h \qquad (4.10)$$

Substation Capacity Limits

This constraint ensures that the total power transferred over the substation transformers is within substation capacity limit, $S^{SS_{Cap}}$.

$$S_{ss,h} \le S^{SS_{Cap}} \quad \forall i \in SS, \forall h \tag{4.11}$$

Voltage Limits

This constraint ensures that the voltage magnitude at a bus is within the minimum and maximum allowable voltage limits.

$$V^{Min} \le V_{i,h} \le V^{Max} \quad \forall i \in N, \forall h \tag{4.12}$$

4.3.3 Stage-III: Assessment of Allowable PEV Penetration

The expected allowable penetration of uncontrolled and maximum allowable penetration of smart charging PEVs are determined, in this stage, using the outcomes of Stages-I and II. It starts with determining the expected maximum allowable uncontrolled PEV charging load profile, which is fed to an optimization model to determine the maximum allowable PEV smart charging load.

Expected Allowable Uncontrolled PEV Penetration

The expected value of the allowable uncontrolled penetration, $E[\alpha_s^{min}]$, is defined at this stage. First, the percentage of the fleet that can be accommodated without violating the hourly DSLM, $\alpha_{h,s}^{Unc}$, for each scenario s of the Monte Carlo simulation and hour h, is calculated as follows:

$$\alpha_{h,s}^{Unc} = \frac{X_{h,s}}{X_{h,s}^{UnP}} \qquad \forall s,h \tag{4.13}$$

Where, $X_{h,s}$ denotes DSLM, and $X_{h,s}^{UnP}$ the uncontrolled charging demand at hour h, for given loading scenarios. For example, if $X_{h,s}$, for hour h, and scenario s, equals 0.2 it means that the system can accommodate up to 20% of the total fleet at that hour.

Similarly, $X_{h,s}^{UnP}$ equals to 0.4 means that 40% of the total fleet is ready to charge in uncontrolled mode at hour h. Accordingly, the value of $\alpha_{h,s}^{Unc}$ will be 0.5, *i.e.*, 50% of the uncontrolled charging fleet can be accommodated by the system at hour h. Then the maximum allowable uncontrolled PEV penetration for scenario s, α_s^{min} , is the minimum of all values of $\alpha_{h,s}^{Unc}$, as follows:

$$\alpha_s^{min} = min \ \left(\alpha_{1,s}^{Unc}, \alpha_{2,s}^{Unc}, ... \alpha_{24,s}^{Unc}\right) \qquad \forall s \tag{4.14}$$

The expected value of α_s^{min} , *i.e.*, $E[\alpha_s^{min}]$, is obtained when the desired level of accuracy is attained in the Monte Carlo simulation. The expected value of $X_{h,s}^{UnP}$, $E[X_{h,s}^{UnP}]$ is also simultaneously obtained after the convergence of the Monte Carlo simulations. The expected allowable uncontrolled PEV charging profile is thereafter obtained as follows:

$$E[P_{i,h}^{PEV_{Unc}}] = E[\alpha_s^{min}]E[X_{h,s}^{UnP}]N_i^V P^{ChL} \qquad \forall i,h$$

$$(4.15)$$

Maximum Allowable Smart PEV Penetration

An optimization model is proposed, considering the expected allowable uncontrolled PEV charging profile obtained earlier $(E[P_{i,h}^{PEV_{Unc}}])$ as a fixed load profile, to determine the maximum allowable smart PEV penetration, $\alpha^{Smt_{Sys}}$. Since the smart charging PEVs are assumed to be controlled by the LDC, and the LDC seeks to understand the maximum allowable loadability, the charging schedules are optimally allocated over the day to maximize the number of PEVs charging.

The proposed mathematical model for evaluating the maximum allowable penetration of smart charging PEVs in the distribution grid is presented here. The objective function (J_2) aims to maximize the system penetration of PEV smart charging, $\alpha^{Smt_{Sys}}$.

$$J_2 = \alpha^{Smt_{Sys}} \tag{4.16}$$

The operational constraints, presented in Section 3.3.2, are now modified to include the expected uncontrolled PEV charging profile, $E[P_{i,h}^{PEV_{Unc}}]$, obtained from Section 3.3.3, in the active power balance constraint as follows:

$$P_{i,h} - \sum_{c,g} P_{c,g,i,h}^{PEVc} - P_{i,h}^{D} - E[P_{i,h}^{PEV_{Unc}}] = f(V_{i,h}, \delta_{i,h}) \quad \forall i \in N, \forall h \quad (4.17)$$

In addition, the PEV accommodation constraints (3.6) and (3.7) are now replaced by constraints (3.18)-(3.22) to ensure that the total energy required by each class c of PEV is equal to the daily charging energy drawn from the grid (3.18) and that the power drawn by PEVs during an hour must be within the charging level (3.19). It is to be noted that the range of SOC and hence the energy required by each PEV class has been divided into a number of discrete intervals, g, and $E_{c,g}$ denotes the energy required by a PEV of class c, in range g, with $X_{c,g}^E$ denoting the percent of fleet that require $E_{c,g}$. Constraint (3.20) ensures that the penetration of smart charging PEVs at a bus i is greater or at least equal to the system smart charging penetration. While (3.21) ensures that the percentage of PEV charging at hour h and bus i does not exceed the percentage of PEVs being home at hour h, $X_h^{PEV_{home}}$.

$$\sum_{h} P_{c,g,i,h}^{PEVc} = \alpha_i^{Smt} X c_c E_{c,g} X_{c,g}^E N_i^V / \eta \qquad \forall i \in N, \forall c, \forall g \qquad (4.18)$$

$$\sum_{c,q} P_{c,g,i,h}^{PEVc} \le N_i^V P^{ChL} X_h^{PEV} \quad \forall i \in N, \forall h$$

$$(4.19)$$

$$\alpha^{Smt_{Sys}} \le \alpha_i^{Smt} \quad \forall i \in N \tag{4.20}$$

$$X_h^{PEV} \le X_h^{PEV_{home}} \quad \forall h \tag{4.21}$$

$$0 \le X_h^{PEV} \le 1 \quad \forall h \tag{4.22}$$

Where X_h^{PEV} is the same as $X_{i,h}^{PEV}$ used in Section-3.3.2 except that the values are same for all buses.

4.4 Case Studies And Assumptions

4.4.1 Descriptions of Case Studies

The main focus of this work is to evaluate the DSLM to accommodate PEVs under uncontrolled and smart charging. The work is carried out considering the impact of different parameters such as charging level, electricity price, and driver behaviour. To achieve this target, nine case studies have been constructed as shown in Table-3.3.

	Charging Level Scenario (refer to Table-3.2)	Charging Scenario
Case-1a		Arrive & Plug
Case-1b	Scenario-1	TOU-I
Case-1c		TOU-II
Case-2a		Arrive & Plug
Case-2b	Scenario-2	TOU-I
Case-2c		TOU-II
Case-3a		Arrive & Plug
Case-3b	Scenario-3	TOU-I
Case-3c		TOU-II

TABLE 4.3Description of Case Studies

4.4.2 Assumptions

In order to evaluate the PEV penetration and the maximum accommodation by the system, the following assumptions are made:

- The distribution system under study comprises 33 buses in radial configuration (Fig.3.2) [87]. The main substation is at bus-0 with a capacity of 6 MVA, and the total system peak demand is 4.54 MVA. The network parameters and the load data are given in the Appendix.
- The IEEE RTS [102] peak day load profile is used. Only the peak day demand is considered in this study to determine the expected allowable PEV penetration, since assessing the same for average or non-peak loads can result in degraded performance during peak load periods. The randomness of bus-wise and hourly loads are simulated based on Gaussian distributions with a standard deviation of 0.15 p.u. Fig. 3.3 and 3.4 presents the original and Monte Carlo generated load profiles for some buses for a peak day.
- NHTS 2009 data [86] has been considered with a sample size of 150,147 usable households, and 309,164 vehicles. The data reflects daily trips over a 24-hour period, and were reportedly collected for different type of vehicles, trips, purposes, trip lengths. Although NHTS pertains to conventional vehicles, the same dataset has been used because of lack of PEV data, to obtain the probability distribution function (pdf) of the daily mileage driven and home arrival times of the PEVs.



Fig. 4.2. Test system configuration and number of vehicles at each bus, shown inside the circles



Fig. 4.3. RTS peak day load profile at some buses

Thereafter, Monte Carlo simulations are carried out to generate random trip mileage and home arrival times to build the uncontrolled charging profiles at each bus. In addition, for the optimization model in Stage-3 (Section-3.3.3), parameters such as $E_{c,g}$, $X_{c,g}^E$, and $X_h^{PEV_{home}}$ are obtained from the NHTS data set.

• The randomness of PEV market share and charging level mix are simulated based on Gaussian distributions with a standard deviations of 0.05 p.u.



Fig. 4.4. Monte Carlo generated load profile at some buses for a given scenario

- A 24 hour time horizon is assumed, with one hour time intervals. However, the proposed model can accommodate smaller time steps if required, but at the cost of increased computational burden.
- The charging efficiency is assumed to be 90% [44, 46, 52].
- With respect to the number of vehicles at each bus, N_i^V , it is assumed that the entire load is residential and there are two vehicles per house and PEV charging occurs only at home. The number of houses at a bus is calculated assuming that the peak load of a house is 2.08 kW. For example, given that bus-23 peak load is 420 kW, as given in [87], the number of houses is 420 kW/2.08 kW \approx 202 houses, and accordingly the number of vehicles at bus-23 is 404. In Fig. 3.2. the numbers inside the circles denotes the number of vehicles at each bus (N_i^V) .
- Given the battery life-cycle considerations, it is assumed that DOD = 0.7 p.u., and $SOC^{min} = 0.2$ p.u. [60].
- The percentage of PEV charging is limited by the percentage of PEVs being home at an hour, $X_h^{PEV_{home}}$, as determined from the NHTS data, Fig. 3.5.



Fig. 4.5. Probability of vehicle being at home

4.5 Results And Discussions

4.5.1 Uncontrolled Charging Profiles

The uncontrolled charging profiles developed from Stage-I, for the considered case studies are presented in Fig. 3.6 wherein the differences among the various profiles are distinct. As can be seen, the charging load in the *Arrive & Plug* charging scenarios, *i.e.*, Cases 1a, 2a, and 3a, are more distributed over a period of time, based on the home arrival rate profile, and has a window of peak load appearing between hours 16 to 21.

On the other hand, the peak time window for *TOU-I* and *TOU-II* scenarios is narrower since all PEV owners react to the same price signal. In addition, most PEVs start charging simultaneously at the beginning of the off-peak price period (hour 19).

Level-2 charging, Cases 2a to 2c, are considered to be of a higher charging level than Level-1, and, therefore, has faster charging times, the peak charging hours are shifted to earlier hours compared to Level-1 charging profiles (1a - 1c). In addition, the charging demands are less distributed over the day and tend to have lower simultaneous charging due to shorter charging duration. Moreover, it is noted that overnight Level-1 charging demand is higher than Level-2 demand. This is due to the fact that lower charging level increases the charging duration.



(c) Cases 3a-3c Expected Uncontrolled Charging Profiles

Fig. 4.6. Uncontrolled charging profiles for the proposed case studies

4.5.2 Distribution System Loading Margin

Fig. 3.7 presents the DSLM profiles considering charging Level-1 and Level-2 respectively, for different loading scenarios, presented as a percentage of the fleet that can be charged. It can be seen from Fig. 3.7, that simultaneous PEV charging is high between midnight and 7 AM which can be attributed to the decreased system demand. Between 11 AM and 3 PM the maximum and minimum percentages of PEVs charging simultaneously are approximately 30% and 4% for Level-1 charging, while 6% and 2% for Level-2 charging. It is noted that the percentages of simultaneous charging considering Level-1 are much higher than those for Level-2. This is due to the fact that lower charging levels have a low level of power drawn from the grid which enables more simultaneous charging PEVs.



Fig. 4.7. DSLM profiles obtained from Stage-II

4.5.3 Maximum Allowable Penetration Level

In this section, the expected allowable penetration of uncontrolled and smart charging PEVs are determined, using the outcomes of Stages-I and II. As discussed earlier in Section-3.3.3, first, the maximum allowable uncontrolled PEV penetration is determined, using Eq. 3.14 for each scenario and its expected value is arrived at after convergence of the Monte Carlo process. Table-3.4 presents the obtained values of $E[\alpha_s^{min}]$, for the different cases considered. Fig. 3.8 shows the convergence to the expected value, obtained by the proposed





Fig. 4.8. Case-3a stopping criteria

It can be seen from Table-3.4 that the Arrive \mathcal{C} Plug charging cases (Cases- 1a, 2a and 3a) allow higher levels of penetration compared to the other charging cases. Since all PEVs react on the price signal in *TOU-I* cases, resulting in a large number of PEVs simultaneously charging at the start of the off-peak price period, the maximum allowable penetration level in these cases (Cases 1b, 2b and 3b) are lower than those of the Arrive \mathcal{C} Plug cases. The same applies to *TOU-II* cases (1c, 2c and 3c), although the difference is somewhat smaller.

With regard to the charging level, it is noted from Table-3.4 that for Level-1 (Cases 1a to 1c), the maximum allowable uncontrolled penetration is greater as compared to Level-2 (Cases 2a to 2c). This is because of the fact that a lower charging level increases the charging time which consequently allows more PEVs to charge simultaneously.

In addition, the new expected peak load hours and consequently the bottleneck hours are determined by comparing the base load and the uncontrolled PEV charging load profiles, and are presented in Table-3.5. It is to be noted that the bottleneck hours vary for different cases. In the TOU cases, most PEVs start charging at the beginning of the off-peak price period (hour 19) and hence the bottlenecks are concentrated at hours 19, 20 and 21; while for the Arrive & Plug cases, the bottleneck hours are spread out since the charging is spread out over a period based on PEV arrival rates; high probability of bottleneck at hour 18 is noted, which coincides with the high arrival rate at this hour.

The expected allowable uncontrolled profile is determined using (3.15), which is then used in the proposed optimization model of Section 3.3.3 to determine the maximum allowable smart PEV penetration, $\alpha^{Smt_{Sys}}$, and is given in Table-3.4. Since the smart charging PEVs are assumed to be controlled by the LDC, their charging schedules are optimally allocated over the day to maximize the number of PEVs charging. Hence the PEV penetration for various cases, considering smart charging, are much higher than that for uncontrolled charging.

	$E[\alpha_s^{min}]$ (% of Fleet)	$\alpha^{Smt_{Sys}}$ (% of Fleet)
Case-1a	46.5	108.4
Case-1b	33.3	122.3
Case-1c	43	113.6
Case-2a	28.1	123
Case-2b	8.5	142.6
Case-2c	19.6	131
Case-3a	36.6	119
Case-3b	17.8	141
Case-3c	31	126

TABLE 4.4MAXIMUM ALOWABLE PENETRATION LEVEL

		Bot	tlene	ck Ho	urs Pi	robabi	lity	
	15	16	17	18	19	20	21	22
Case-1a	0.01	0.02	0.11	0.36	0.21	0.15	0.06	0.09
Case-1b					0.38	0.39	0.23	
Case-1c					0.03	0.41	0.26	0.31
Case-2a	0.04	0.13	0.34	0.41	0.05	0.00	0.04	
Case-2b					1.00			
Case-2c				0.02	0.54	0.39	0.05	
Case-3a	0.02	0.05	0.19	0.46	0.14	0.09	0.05	
Case-3b					0.95	0.04	0.01	
Case-3c				0.01	0.30	0.48	0.16	0.05

TABLE 4.5BOTTLENECK HOURS FOR THE PROPOSED CASE STUDIES

Fig.3.9 presents the resultant expected allowable uncontrolled charging, smart charging, typical, and total load (Base+Uncontrolled+Smart) profiles for three specific cases. It can be seen, in all cases, that the peak demand from the uncontrolled charging profiles are concentrated between hour 17 to 22 due to customers' arrival rate and TOU prices while

smart charging demands are mostly distributed between midnight and 7 AM which can be attributed to the decreased system demand. It is to be noted that the smart charging demand profile coordinates with the uncontrolled profile and reduces charging during peak hours when uncontrolled charging is high.



(c) Case-3a Expected Uncontrolled and Smart Charging Profiles

Fig. 4.9. Expected Allowable Uncontrolled and Smart Charging Profiles

It is to be noted that current TOU electricity prices were designed to reduce the system peak load and flatten the load profile, and bring about many benefits to the system. However, TOU prices can adversely impact the loadability of the distribution system and hence the uncontrolled PEV penetration. Therefore, further research should be undertaken to investigate the optimal TOU pricing scheme considering the penetration of PEV loads. Furthermore, the charging level is clearly a major factor that affects the uncontrolled PEV charging load profile. From a customers' perspective, a higher charging level is preferable because it requires shorter time to charge the vehicle. It is also worth noting that there is enough margin for the whole fleet for smart charging. Unlike typical loads, PEVs are considered more flexible and elastic, and can be controlled by LDCs with the right incentives, which could be beneficial for LDCs without affecting customers convenience.

It should be pointed out that with increased PEV penetration into the grid, and with shorter charging durations, there will be increased charging demand appearing on the grid. This will indeed impact the system operational aspects and therefore unit ramping capabilities, frequency regulation, etc.. While some researchers have considered the problem of frequency regulation using grid-to-vehicle charging operation [103, 104], some others have examined these aspects considering the vehicle-to-grid operation mode [105–107]. However, since the framework proposed in this chapter seeks to assess distribution system margins in accommodating PEVs in the long-run, the very short-term system operational dynamics are not considered.

4.5.4 Effect of Charging Level

To study the impact of charging level on the penetration of uncontrolled and smart charging loads, three more combinations of charging levels are studied, for the Arrive and Plug scenario, namely 75%-25%, 50%-50%, and 25%-75% of Level-1 and 2 respectively, as shown in Fig. 3.10. It can be seen that as the share of Level-2 charging increases, the penetration of uncontrolled charging PEVs reduce. This is due to the fact that more Level-2 charging reduces the charging time and also involves more power being drawn from the grid, consequently reducing the maximum allowable uncontrolled penetration. On the other hand, smart charging penetration increases with the increase in share of Level-2 charging since the reduction in energy drawn by the uncontrolled fleet will lead to more room for smart charging.

4.5.5 Effect of Reactive Compensation Devices

When reactive devices are considered in the distribution system, the system will be able to accommodate an increased level of penetration of PEVs. Inclusion of reactive power compensation devices will improve the voltage profile and reduce losses. In order to examine this, a case study with similar assumptions of Case-3a, with two capacitors of 100 kVAR and 300 kVAR at buses 12 and 29, respectively, are considered. The location of the



Fig. 4.10. Impact of charging level to the penetration of PEVs

capacitors has been chosen based on reference [108] which uses the same test system. It is noted in Table-3.6 that the maximum allowable uncontrolled ($E[\alpha_s^{min})$) and smart charging penetrations ($\alpha^{Smt_{Sys}}$) are 45% and 139.2%, respectively, which means an increase in both uncontrolled and smart charging penetration by approximately 9% and 20%, respectively.

Table generated by Excel2LaTeX from sheet 'Sheet1'

TABLE 4.6EFFECT OF REACTIVE COMPENSATION DEVICES ON CASE-3A

	$E[\alpha_s^{min}]$ (% of Fleet)	$\alpha^{Smt_{Sys}}$ (% of Fleet)
Case-3a	36.6	119
Case-3a (considering reactive devices of reference $[108]$)	45.03	139.2

4.5.6 Validation of Results

To examine the validity of the assessment, load flow runs are carried out considering the obtained uncontrolled and smart charging penetrations for Case-3a, namely 36.6% and 119% respectively, as a fixed loads. From Fig. 3.11, it can be seen that bus voltages at hour-14 are all within the allowable limits. Voltages at buses-15, 16, and 17 are close to the lower limit of 0.95 p.u. due to their location at the end of the feeder.

Furthermore, the penetration of the smart charging fleet is increased slightly to 119.1%. It can be seen that the voltages at buses 12-17 violated the allowable voltage limits due to the marginally increased penetration. These results validate the fact that the margin

determined from the proposed framework are the critical limits which the planners need to take into account in their planning studies.



Fig. 4.11. Case-3a voltage profile for the system buses at hour-14

4.5.7 Computational Aspects

The OPF models, in Stage-II and Stage-III, are solved using the SNOPT solver [109] which is suitable for non-linear programming (NLP) problems. The considered test system is programmed and executed on a Dell PowerEdge R810 server, in GAMS environment [110], Windows 64-bit operating system, with 4 Intel-Xeon 1.87 GHz processors and 64 GB of RAM.

The model and solver statistics are given in Table-3.7. The OPF model in Stage-II requires 2.153 seconds per iteration, while in Stage-III it requires about 401 seconds. Note that the OPF in Stage-II is executed up to 350 times (until convergence of Monte Carlo simulation) while the Stage-III OPF is executed once. Therefore, the total time required to perform one case study is $(2.153 \times 350 + 401)$ seconds, i.e., 19 minutes; which is reasonable, for an essentially offline, long-term assessment study. Therefore, the proposed approach is scalable to large real distribution systems.

TABLE 4.7MODEL STATISTICS

	Stage-II Distribution Assessment Margin Model	Stage-III Maximum Smart PEV Penetration Model
Model Type	NLP	NLP
Solver Used	SNOPT	SNOPT
Single Equations	6,217	20,097
Single Variables	5,521	207,538
CPU time, s	2.153	401.016

4.6 Conclusions

This chapter presented a novel framework to assess and determine the level of uncontrolled and smart PEV penetration that a distribution system can accommodate without any additional investments or upgrades. The model incorporated PEV driver habits, various vehicle types and charging levels in order to capture the charging pattern of PEV owners. In addition, Monte Carlo simulations were used to take various uncertainties into account. Various case studies were presented to demonstrate the performance of the proposed model, which can be used by LDCs to quantify the impacts, decide on the right actions and policies, and the required upgrades in their distribution networks.

This chapter fills the gap between current distribution system status and ongoing research on future grid by developing a generic framework that can be used to assess the DSLM to accommodate PEV charging loads.

Chapter 5

Design of Optimal Incentives for Smart Charging Considering Utility-Customer Interactions and Distribution Systems Impact¹

5.1 Introduction

Electrifying the transportation sector will impact the distribution networks with increased peak load, increased losses, deterioration in voltage profile and change in load pattern. To mitigate these effects, the LDCs need to adopt the right actions and policies, and develop associated infrastructure. In the context of smart grids, the LDCs can control PEV charging demand while also considering customer preferences, which can lead to benefits such as deferment of the decisions on reinforcement and other investments, and maximize the use of existing infrastructure. In addition, LDCs can establish rate structures that incentivize the use of smart charging and increase the adoption and use of PEVs. Therefore, electrifying the transport sector can benefit both the LDCs and the customer.

To encourage PEV owners adopt smart charging, the LDC can offer appropriate programs and/or incentives to PEV customers. However, an LDC managed smart

¹This chapter has been submitted for publication in:

A. Bin Humayd, and K. Bhattacharya. "Design of Optimal Incentives for Smart Charging Considering Utility-Customer Interactions and Distribution Systems Impact." IEEE Transactions on Smart Grid. (in review).

charging program involves a business relationship between the two parties. Assuming rational behaviour of PEV customers, their participation in smart charging programs will depend on the incentive amount, higher the incentive, more PEV customers are likely to adopt smart charging. On the other hand, from the LDC's viewpoint, high incentives result in increased financial burden to itself, and adoption of smart charging will only be beneficial in the short-run, by virtue of its flattening the overall system load profile to reduce its demand charges; and in the long-run, when deferral costs of capacity additions are taken into account.

Denoting X^S as the share of PEV fleet that arrive at peak hours and participate in the smart charging program, it can be assumed that there exists a relationship between the incentives offered by the LDC and X^S . Also, a relationship exists between X^S and the peak load reduction achieved by the LDC through smart charging, in the short-run; and the resulting economic benefit of capacity deferral accrued there from in the long-run.

None of the reported works have considered these inter-relationships in smart charging programs, *i.e.*, between X^S and the incentives offered by the LDC; or between X^S and the resulting economic benefits of capacity deferral. The optimal participation of PEV customers in smart charging programs, arising from such inter-relationships and the optimal incentive mechanisms, from a long-term perspective, have not been investigated. Therefore, there is a need to develop a generic framework to determine these inter-relationships between the LDC and PEV customers, while considering their own perspectives of system operations and economic returns from such a program. There is also a need to determine the optimal participation of PEV customers that would result in the optimal benefits to both parties; and what optimal incentive would drive such an optimal participation.

In this chapter, a PEV smart charging approach is proposed where the charging loads are controlled and incentivized by the LDC for every unit of energy controlled. A novel framework is proposed to determine the optimal participation of PEVs in the smart charging program and optimal incentives paid by the LDC to PEV customers, such that both parties are economically benefited. The proposed framework models the relationship between customers' participation and incentives offered by the LDC. The relationship between the expected investment deferral and hence the economic benefits from smart charging participation are considered as well. Monte Carlo simulation is used to simulate the uncertainty of demand, electricity market price, drivers' behaviour, PEV market share, and charging level share.

The structure of the chapter is as follows. The nomenclature used in this chapter is presented in Section-4.2. In Section-4.3, a description of the proposed framework and the

associated mathematical models are presented. This is followed by a description of the case studies and assumptions in Section-4.4. In Section-4.5, the proposed framework is applied to a distribution system and the results are presented to demonstrate the effectiveness of the proposed model. Conclusions are drawn in Section-4.6.

5.2 Nomenclature

Indices

Index for charging level type, charging level 1 and 2
Index for PEV groups classified by their charging duration, $g =$
1, 2,G
Index for hours, $h, k = 1, 2,24$
Index for buses, $i = 1, 2,N$
Total number of buses
Scenarios of uncertainty
Subset of substation buses $(SS \in i)$

Parameters

C^{Peak}	Peak demand charge, \$/MW
C^{Sm}	Cost of smart charging device and communication, \$/unit
C_h^{TOU}	Time-of-Use electricity price, \$/MWh
ChD_g	Charging duration, hours
$E_{g,ct}$	Energy required to charge group g of PEVs at charging level ct
N_i^{PEV}	Number of PEVs at bus i
P_{ct}^{ChL}	Power drawn by a PEV at charging level ct , p.u.
$P_{i,h}^D, Q_{i,h}^D$	Real and reactive power demand at bus i and hour h , p.u.
V^{Min}, V^{Max}	Lower and upper limits of voltage magnitude, p.u.
$X^A_{i,k}$	Share of PEV fleet that arrive at hour k and require ChD_g charging
	duration
X_i^P	Share of PEV fleet that arrive during on-peak period, p.u.
$X_{i,g,ct}^{ChD}$	Share of PEV fleet that uses charging level ct and require charging
	duration of ChD_g , p.u.
$ ho_h$	Hourly market price, \$/MWh

η	Charging efficiency
$ heta_{i,j}$	Angle of bus admittance matrix element i - j , rad

Variables

$P_{i,j,h}^{Fdr}, Q_{i,j,h}^{Fdr}$	Real and reactive power flow from i to j at hour h , p.u.
P^{Peak}	Daily peak load imported by substation, p.u.
$P^S_{i,h}$	Total power drawn by PEV fleet in smart mode at bus i and hour h ,
	p.u.
$P^{Sm}_{i,g,h,ct}$	Power drawn by PEV fleet in smart mode of group g , using charging
	level ct , at bus i and hour h , p.u.
$P^U_{i,h}$	Total power drawn by PEV fleet in uncontrolled mode at bus i and
	hour h , p.u.
$P_{i,h,ch}^{UncOf}, P_{i,h,ch}^{UncP}$	Off-peak and on-peak power drawn by PEV fleet in uncontrolled
	mode using charging level ct , at bus i and hour h , p.u.
P_h^{SS}, Q_h^{SS}	Real and reactive power imported by LDC via substation at hour $\boldsymbol{h},$
	p.u.
$V_{i,h}$	Voltage magnitude at bus i and hour h , p.u.
X^S	Share of PEV fleet that arrive at peak hours and participate in smart
	charging program, p.u.
$\delta_{i,h}$	Voltage phase angle at bus i and hour h , rad

5.3 Proposed Framework

In this chapter, it is assumed that the PEV owners can either adopt uncontrolled charging or participate in a smart charging program by transferring the control access of their vehicle to the LDC to regulate their charging and receive incentives. The optimal participation of smart charging customers, resulting from such an incentive-driven program, and the optimal incentives that would not cause financial loss to the LDC, are determined simultaneously.

Fig. 4.1 presents the overall schematic of the proposed framework which comprises three models, the Cost of Capacity Saving Model (CCSM), the Customer Participation Model (CPM), and Optimal Incentives for Smart Charging Model (OISCM). In CCSM, a new mathematical model is proposed that quantifies the peak demand reduction accrued due to the participation of PEV customers in smart charging program and hence a relationship is determined between smart charging participation and the cost of capacity savings, achieved thereby, by the LDC. While in CPM, a model is proposed to determine the relationship between customers' participation in smart charging programs vis-a-vis the promotional incentives offered by the LDC. In the third model, OISCM, a model is proposed to determine the optimal incentives that the LDC should offer to PEV smart charging customers, that will in-turn induce the optimal smart charging participation (X^S) . The outcomes from the first two models are fed into the third model. Monte Carlo simulation is used to generate a large number of residential load profiles and PEV charging profiles, considering different charging levels, PEV types, and electricity market prices.

5.3.1 Cost of Capacity Saving Model (CCSM)

With continuous increase in the system demand from increasing penetration of PEVs, there will be need for reinforcements, or even new investments in the distribution system in the long-run. However, adequate participation of PEV owners in smart charging programs will help alleviate the peak demand growth and defer the need for upgrades. In this section, a new mathematical model is presented that quantifies the peak demand reduction accrued due to the participation of PEV customers in smart charging and hence a relationship is determined between smart charging participation and the cost of capacity savings, achieved thereby, by the LDC.

Objective Function

The objective function (J_1) seeks to minimize the LDC's operations costs, and is given as follows:

$$J1 = \sum_{h=1}^{24} \rho_h P_h^{SS} + \sum_h P^{Peak} C^{Peak} - \sum_i \sum_h C_h^{TOU} (P_{i,h}^D + P_{i,h}^U + P_{i,h}^S)$$
(5.1)

where the first term of (4.1) is the cost of purchased power from the grid at market price, the second term represents the peak demand charge, and the last term denotes the revenue earned by the LDC from sell of energy to customers at time-of-use (TOU) price which includes the uncontrolled and smart charging PEV energy sold, as well. Note that since TOU price is in force in Ontario, Canada, the same is considered in this work;



Fig. 5.1. Framework for LDC-PEV customer interaction based smart charging

however, C_h^{TOU} is generic and can represent any other price structure as well. The associated operational constraints are as follows.

Power Flow Equations

The injected power at bus i and hour h is the power drawn from the substation, net of the load, the uncontrolled, and smart PEV charging loads; and is governed by the ac power

flow equations:

$$P_{i,h}^{SS} - P_{i,h}^{U} - P_{i,h}^{S} - P_{i,h}^{D} = \sum_{j \in N} V_{i,h} V_{j,h} Y_{i,j} cos(\theta_{i,j} + \delta_{j,h} - \delta_{i,h}) \qquad \forall i \in N, \forall h \quad (5.2)$$

$$Q_{i,h}^{SS} - Q_{i,h}^{D} = -\sum_{j \in N} V_{i,h} V_{j,h} Y_{i,j} sin(\theta_{i,j} + \delta_{j,h} - \delta_{i,h}) \qquad \forall i \in N, \forall h \quad (5.3)$$

Note that (4.3) does not considers any reactive component in the PEV charging.

PEV Uncontrolled Charging Constraints

It is to be noted that X^S is considered as a portion of peak arrivals, X^P . While the share of customers not participating in smart charging program (uncontrolled charging) includes off-peak arrivals, $X^A - X^P$, and peak arrivals not participating in smart charging program, $X^P - X^S$, as follows:

$$X^{A} = \begin{cases} X^{P}, & \begin{cases} X^{S}, & Smart \\ X^{P} - X^{S}, & Uncontrolled \end{cases}$$
(5.4)
$$X^{A} - X^{P}, & Uncontrolled \end{cases}$$

where X^A is the total share of arrivals. The uncontrolled charging load at any given hour over the off-peak period (7 PM to 7 AM) [111], is denoted by the relation given below, and comprises all accumulated charging loads from $h, h - 1, h - 2, ..., h - (ChD_g + 1)$, which means all charging loads from previous hours that have not completed their charging cycle. Note that the charging duration, ChD_g , are integer values that are equal to the order of its index g, *i.e.* $ChD_1 = 1, ChD_2 = 2, ..ChD_G = G$.

$$P_{i,h,ct}^{UncOff} = \sum_{g} \sum_{\substack{h-ChD_g+1 \le k \le h\\k \in h^{OffPeak}}} X_{i,k}^A X_{i,g,ct}^{ChD} N_i^{PEV} P_{ct}^{ChL} \qquad \forall i, \forall ct, \forall h \quad (5.5)$$

For example, the uncontrolled charging load at h = 5 comprises:

Vehicles start charging at:	k = h = 5
Need charging duration of:	g = 1, 2, 3, G
Vehicles start charging at:	k = h - 1 = 4
Need charging duration of:	g = 2, 3, 4, G
:	:
Vehicles start charging at:	k = h - G + 1
Need charging duration of:	g = G

On the other hand, the uncontrolled charging arising from $(1 - X^S)$ fraction of the PEVs arriving home during on-peak hours and opting for uncontrolled charging, is given as follows:

$$P_{i,h,ct}^{UncOn} = \sum_{\substack{g \ h-ChD_g+1 \le k \le h \\ k \in h^{OnPeak}}} \sum_{\substack{X_{i,k}^A X_{i,g,ct}^{ChD} N_i^{PEV} P_{ct}^{ChL} (1-X^S)} \quad \forall i, \forall ct, \forall h \quad (5.6)$$

It is to be noted from (4.6) that only on-peak arrivals are candidates for smart charging. The total uncontrolled charging demand is hence obtained as follows:

$$P_{i,h}^{U} = \sum_{ct} P_{i,h,ct}^{UncOf} + P_{i,h,ct}^{UncOn} \qquad \forall i, \forall h$$
(5.7)

PEV Smart Charging Constraints

These constraints ensure that the total charging energy required by each PEV group g using smart charging is equal to the daily energy needed to charge the battery. In addition, these constraints ensure that the smart charging window is from the time of their home arrival till 10 AM next day.

$$\sum_{h \in h^{OffPeak}} P_{i,g,h,ct}^{Sm} = X_i^P X_{i,g,ct}^{ChD} N_i^{PEV} E_{g,ct} X^S \qquad \forall i, \forall g, \forall ct \quad (5.8)$$

where

$$X_i^P = \sum_{h \in h^{OnPeak}} X_{i,h}^A \tag{5.9}$$

Moreover, the power drawn by PEVs during any hour must be within the charging level, as follows:

$$P_{i,g,h,ct}^{Sm} \leq X_i^P X_{i,g,ct}^{ChD} N_i^{PEV} P_{ct}^{ChL} X^S \qquad \forall i, h, g, ct \quad (5.10)$$

Accordingly, the total smart charging demand is obtained as follows:

$$P_{i,h}^{S} = \sum_{ct,g} P_{i,g,h,ct}^{Sm} \qquad \forall i, \forall h \qquad (5.11)$$

Peak Load Constraints

LDCs are charged for their monthly peak demand [112] and hence it is important that the PEV charging loads do not increase the system peak demand beyond acceptable limits. The following constraint, in conjunction with (4.1), ensures that the peak demand is minimized:

$$P_{i,h}^{SS} \le P^{Peak} \qquad \forall i \in SS, \forall h \tag{5.12}$$

Voltage Limits

These constraints ensure that the voltage magnitudes at all buses are within the minimum and maximum allowable limits.

$$V^{Min} \le V_{i,h} \le V^{Max} \qquad \forall i \in N, \forall h \qquad (5.13)$$

Step-by-Step Procedure

The step-by-step procedure of quantifying the expected economic benefit to the LDC from deferral of capacity investments accrued from smart charging participation by PEV owners is as follows:.

- 1. Set the share of PEV smart charging $X^S = 10\%$.
- 2. Randomly generate the market price profile (ρ_h) , system load profiles, and PEV uncontrolled charging loads for each bus in the system.

- 3. Execute the proposed CCSM to determine the Peak Reduction Ratio (PRR), denoted by (λ) , which is the ratio of system peak demand with smart charging to that with no smart charging.
- 4. Calculate the expected value of λ , denoted by $E[\lambda]$, for multiple simulations.
- 5. Calculate the expected deferral of investment costs, in years, $(E[\Delta N])$ due to smart charging participation by PEV owners, as follows [113]:

$$E[\Delta N] = \frac{\log \frac{1}{E[\lambda]}}{\log(1+\tau)}$$
(5.14)

where τ is the annual growth rate of the system load.

6. Calculate the expected benefit accrued to the LDC from the capacity deferral using the following:

$$E[B] = C^{Inv} \left(1 - \left(\frac{1+i}{1+d}\right)^{E[\Delta N]} \right)$$
(5.15)

where C^{Inv} denotes the transformer capital cost (\$), *i* and *d* are inflation and discount rates, respectively.

- 7. Repeat steps 1- 6 and tabulate E[B] for various levels of smart charging participation by PEV owners.
- 8. Stop if $X^S = 100\%$ and estimate the relationship between X^S and E[B] using the data obtained, as follows:

$$E[B] = f^{CS}(X^S) \tag{5.16}$$

5.3.2 Customer Participation Model (CPM)

A PEV smart charging program need develop a business model between the LDC and PEV customers as they are both impacted by this relationship. To date, there is no real data or PEV customer surveys that examine the relationship between customers' participation in smart charging programs vis-a-vis the promotional incentives offered by the LDC.

In this work, customer participation is modelled as a function of the incentive provided by the LDC. A two-part incentive structure comprising a fixed component (α^{Rebate}) and a variable component (α^{Inc}), is proposed, as follows:

- α^{Inc} : PEV customers are paid for the amount of charging energy shifted through smart charging control (\$/kWh).
- α^{Rebate} : PEV customers are paid a portion of the fixed cost of the smart charging device to be installed (\$).

The expected annual economic benefit (E[AEB]) for PEV customers can be estimated from the reduction in the annual energy cost based on the incentive paid by the LDC, α^{Inc} , and the rebate received for smart charging device, α^{Rebate} , and can be expressed as follows:

$$E[AEB] = E[E^{PEV}]\alpha^{Inc} - (C^{Sm} - \alpha^{Rebate})$$
(5.17)

where $E[E^{PEV}]$ is the expected annual PEV charging energy and C^{Sm} is the annualized cost of smart charging device. It is assumed that X^S is proportional to E[AEB], normalized with respect to its PEV charging cost in 100% uncontrolled mode, as given below:

$$X^{S} = \frac{E[AEB]}{E[C^{Chg}]} \tag{5.18}$$

Therefore, from (4.18) it is noted that if the incentives are very low, or null, E[AEB] will be very low or negative, and $X^S = \text{zero}$; while if E[AEB] covers the entire annual charging cost $(E[C^{Chg}])$, *i.e.*, E[AEB] = $E[C^{Chg}]$, the participation is 100%.

5.3.3 Optimal Incentives for Smart Charging Model (OISCM)

In this section, a novel mathematical model is proposed to determine the optimal incentives $(\alpha^{Inc} \text{ and } \alpha^{Rebate})$, that the LDC should offer to PEV smart charging customers, that will in-turn induce the optimal smart charging participation (X^S) .

Objective Function

The objective function (J_2) considered for minimization, is a comprehensive representation of the LDC's cost of system operation, promotion and adoption of PEV smart charging infrastructure, as well as the long-term economic benefit from capacity deferral accrued from such policy implementation, and is given as follows:

$$J2 = \sum_{h=1}^{24} \rho_h P_h^{SS} \quad \textcircled{a}$$

$$+ \sum_i X^S N_i^{PEV} \alpha^{Rebate} \quad \textcircled{b}$$

$$+ \sum_i \sum_h \alpha^{Inc} P_{i,h}^S \quad \textcircled{c}$$

$$+ \sum_h P^{Peak} C^{Peak} \quad \textcircled{d}$$

$$- \sum_i \sum_h C_h^{TOU} (P_{i,h}^D + P_{i,h}^U + P_{i,h}^S) \quad \textcircled{e}$$

$$- f^{CS} (X^S) \quad \textcircled{f}$$

$$(5.19)$$

where

- *a* Purchased power from grid at market price.
- b Rebate paid to customers for adopting smart charging devices
- c Incentives paid to customers participating in smart charging program
- d Peak demand charge incurred by LDC (\$/kW)
- *e* Revenue from customers for energy sold at TOU price
- f Cost of capacity saving, obtained from Section-4.3.1, and given by (4.16).

Customers' Participation Rate Constraint

Applying (4.17) in (4.18), the customers' participation in smart charging (X^S) can be modelled as follows:

$$X^S = A\alpha^{Inc} + B\alpha^{Rebate} + C \tag{5.20}$$

The objective function (4.19), constraints (4.2)-(4.3), (4.5)-(4.13) and (4.20) comprises the OISCM, which is a non-linear programming (NLP) model. A Monte Carlo simulation of the load profile, considering a range of PEV data (such as charging duration, starting time of charging), and electricity market price, is carried out to arrive at a large number of input profiles, which are then fed into the OISCM to determine the expected smart charging participation $(E[X^S])$ and the expected incentives.

5.4 Test System and Assumptions

The distribution system under study comprises 33 buses in radial configuration [87]. The total system peak demand is 4.54 MVA. The network parameters and the load data are given in the Appendix. Three case studies with different PEV market penetrations, namely 25%, 50%, and 75%, have been considered. In order to evaluate the optimal PEV smart charging participation and the optimal incentives paid to customers, the following assumptions are made:

- The IEEE RTS [102] base load profile is used; the randomness of bus-wise and hourly loads are simulated using Gaussian distributions with a standard deviation of 0.15 p.u. around the nominal.
- Daily trip data was extracted from the US National Household Travel Survey (NHTS) 2009 [86] of a sample size of 150,147 households, and 309,164 vehicles, for different type of vehicles, and trips, to obtain the probability distribution function (pdf) of daily mileage driven and home arrival times of the PEVs (Fig.4.2 and 4.3). Thereafter, Monte Carlo simulations are carried out to generate random trip mileage and home arrival times.



Fig. 5.2. Probability distribution function for home last arrival time



Fig. 5.3. Probability distribution function for daily mileage driven

- Eleven common makes of PEVs are considered, which covers 95% of the total market of PEVs in USA, and these are grouped into four classes (Table 3.1).
- Two residential charging levels are considered (Level-1 and Level-2) which have been used in several studies, with shares of 65% and 35%, respectively [101].
- The market shares of the PEV classes are uncertain, and so is the mix of the charging level. Monte Carlo simulations are carried out considering Gaussian distributions with a standard deviation of 0.05 p.u.
- The charging efficiency is 90%.
- The entire load is residential, there are two vehicles per house and PEV charging occurs only at home. The number of houses at a bus is calculated assuming the house peak load to be 2.08 kW. Therefore, the number of bus at each bus is known.
- Given the battery life-cycle considerations, it is assumed that the maximum allowable depth of discharge is 70%. [60].
- The daily market price profiles for 2015 of Hourly Ontario Energy Price (HOEP), and Ontario TOU electricity price are considered for the simulations. The uncertainty in HOEP is simulated using Gaussian distribution with standard deviation of 0.15 p.u.
- The cost of a smart charging device, C^{Sm} , is assumed to be \$150 [114]; the annualized cost considering a life of five years and 10% discount rate is obtained to be \$40.

Furthermore, C^{Peak} is assumed to be 6.55\$/kW which is the provincial transmission service charge in Ontario. For calculating the deferrals, (4.14)- (4.15), annual growth rate (τ) = 1.5%; and inflation rate (i) = 1.5% are assumed.

5.5 **Results And Discussions**

5.5.1 Cost of Capacity Saving Function

Table-4.2 presents the impact on the expected PRR $(E[\lambda])$, expected capacity deferral in years (ΔN) , and the equivalent economic benefits (E[B]) accrued to the system, with increasing X^S for varying degrees of market penetration of PEVs, obtained using the CCSM. Since the peak load increases as X^S increases in the three cases, the deferred substation capacity upgrades will also increase; and are assumed to be 3, 4, and 5 MW, and the associated deferred investment costs are \$350,000, \$400,000, and \$450,00, respectively.

The expected PRR $(E[\lambda])$ and the expected economic benefit due to investment deferral E[B] are presented in Fig. 4.4 and Fig. 4.5, respectively. As seen from Fig. 4.4 and Table-4.2, the PRR decreases with the increase in X^S , attains a minimum, and then increases, with further increase in X^S because a new peak is now created at off-peak periods with high participation of PEVs in smart charging. This trend is observed for all the three cases, but in Case 3, the expected PRR attains the lowest value amongst the three, and the trend is more pronounced, because of the high market penetration of PEVs.

As seen from Table-4.2 and Fig. 4.5, the capacity deferral and hence the expected benefit accrued are proportional to X^S . A linear curve fit function of E[B] versus X^S is also presented in Fig. 4.5, which is used in the OISCM.

5.5.2 Customer Participation Rate

The CPM discussed earlier, is solved to determine the PEV customers participation in the smart charging program as a function of the incentive (α^{Inc}) and rebate (α^{Rebate}) paid by the LDC. As discussed in Section-II.B, the first step is to determine E[AEB] as per (14) and $E[C^{Chg}]$ in uncontrolled mode. This is done by generating a number of scenarios on daily mileage driven and where the start time of charging is based on arrive and plug, and using the pdfs obtained earlier.

In Table-4.3, it is noted that $E[C^{Chg}]$ increases with increase in battery capacity and Level-2 charging leads to higher energy cost as compared to Level-1 for the same PEV

TABLE 5.2					
EXPECTED PRR, CAPACITY I	Deferral, and Benefit fo	or the Considered	CASE STUDIES		

	Case-1	(25% PEV	Market l	Penetration)	Case-2	(50% PEV)	Market I	Penetration)	Case-3	(75% PEV	Market I	Penetration)
$\begin{array}{c} X^S \\ (\text{p.u.}) \end{array}$	$\begin{array}{c} E[\lambda] \\ (p.u.) \end{array}$	ΔN (Years)	E[B] (\$)	Annualized E[B] (\$)	$\begin{array}{c} E[\lambda] \\ (p.u.) \end{array}$	ΔN (Years)	E[B] (\$)	Annualized E[B] (\$)	$\begin{array}{c} E[\lambda] \\ (p.u.) \end{array}$	ΔN (Years)	E[B] (\$)	Annualized E[B] (\$)
0	1	0	0	0	1	0	0	0	1	0	0	0
0.1	0.982	1.2	29143	7492	0.968	2.2	58088	14934	0.961	2.7	77737	19986
0.2	0.965	2.4	55158	14181	0.937	4.3	106572	27399	0.923	5.4	143156	36804
0.3	0.948	3.6	78309	20133	0.911	6.2	143540	36903	0.890	7.8	192734	49551
0.4	0.936	4.4	95002	24424	0.893	7.6	167296	43011	0.868	9.5	222093	57098
0.5	0.925	5.2	108730	27954	0.880	8.6	183426	47157	0.853	10.7	240227	61761
0.6	0.917	5.8	119161	30635	0.871	9.3	193447	49734	0.843	11.4	250854	64493
0.7	0.911	6.3	126147	32431	0.866	9.7	199442	51275	0.839	11.8	255739	65749
0.8	0.906	6.6	131275	33750	0.863	9.9	202203	51985	0.840	11.7	255178	65604
0.9	0.904	6.8	134427	34560	0.864	9.8	201368	51770	0.849	11.0	244600	62885
1	0.902	6.9	136134	34999	0.870	9.4	195114	50162	0.869	9.4	220282	56633



Fig. 5.4. Expected PRR $(E[\lambda])$ versus smart charging participation

class. This is due to the shorter charging duration, compared to Level-1 users, and high arrival rate during on-peak period which means a significant portion of charging occurs during on-peak TOU price. On the other hand, Level-1 users need longer charging duration and hence a major portion of their charging occurs during off-peak periods. The weighted average values for the considered PEV classes are obtained as shown in the last row of Table-4.3, which are then used to calculate the parameters of the customer participation function (4.18). The estimated customers' participation function can be written as follows:

$$X^{S} = 8.7895\alpha^{Inc} + 0.0038\alpha^{Rebate} - 0.1489$$
(5.21)



Fig. 5.5. Expected benefit versus smart charging participation

Fig. 4.6 presents the variation in customer participation with changes in incentive and rebate, as obtained in (4.21). It is noted that α^{Inc} has a more pronounced effect on X^S as compared to α^{Rebate} .

TABLE 5.3 Expected Annual Energy, Energy Cost, and Average Cost for the Considered PEV Classes

	$E[C^{Chg}] (\$)$	$E[E^{PEV}]$ (kWh)	Expected Annual Average Cost \$/kWh
PHEV-I (Level-1)	162	1,388	0.117
PHEV-I (Level-2)	171	1,383	0.124
PHEV-II (Level-1)	293	2,710	0.108
PHEV-II (Level-2)	340	2,757	0.123
BEV-I (Level-1)	271	2,496	0.109
BEV-I (Level-2)	303	2,513	0.121
BEV-II (Level-1)	296	2,743	0.108
BEV-II (Level-2)	391	3,348	0.117
Weighted Average Values	266	2,338	0.114


Fig. 5.6. Customer participation with respect to incentive and rebate

5.5.3 Optimal Share of PEV Smart Charging

The optimal value of α^{Inc} , α^{Rebate} , and X^S of PEV customers are now determined using the proposed OISCM which includes the outcomes from CCSM and CPM, discussed earlier. The expected values of these variables are arrived at, after convergence of the Monte Carlo process, as presented in Table-4.4 for the different cases considered. Fig. 4.7 shows the convergence to the expected value, obtained from the proposed framework, in Case-3.

It can be seen from Table-4.4 that there is no significant difference in terms of α^{Inc} , α^{Rebate} , and X^S for the different cases of market penetrations of PEVs. However, there will be a proportionate increase in the absolute number of PEVs participating in the smart charging program as market penetration of PEVs increase. The optimal incentives obtained are 3, 2.9, and 2.9 cents per kWh, which implies about 33% savings in annual PEV charging cost since they will be credited for every kWh controlled and PEV load will be shifted to the off-peak period and hence charged at off-peak prices.

Fig. 4.8 presents the resultant profiles for the considered PEV market penetration cases. It can be seen that there is a proportional increase in the demand of uncontrolled and smart charging with the increase in the market penetration of PEVs.

 TABLE 5.4

 Optimal Smart Charging Share, Incentives, and Rebate for the Considered Case Studies

	Optimal Smart Charging Share (X^S) , %	Optimal number of PEVs	Optimal Incentive, cents	Optimal Rebate, \$
Case-1 (25% PEV Pen.)	14	104	3	22
Case-2 (50% PEV Pen.)	13	193	2.9	23
Case-3 (75% PEV Pen.)	12.9	287	2.9	22



Fig. 5.7. Convergence of OISCM Model for Case-3

5.6 Conclusions

This chapter presented a novel smart charging control algorithm to mitigate the additional PEV charging load. The proposed algorithm presented the PEV owners with the option to either adopt uncontrolled charging or participate in a smart charging program by transferring control access of their vehicle to the LDC to regulate their charging load and receive incentives. The optimal participation of smart charging customers, resulting from such an incentive-driven program, and the optimal incentives that would not cause financial loss to the LDC, were determined simultaneously. The model incorporated vehicle driver habits, various vehicle types and charging levels in order to capture the charging pattern of PEV owners. In addition, Monte Carlo simulations were used to consider various uncertainties. Three case studies were presented to demonstrate the performance of the proposed model, which can be used by LDCs to assess and quantify the optimal participation of smart charging customers and decide on the right incentives to be paid to them.



Fig. 5.8. Expected typical, uncontrolled, smart, and aggregated load profiles for the considered case studies

Chapter 6

Conclusions

6.1 Summary and Conclusions

In summary, the goal of this research was to develop models to investigate and address the problem of distribution system planning in the presence of PEV charging loads. The motivations for this research, and review of associated literature, which laid out the main research objectives, were presented in Chapter 1.

In Chapter 2, the relevant topics related to the research have been discussed. A background to distribution systems was discussed, followed by traditional distribution system planning, and distribution system planning in the presence of DERs and DGs. This section also included a discussion on supply adequacy of distribution systems and DR. Thereafter, a discussion on PEV types, and PEV load characteristics was presented.

In Chapter 3, a comprehensive framework for multi-year distribution system planning was presented. The proposed framework aimed to determine the optimal upgrade plan, siting and sizing of DGs, substations, capacitors and feeders in distribution systems, considering the penetration of PEVs in an uncontrolled as well as smart charging environment, and DR options. Two criteria, BCR and adequacy analysis, were incorporated within the proposed framework, to determine the optimal distribution system plan. The uncontrolled PEV charging load model considered driver behaviour, arrival time, mileage, and other information to develop a charging load profile for inclusion in the planning problem. A novel set of PEV operational constraints were included in the planning model, to capture PEV smart charging decisions. The proposed planning framework was applied to two test systems. Four case studies were considered to investigate the impact of PEV smart and uncontrolled charging loads as well as DR options on the distribution plan.

Chapter 4 presented a novel framework to assess and determine the level of uncontrolled and smart PEV penetration that a distribution system can accommodate without any additional investments or upgrades. The model incorporated vehicle driver habits, various vehicle types and charging levels in order to capture the charging pattern of PEV owners. In addition, Monte Carlo simulation was carried out to take various uncertainties into account. Various case studies were presented to demonstrate the performance of the proposed model, which can be used by LDCs to quantify the impacts, decide on the right actions and policies, and the required upgrades in their distribution networks.

In Chapter 5, a novel smart charging control algorithm to mitigate the effect of additional PEV charging load was proposed. The proposed algorithm presented the PEV owners with the option to either adopt uncontrolled charging or participate in a smart charging program by transferring control access of their vehicle to the LDC to regulate their charging load and receive incentives. The optimal participation of smart charging customers, resulting from such an incentive-driven program, and the optimal incentives that would not cause financial loss to the LDC, were determined simultaneously. The model incorporated vehicle driver habits, various vehicle types and charging levels in order to capture the charging pattern of PEV owners. In addition, Monte Carlo simulations were used to consider various uncertainties. Three case studies were presented to demonstrate the performance of the proposed model, which can be used by LDCs to assess and quantify the optimal participation of smart charging customers and decide on the right incentives to be paid to them.

The following conclusions can be drawn from the thesis:

• The studies revealed that it was important for planners to take into consideration the effects of PEV penetration while deriving their plan outcomes, particularly, when the share of uncontrolled PEV charging is high. It was noted that the present worth of the plan cost in the PEV uncontrolled charging case was much higher than that in the Base Case (No PEV), while the impact of PEV penetration was much damped in terms of added cost and added capacity when smart charging was considered. Compared to the Base Case, with uncontrolled PEV charging loads, there was a 77% and 63% increase in the present worth of the plan cost and in the added capacity (in MW), respectively. The reason behind this significant increase was that the system peak demand coincided with the uncontrolled charging demand due to the coincident home arrival rates of vehicles during peak hours. On the other hand, it was noted that there was only a 11% and 24% increase in the present worth of plan cost and in the added capacity (in MW), respectively, when smart charging was used. There was a decrease in peak demand compared to uncontrolled charging because of smart charging which scheduled the charging demand to off-peak periods. In addition, different optimal size and site of DG units were obtained, when PEV charging loads were considered.

- Current TOU electricity prices were designed to reduce the system peak load and flatten the load profile, and bring about many benefits to the system. However, the research results reveal that in the presence of PEV charging loads, TOU prices can adversely impact the loadability of the distribution system and hence the uncontrolled PEV penetration.
- It is noted that the charging level is clearly a major factor that affects the uncontrolled PEV charging load profile. From a customers' perspective, a higher charging level is preferable because it requires shorter time to charge the vehicle. It is also worth noting that there is enough margin for the whole fleet for smart charging. Unlike typical loads, PEVs are considered more flexible and elastic, and can be controlled by LDCs with the right incentives, which could be beneficial for LDCs without affecting customers convenience.
- LDCs need to monitor PEV sales and upcoming PEV technologies to prepare effective plans for required infrastructure, smart charging programs, incentives, and appropriate electricity tariff. In addition, an effective interaction between PEV companies and LDCs should be established to help forecasting future PEV penetration, PEV technologies and allocating and offering new PEV owners with proper incentives to participate in smart charging programs.

6.2 Contributions

The main contributions of this research can be summarized as follows:

• A novel, comprehensive, multi-year planning framework was developed that simultaneously determine the optimal sizing, placement and investment timelines of various resource alternatives for LDCs such as DG sources, DR option, substations, capacitors, and feeders. A novel set of PEV operational constraints in the distribution planning model was developed, to capture the impact of PEV smart charging loads. In addition, a model for estimating the uncontrolled PEV charging load profiles considering real transportation data including driver behaviour, arrival time, mileage, and other information was developed and included in the distribution planning problem.

- A novel adequacy check module was developed in the planning process to ensure that a desired level of reserve margin was maintained in the distribution system.
- A generic and novel framework to assess the DSLM was developed to accommodate PEV charging loads, *i.e.* how much additional PEV uncontrolled and smart charging load the distribution system can accommodate, without requiring any capacity reinforcement. Vehicle mobility data was used in a novel manner to develop the uncontrolled PEV charging load profile considering the uncertainties associated with drivers' behaviour (arrival time and mileage driven), PEV market share, and share of charging level. In addition, different charging scenarios have been considered to assess the impact of TOU electricity price on the starting time of charging.
- A generic and novel framework was proposed to determine the optimal incentives for smart charging considering LDC-customer interactions. The framework builds upon two aspects, the relationship between the incentives offered by the LDC and the participation of PEV customers (X^S); and the relationship between X^S and the long-term economic benefit of capacity deferral accrued from smart charging. Mathematical models were developed to represent these inter-relationships, for the first time. The proposed framework included a new mathematical model along with a step-by-step approach to determine the economic benefits to the LDC from deferral of capacity investments. In addition, a two-part incentive structure was proposed for the first time, for PEV customers to adopt smart charging, and the relation of PEV participation with this incentive structure was modelled. Finally, a new mathematical model is developed to simultaneously determine the optimal participation of PEV customers in the smart charging program and the optimal incentives to be offered by the LDC, taking into consideration the distribution system operational aspects.

6.3 Future Work

Further research can be conducted based on the work presented in this thesis. Some ideas are presented below:

- The issues of power pricing, bidding, and risk management are not considered in this thesis. The price at which the LDC purchases power from the external grid is assumed to be known a priori, the LDC does not participate in the wholesale electricity market to purchase this power, but has a standing power purchase contract. These issues are very interesting and considering the retail market competition aspects can be taken up as a future work.
- In Chapter 3, dispatchable DG (natural gas turbine) units were considered for the studies. The impact of renewable based DG units, such as wind and solar, considering Ontario FIT program on the planning decisions need be examined.
- The utilization of V2G and ancillary services such as frequency regulation through smart charging program are not considered in the proposed framework, in Chapter 5. It may be useful to examine such services which will effect the benefit to the LDC and hence the incentives and smart charging share.
- It was assumed in this thesis that PEV charging occurred only at home; while charging at work, parking lots, and fast charging stations were not considered, which could be possible avenues of future work.
- In the presence of PEV charging loads, current TOU electricity prices can adversely impact the loadability of the distribution system and hence the uncontrolled PEV penetration. Therefore, further research should be undertaken to investigate the optimal TOU pricing scheme or different pricing schemes considering the penetration of PEV loads.

References

- T. Ackermann, G. Andersson, and L. Söder, "Distributed generation: a definition," Electric power systems research, vol. 57, no. 3, pp. 195–204, 2001.
- [2] B. Global and B. Worldwide, "Bp energy outlook 2035," 2015.
- [3] Natural Resources Canada. (2016, Last Accessed:30-03-2017) About renewable energy. [Online]. Available: https://www.nrcan.gc.ca/energy/renewable-electricity/ 7295#wind
- [4] International Energy Agency (iea). (2016, Last Accessed:02-03-2017) Global EV outlook 2016. [Online]. Available: https://www.iea.org/publications/ freepublications/publication/Global_EV_Outlook_2016.pdf
- [5] Ministry of the Environment and Climate Change. (2016, Last Accessed:02-03-2017) Ontarios five year climate change action plan 2016 - 2020. [Online]. Available: http://www.applications.ene.gov.on.ca/ccap/products/CCAP_ENGLISH.pdf
- [6] S. K. Khator and L. C. Leung, "Power distribution planning: a review of models and issues," *IEEE Transactions on Power Systems*, vol. 12, no. 3, pp. 1151–1159, 1997.
- [7] O. M. Mikic, "Mathematical dynamic model for long-term distribution system planning," *IEEE transactions on power systems*, vol. 1, no. 1, pp. 34–40, 1986.
- [8] M. Ponnavaikko and K. P. Rao, "Optimal distribution system planning," IEEE Transactions on Power Apparatus and Systems, no. 6, pp. 2969–2977, 1981.
- [9] R. Adams and M. Laughton, "Optimal planning of power networks using mixedinteger programming. part 1: Static and time-phased network synthesis," *Electrical Engineers, Proceedings of the Institution of*, vol. 121, no. 2, pp. 139–147, 1974.

- [10] D. M. Crawford and S. B. Holt, "A mathematical optimization technique for locating and sizing distribution substations, and deriving their optimal service areas," *IEEE Transactions on Power Apparatus and Systems*, vol. 94, no. 2, pp. 230–235, 1975.
- [11] D. Wall, G. Thompson, and J. Northcote-Green, "An optimization model for planning radial distribution networks," *IEEE Transactions on Power Apparatus and Systems*, vol. 3, no. PAS-98, pp. 1061–1068, 1979.
- [12] T. Fawzi, K. Ali, and S. El-Sobki, "A new planning model for distribution systems," *IEEE Transactions on Power Apparatus and Systems*, no. 9, pp. 3010–3017, 1983.
- [13] G. L. Thompson and D. Wall, "A branch and bound model for choosing optimal substation locations," *IEEE Transactions on Power Apparatus and Systems*, no. 5, pp. 2683–2688, 1981.
- [14] V. Quintana, H. Temraz, and K. Hipel, "Two-stage power-system-distributionplanning algorithm," in *IEE Proceedings C-Generation*, Transmission and Distribution, vol. 140, no. 1. IET, 1993, pp. 17–29.
- [15] K. Aoki, K. Nara, T. Satoh, M. Kitagawa, and K. Yamanaka, "New approximate optimization method for distribution system planning," *IEEE Transactions on power* systems, vol. 5, no. 1, pp. 126–132, 1990.
- [16] K. Nara, T. Satoh, K. Aoki, and M. Kitagawa, "Multi-year expansion planning for distribution systems," *IEEE Transactions on Power Systems*, vol. 6, no. 3, pp. 952– 958, 1991.
- [17] K. Nara, T. Satoh, H. Kuwabara, K. Aoki, M. Kitagawa, and T. Ishihara, "Distribution systems expansion planning by multi-stage branch exchange," *IEEE transactions on power systems*, vol. 7, no. 1, pp. 208–214, 1992.
- [18] H. L. Willis, "Analytical methods and rules of thumb for modeling dg-distribution interaction," in *Power Engineering Society Summer Meeting*, 2000. IEEE, vol. 3. IEEE, 2000, pp. 1643–1644.
- [19] C. Wang and M. H. Nehrir, "Analytical approaches for optimal placement of distributed generation sources in power systems," *IEEE Transactions on Power* Systems, vol. 19, no. 4, pp. 2068–2076, 2004.
- [20] A. Keane and M. O'Malley, "Optimal allocation of embedded generation on distribution networks," *IEEE Transactions on Power Systems*, vol. 20, no. 3, pp. 1640–1646, 2005.

- [21] A. Keane, Q. Zhou, J. W. Bialek, and M. O'Malley, "Planning and operating nonfirm distributed generation," *IET Renewable Power Generation*, vol. 3, no. 4, pp. 455–464, 2009.
- [22] Y. Atwa and E. El-Saadany, "Probabilistic approach for optimal allocation of wind-based distributed generation in distribution systems," *IET Renewable Power Generation*, vol. 5, no. 1, pp. 79–88, 2011.
- [23] L. F. Ochoa and G. P. Harrison, "Minimizing energy losses: Optimal accommodation and smart operation of renewable distributed generation," *IEEE Transactions on Power Systems*, vol. 26, no. 1, pp. 198–205, 2011.
- [24] A. Kumar and W. Gao, "Optimal distributed generation location using mixed integer non-linear programming in hybrid electricity markets," *IET generation, transmission* & distribution, vol. 4, no. 2, pp. 281–298, 2010.
- [25] G. Celli, E. Ghiani, S. Mocci, and F. Pilo, "A multiobjective evolutionary algorithm for the sizing and siting of distributed generation," *IEEE Trans. Power Syst.*, vol. 20, no. 2, pp. 750–757, 2005.
- [26] M. F. Shaaban, Y. M. Atwa, and E. F. El-Saadany, "DG allocation for benefit maximization in distribution networks," *IEEE Trans. Power Syst.*, vol. 28, no. 2, pp. 639–649, 2013.
- [27] D. Singh, D. Singh, and K. Verma, "Multiobjective optimization for dg planning with load models," *IEEE transactions on power systems*, vol. 24, no. 1, pp. 427–436, 2009.
- [28] W. El-Khattam, Y. Hegazy, and M. Salama, "An integrated distributed generation optimization model for distribution system planning," *IEEE Transactions on Power* Systems, vol. 20, no. 2, pp. 1158–1165, 2005.
- [29] S. Wong, K. Bhattacharya, and J. D. Fuller, "Electric power distribution system design and planning in a deregulated environment," *IET Gener. Transm. Distrib.*, vol. 3, no. 12, pp. 1061–1078, 2009.
- [30] V. F. Martins and C. L. Borges, "Active distribution network integrated planning incorporating distributed generation and load response uncertainties," *IEEE Transactions on Power Systems*, vol. 26, no. 4, pp. 2164–2172, 2011.

- [31] E. Naderi, H. Seifi, and M. S. Sepasian, "A dynamic approach for distribution system planning considering distributed generation," *IEEE Trans. Power Del.*, vol. 27, no. 3, pp. 1313–1322, 2012.
- [32] K. Zou, A. P. Agalgaonkar, K. M. Muttaqi, and S. Perera, "Distribution system planning with incorporating dg reactive capability and system uncertainties," *IEEE Transactions on Sustainable Energy*, vol. 3, no. 1, pp. 112–123, 2012.
- [33] A. Zidan, M. F. Shaaban, and E. F. El-Saadany, "Long-term multi-objective distribution network planning by dg allocation and feeders reconfiguration," *Electric Power Systems Research*, vol. 105, pp. 95–104, 2013.
- [34] S. Haffner, L. F. A. Pereira, L. A. Pereira, and L. S. Barreto, "Multistage model for distribution expansion planning with distributed generationpart i: Problem formulation," *IEEE Transactions on Power Delivery*, vol. 23, no. 2, pp. 915–923, 2008.
- [35] —, "Multistage model for distribution expansion planning with distributed generationpart ii: Numerical results," *IEEE Transactions on Power Delivery*, vol. 23, no. 2, pp. 924–929, 2008.
- [36] N. Kanwar, N. Gupta, K. Niazi, A. Swarnkar, and R. Bansal, "Simultaneous allocation of distributed energy resource using improved particle swarm optimization," *Applied Energy*, 2016.
- [37] M. Kalantari and A. Kazemi, "Placement of distributed generation unit and capacitor allocation in distribution systems using genetic algorithm," in *Environment and Electrical Engineering (EEEIC)*, 2011 10th International Conference on. IEEE, 2011, pp. 1–5.
- [38] P. Kayal, T. Ashish, and C. K. Chanda, "Simultaneous placement and sizing of renewable dgs and capacitor banks in distribution network," in *Circuit, Power and Computing Technologies (ICCPCT), 2014 International Conference on.* IEEE, 2014, pp. 607–611.
- [39] I. Ziari, G. Ledwich, A. Ghosh, and G. Platt, "Optimal distribution network reinforcement considering load growth, line loss, and reliability," *IEEE Transactions* on Power Systems, vol. 28, no. 2, pp. 587–597, 2013.

- [40] R.-C. Leou, C.-L. Su, and C.-N. Lu, "Stochastic analyses of electric vehicle charging impacts on distribution network," *IEEE Trans. Power Syst.*, vol. 29, no. 3, pp. 1055– 1063, 2014.
- [41] M. K. Gray and W. G. Morsi, "Power quality assessment in distribution systems embedded with plug-in hybrid and battery electric vehicles," *IEEE Trans. Power* Syst., vol. 30, no. 2, pp. 663–671, 2015.
- [42] S. Shafiee, M. Fotuhi-Firuzabad, and M. Rastegar, "Investigating the impacts of plug-in hybrid electric vehicles on power distribution systems," *IEEE Trans. Smart Grid*, vol. 4, no. 3, pp. 1351–1360, 2013.
- [43] K. Clement-Nyns, E. Haesen, and J. Driesen, "The impact of charging plug-in hybrid electric vehicles on a residential distribution grid," *IEEE Trans. Power Syst.*, vol. 25, no. 1, pp. 371–380, 2010.
- [44] M. S. ElNozahy, M. M. Salama *et al.*, "A comprehensive study of the impacts of pheves on residential distribution networks," *IEEE Trans. Sustain. Energy*, vol. 5, no. 1, pp. 332–342, 2014.
- [45] E. Sortomme, M. M. Hindi, S. J. MacPherson, and S. Venkata, "Coordinated charging of plug-in hybrid electric vehicles to minimize distribution system losses," *IEEE Trans. Smart Grid*, vol. 2, no. 1, pp. 198–205, 2011.
- [46] J. de Hoog, T. Alpcan, M. Brazil, D. A. Thomas, and I. Mareels, "Optimal charging of electric vehicles taking distribution network constraints into account," *IEEE Trans. Power Syst.*, vol. 30, no. 1, pp. 365–375, 2015.
- [47] I. Sharma, C. Canizares, and K. Bhattacharya, "Smart charging of pevs penetrating into residential distribution systems," *IEEE Trans. Smart Grid*, vol. 5, no. 3, pp. 1196–1209, 2014.
- [48] M. F. Shaaban, M. Ismail, E. F. El-Saadany, and W. Zhuang, "Real-time pev charging/discharging coordination in smart distribution systems," *IEEE Trans. Smart Grid*, vol. 5, no. 4, pp. 1797–1807, 2014.
- [49] L. Hua, J. Wang, and C. Zhou, "Adaptive electric vehicle charging coordination on distribution network," *IEEE Trans. Smart Grid*, vol. 5, no. 6, pp. 2666–2675, 2014.
- [50] J. G. Vlachogiannis, "Probabilistic constrained load flow considering integration of wind power generation and electric vehicles," *IEEE Trans. Power Syst.*, vol. 24, no. 4, pp. 1808–1817, 2009.

- [51] E. Veldman, R. Verzijlbergh *et al.*, "Distribution grid impacts of smart electric vehicle charging from different perspectives," *IEEE Trans. Smart Grid*, vol. 6, no. 1, pp. 333–342, 2015.
- [52] E. Sortomme, M. El-Sharkawi et al., "Optimal charging strategies for unidirectional vehicle-to-grid," *IEEE Trans. Smart Grid*, vol. 2, no. 1, pp. 131–138, 2011.
- [53] M. G. Vay and G. Andersson, "Self scheduling of plug-in electric vehicle aggregator to provide balancing services for wind power," *IEEE Transactions on Sustainable Energy*, vol. 7, no. 2, pp. 886–899, April 2016.
- [54] R. C. Green, L. Wang, and M. Alam, "The impact of plug-in hybrid electric vehicles on distribution networks: A review and outlook," *Renewable and Sustainable Energy Reviews*, vol. 15, no. 1, pp. 544–553, 2011.
- [55] W. Yao, J. Zhao, F. Wen, Z. Dong, Y. Xue, Y. Xu, and K. Meng, "A multi-objective collaborative planning strategy for integrated power distribution and electric vehicle charging systems," *IEEE Trans. Power Syst.*, vol. 29, no. 4, pp. 1811–1821, 2014.
- [56] M. F. Shaaban and E. F. El-Saadany, "Accommodating high penetrations of PEVs and renewable DG considering uncertainties in distribution systems," *IEEE Trans. Power Syst.*, vol. 29, no. 1, pp. 259–270, 2014.
- [57] M. F. Shaaban, Y. M. Atwa, and E. F. El-Saadany, "Pevs modeling and impacts mitigation in distribution networks," *IEEE Trans. Power Syst.*, vol. 28, no. 2, pp. 1122–1131, 2013.
- [58] Z. Liu, F. Wen, and G. Ledwich, "Optimal siting and sizing of distributed generators in distribution systems considering uncertainties," *IEEE Trans. Power Del.*, vol. 26, no. 4, pp. 2541–2551, 2011.
- [59] A. H. Hajimiragha, C. Cañizares, M. W. Fowler, S. Moazeni, A. Elkamel *et al.*, "A robust optimization approach for planning the transition to plug-in hybrid electric vehicles," *IEEE Trans. Power Syst.*, vol. 26, no. 4, pp. 2264–2274, 2011.
- [60] A. Hajimiragha, C. Cañizares, M. W. Fowler, A. Elkamel *et al.*, "Optimal transition to plug-in hybrid electric vehicles in ontario, canada, considering the electricity-grid limitations," *IEEE Trans. Ind. Electron.*, vol. 57, no. 2, pp. 690–701, 2010.
- [61] D. Steen, L. A. Tuan, O. Carlson, and L. Bertling, "Assessment of electric vehicle charging scenarios based on demographical data," *IEEE Trans. Smart Grid*, vol. 3, no. 3, pp. 1457–1468, 2012.

- [62] S. Han, S. Han, and K. Sezaki, "Development of an optimal vehicle-to-grid aggregator for frequency regulation," *IEEE Transactions on smart grid*, vol. 1, no. 1, pp. 65–72, 2010.
- [63] T. Ma and O. Mohammed, "Economic analysis of real-time large scale pevs network power flow control algorithm with the consideration of v2g services," in *Industry Applications Society Annual Meeting*, 2013 IEEE. IEEE, 2013, pp. 1–8.
- [64] S. Shao, M. Pipattanasomporn, and S. Rahman, "Demand response as a load shaping tool in an intelligent grid with electric vehicles," *IEEE Transactions on Smart Grid*, vol. 2, no. 4, pp. 624–631, 2011.
- [65] A. D. Hilshey, P. D. Hines, P. Rezaei, and J. R. Dowds, "Estimating the impact of electric vehicle smart charging on distribution transformer aging," *IEEE Transactions on Smart Grid*, vol. 4, no. 2, pp. 905–913, 2013.
- [66] Y. Mou, H. Xing, Z. Lin, and M. Fu, "Decentralized optimal demand-side management for phev charging in a smart grid," *IEEE Transactions on Smart Grid*, vol. 6, no. 2, pp. 726–736, 2015.
- [67] S. M. Mohammadi-Hosseininejad, A. Fereidunian, A. Shahsavari, and H. Lesani, "A healer reinforcement approach to self-healing in smart grid by phevs parking lot allocation," *IEEE Transactions on Industrial Informatics*, vol. 12, no. 6, pp. 2020– 2030, 2016.
- [68] S. Gao, K. Chau, C. Liu, D. Wu, and C. Chan, "Integrated energy management of plug-in electric vehicles in power grid with renewables," *IEEE Transactions on Vehicular Technology*, vol. 63, no. 7, pp. 3019–3027, 2014.
- [69] Z. Luo, Z. Hu, Y. Song, Z. Xu, H. Liu, L. Jia, and H. Lu, "Economic analyses of plug-in electric vehicle battery providing ancillary services," in *Electric Vehicle Conference (IEVC)*, 2012 IEEE International. IEEE, 2012, pp. 1–5.
- [70] J. Tan and L. Wang, "Enabling reliability-differentiated service in residential distribution networks with phevs: A hierarchical game approach," *IEEE Transactions on Smart Grid*, vol. 7, no. 2, pp. 684–694, 2016.
- [71] W. Tushar, W. Saad, H. V. Poor, and D. B. Smith, "Economics of electric vehicle charging: A game theoretic approach," *IEEE Transactions on Smart Grid*, vol. 3, no. 4, pp. 1767–1778, 2012.

- [72] F. Ye, Y. Qian, and R. Q. Hu, "Incentive load scheduling schemes for phev battery exchange stations in smart grid."
- [73] M. Alizadeh, Y. Xiao, A. Scaglione, and M. Van Der Schaar, "Dynamic incentive design for participation in direct load scheduling programs," *IEEE Journal of Selected Topics in Signal Processing*, vol. 8, no. 6, pp. 1111–1126, 2014.
- [74] E. Bitar and Y. Xu, "Deadline differentiated pricing of deferrable electric loads," arXiv preprint arXiv:1407.1601, 2014.
- [75] S. Vandael, B. Claessens, M. Hommelberg, T. Holvoet, and G. Deconinck, "A scalable three-step approach for demand side management of plug-in hybrid vehicles," *IEEE Transactions on Smart Grid*, vol. 4, no. 2, pp. 720–728, 2013.
- [76] A. Von Meier, *Electric power systems: a conceptual introduction*. John Wiley & Sons, 2006.
- [77] J. P. Lopes, N. Hatziargyriou, J. Mutale, P. Djapic, and N. Jenkins, "Integrating distributed generation into electric power systems: A review of drivers, challenges and opportunities," *Electric power systems research*, vol. 77, no. 9, pp. 1189–1203, 2007.
- [78] P. S. Georgilakis and N. D. Hatziargyriou, "Optimal distributed generation placement in power distribution networks: models, methods, and future research," *IEEE Trans. Power Syst.*, vol. 28, no. 3, pp. 3420–3428, 2013.
- [79] R. Billington and R. N. Allan, *Reliability evaluation of power systems*. Springer Science & Business Media, 2013.
- [80] R. Allan and R. Billinton, "Probabilistic assessment of power systems," Proceedings of the IEEE, vol. 88, no. 2, pp. 140–162, 2000.
- [81] R. Karki and R. Billinton, "Cost-effective wind energy utilization for reliable power supply," *IEEE Transactions on Energy Conversion*, vol. 19, no. 2, pp. 435–440, 2004.
- [82] H. G. Stoll and L. J. Garver, *Least-cost electric utility planning*. J. Wiley, 1989.
- [83] IEEE-USA. (2007) Position statement plug-in electric hybrid vehicles. [Online]. Available: http://www.engr.uvic.ca/~mech459/Pub_References/PHEV0607.pdf

- [84] C. Canizares, J. Nathwani, K. Bhattacharya, M. Fowler, M. Kazerani, R. Fraser, I. Rowlands, and H. Gabbar, "Towards an ontario action plan for plug-in-electric vehicles (pevs)," *Waterloo Institute for Sustainable Energy, University of Waterloo*, 2010.
- [85] Hydro One, "Distributed generation technical interconnection requirements interconnection at voltages 50 kV and below," Toronto, ON, Canada, Rep. no. DT-10-015, vol. 2, p. 161, 2009.
- [86] U.S. Department of Transportation, Federal Highway Administration. (2015) National Household Travel Survey 2009. [Online]. Available: URL:http://nhts.ornl. gov
- [87] M. Baran and F. Wu, "Network reconfiguration in distribution systems for loss reduction and load balancing," *IEEE Trans. Power Del.*, vol. 4, no. 2, pp. 1401– 1407, Apr 1989.
- [88] J. Savier and D. Das, "Impact of network reconfiguration on loss allocation of radial distribution systems," *IEEE Trans. Power Del.*, vol. 22, no. 4, pp. 2473–2480, 2007.
- [89] Y. Atwa, E. El-Saadany, and A.-C. Guise, "Supply adequacy assessment of distribution system including wind-based dg during different modes of operation," *IEEE Transactions on Power Systems*, vol. 25, no. 1, pp. 78–86, 2010.
- [90] A. S. B. Humayd and K. Bhattacharya, "Comprehensive multi-year distribution system planning using back-propagation approach," *IET Gener. Transm. Distrib.*, vol. 7, no. 12, pp. 1415–1425, 2013.
- [91] W. El-Khattam, K. Bhattacharya, Y. Hegazy, and M. Salama, "Optimal investment planning for distributed generation in a competitive electricity market," *IEEE Trans. Power Syst.*, vol. 19, no. 3, pp. 1674–1684, 2004.
- [92] R. C. Dugan, T. E. McDermott, and G. J. Bal, "Planning for distributed generation," *IEEE Instrum. Meas. Mag.*, vol. 7, no. 2, pp. 80–88, 2001.
- [93] IESO. (2015) Hourly Ontario Energy Price (HOEP). [Online]. Available: http://www.ieso.ca
- [94] B. Zeng, J. Zhang, X. Yang, J. Wang, J. Dong, and Y. Zhang, "Integrated planning for transition to low-carbon distribution system with renewable energy generation and demand response," *IEEE Trans. Power Syst.*, vol. 29, no. 3, pp. 1153–1165, 2014.

- [95] J. J. Michalek, M. Chester, P. Jaramillo, C. Samaras, C.-S. N. Shiau, and L. B. Lave, "Valuation of plug-in vehicle life-cycle air emissions and oil displacement benefits," *Proceedings of the National Academy of Sciences*, vol. 108, no. 40, pp. 16554–16558, 2011.
- [96] Environmental and Protection Agency, "Fuel economy testing and labeling," Office of Transportation and Air Quality, Tech. Rep., 2014.
- [97] H. Huo, Z. Yao, K. He, and X. Yu, "Fuel consumption rates of passenger cars in china: labels versus real-world," *Energy Policy*, vol. 39, no. 11, pp. 7130–7135, 2011.
- [98] P. Plötz, S. Funke, and P. Jochem, "Real-world fuel economy and co2 emissions of plug-in hybrid electric vehicles," Fraunhofer Institute for Systems and Innovation Research (ISI), Tech. Rep., 2015.
- [99] U.S. Department of Energy. (2015) U.S. PEV Sales by Model. [Online]. Available: http://www.afdc.energy.gov/data/
- [100] Hybrid Committee, "SAE electric vehicle and plug in hybrid electric vehicle conductive charge coupler," SAE international, Detroit, Michigan, standard J1772, 2012.
- [101] J. Axsen, H. J. Bailey, and G. Kamiya, "The canadian plug-in electric vehicle survey (cpevs 2013): Anticipating purchase, use, and grid interactions in british columbia," 2013.
- [102] R. Billinton, S. Kumar, N. Chowdhury, K. Chu, K. Debnath, L. Goel, E. Khan, P. Kos, G. Nourbakhsh, and J. Oteng-Adjei, "A reliability test system for educational purposes-basic data," *IEEE Trans. Power Syst.*, vol. 4, no. 3, pp. 1238–1244, 1989.
- [103] S. Falahati, S. A. Taher, and M. Shahidehpour, "A new smart charging method for evs for frequency control of smart grid," *International Journal of Electrical Power & Energy Systems*, vol. 83, pp. 458–469, 2016.
- [104] C. B. Harris and M. E. Webber, "The impact of vehicle charging loads on frequency regulation procurements in ercot," in *Innovative Smart Grid Technologies Conference* (*ISGT*), 2014 IEEE PES. IEEE, 2014, pp. 1–5.
- [105] J. Kang, S. J. Duncan, and D. N. Mavris, "Real-time scheduling techniques for electric vehicle charging in support of frequency regulation," *Proceedia Computer Science*, vol. 16, pp. 767–775, 2013.

- [106] J. Tan and L. Wang, "Integration of plug-in hybrid electric vehicles into residential distribution grid based on two-layer intelligent optimization," *IEEE Transactions on Smart Grid*, vol. 5, no. 4, pp. 1774–1784, 2014.
- [107] Q. Wang, X. Liu, J. Du, and F. Kong, "Smart charging for electric vehicles: A survey from the algorithmic perspective," *IEEE Communications Surveys & Tutorials*, vol. 18, no. 2, pp. 1500–1517, 2016.
- [108] M. Haque, "Capacitor placement in radial distribution systems for loss reduction," *IEE Proceedings-Generation Transmission and Distribution*, vol. 146, no. 5, pp. 501– 505, 1999.
- [109] P. E. Gill, W. Murray, M. A. Saunders, A. Drud, and E. Kalvelagen, "GAMS/SNOPT: an SQP algorithm for large-scale constrained optimization," *GAMS-The Solver Manuals*, 2000.
- [110] GAMS Development Corporation, "General Algebraic Modeling System (GAMS)," http://www.gams.com, 2015.
- [111] Ontario Hydro. (2017) Time-of-use pricing. [Online]. Available: http://www.ontario-hydro.com/current-rates
- [112] O. E. Board, "2016 uniform transmission rates,", Tech. Rep., 2016.
- [113] F. A. Chacra, P. Bastard, G. Fleury, and R. Clavreul, "Impact of energy storage costs on economical performance in a distribution substation," *IEEE Transactions* on Power Systems, vol. 20, no. 2, pp. 684–691, 2005.
- [114] J. Tomić and W. Kempton, "Using fleets of electric-drive vehicles for grid support," Journal of Power Sources, vol. 168, no. 2, pp. 459–468, 2007.

APPENDICES

Appendix A

Distribution System Data

Line i-j	R (ohms)	X (ohms)
1.2	0.0922	0.0477
2.3	0.493	0.2511
3.4	0.366	0.1864
4.5	0.3811	0.1941
5.6	0.819	0.707
6.7	0.1872	0.6188
7.8	0.7114	0.2351
8.9	1.03	0.74
9.1	1.044	0.74
10.11	0.1966	0.065
11.12	0.3744	0.1238
12.13	1.468	1.155
13.14	0.5416	0.7129
14.15	0.591	0.526
15.16	0.7463	0.545
16.17	1.289	1.721
17.18	0.732	0.574
2.19	0.164	0.1565
19.2	1.5042	1.3554
20.21	0.4095	0.4784
21.22	0.7089	0.9373
3.23	0.4512	0.3083
23.24	0.898	0.7091
24.25	0.896	0.7011
6.26	0.203	0.1034
26.27	0.2842	0.1447
27.28	1.059	0.9337
28.29	0.8042	0.7006
29.3	0.5075	0.2585
30.31	0.9744	0.963
31.32	0.3105	0.3619
32.33	0.341	0.5302

TABLE A.1 Data for 33-Bus system: Feeder Parameters

Bus i	P (kW)	Q (kVAR)
2	100	60
3	90	40
4	120	80
5	60	30
6	60	20
7	200	100
8	200	100
9	60	20
10	60	20
11	45	30
12	60	35
13	60	35
14	120	80
15	60	10
16	60	20
17	60	20
18	90	40
19	90	40
20	90	40
21	90	40
22	90	40
23	90	50
24	420	200
25	420	200
26	60	25
27	60	25
28	60	20
29	120	70
30	200	600
31	150	70
32	210	100
33	60	40

TABLE A.2 Data for 33-Bus system: Load

Line i-j	R (ohms)	X (ohms)
1.2	0.0005	0.0012
2.3	0.0005	0.0012
3.4	0.0015	0.0036
4.5	0.0251	0.0294
5.6	0.366	0.1864
6.7	0.3811	0.1941
7.8	0.0922	0.047
8.9	0.0493	0.0251
9.1	0.819	0.2707
10.11	0.1872	0.0691
11.12	0.7114	0.2351
12.13	1.03	0.34
13.14	1.044	0.345
14.15	1.058	0.3496
15.16	0.1966	0.065
16.17	0.3744	0.1238
17.18	0.0047	0.0016
18.19	0.3276	0.1083
19.2	0.2106	0.069
20.21	0.3416	0.1129
21.22	0.014	0.0046
22.23	0.1591	0.0526
23.24	0.3463	0.1145
24.25	0.7488	0.2745
25.26	0.3089	0.1021
26.27	0.1732	0.0572
3.28	0.0044	0.0108
28.29	0.064	0.1565
29.3	0.3978	0.1315
30.31	0.0702	0.0232
31.32	0.351	0.116
32.33	0.839	0.2816
33.34	1.708	0.5646

TABLE A.3 Data for 69-Bus system: Feeder Parameters

Continued on next page

Line i-j	R (ohms)	X (ohms)
34.35	1.474	0.4673
3.36	0.0044	0.0108
36.37	0.064	0.1565
37.38	0.1053	0.123
38.39	0.0304	0.0355
39.4	0.0018	0.0021
40.41	0.7283	0.8509
41.42	0.31	0.3623
42.43	0.041	0.0478
43.44	0.0092	0.0116
44.45	0.1089	0.1373
45.46	0.0009	0.0012
4.47	0.0034	0.0084
47.48	0.0851	0.2083
48.49	0.2898	0.7091
49.5	0.0822	0.2011
8.51	0.0928	0.0473
51.52	0.3319	0.1114
9.53	0.174	0.0886
53.54	0.203	0.1034
54.55	0.2842	0.1447
55.56	0.2813	0.1433
56.57	1.59	0.5337
57.58	0.7837	0.263
58.59	0.3042	0.1006
59.6	0.3861	0.1172
60.61	0.5075	0.2585
61.62	0.0974	0.0496
62.63	0.145	0.0738
63.64	0.7105	0.3619
64.65	1.041	0.5302
11.66	0.2012	0.0611
66.67	0.0047	0.0014
12.68	0.7394	0.2444

TABLE A.3 Data for 69-Bus system: Feeder Parameters

Continued on next page

Line i-j	R (ohms)	X (ohms)
68.69	0.0047	0.0016

TABLE A.3 Data for 69-Bus system: Feeder Parameters

Bus i	P(kW)	Q (kVAR)	Bus i	P(kW)	Q (kVAR)
2	0	0	36	26	18.55
3	0	0	37	26	18.55
4	0	0	38	0	0
5	0	0	39	24	17
6	2.6	2.2	40	24	17
7	40.4	30	41	1.2	1
8	75	54	42	0	0
9	30	22	43	6	4.3
10	28	19	44	0	0
11	145	104	45	39.22	26.3
12	145	104	46	39.22	26.3
13	8	5.5	47	0	0
14	8	5.5	48	79	56.4
15	0	0	49	384.7	274.5
16	45.5	30	50	384	274.5
17	60	35	51	40.5	28.3
18	60	35	52	3.6	2.7
19	0	0	53	4.35	3.5
20	1	0.6	54	26.4	19
21	114	81	55	24	17.2
22	5.3	3.5	56	0	0
23	0	0	57	0	0
24	28	20	58	0	0
25	0	0	59	100	72
26	14	10	60	0	0
27	14	10	61	1244	888
28	26	18.6	62	32	23
29	26	18.6	63	0	0
30	0	0	64	227	162
31	0	0	65	59	42
32	0	0	66	18	13
33	14	10	67	18	13
34	19.5	14	68	28	20
35	6	4	69	28	20

TABLE A.4 Data for 69-Bus system: Load